

New insights in pulmonary cachexia

Citation for published version (APA):

Sanders, K. J. C. (2020). New insights in pulmonary cachexia: a multimodal imaging approach. [Doctoral Thesis, Maastricht University]. Maastricht University. https://doi.org/10.26481/dis.20201105ks

Document status and date: Published: 01/01/2020

DOI: 10.26481/dis.20201105ks

Document Version: Publisher's PDF, also known as Version of record

Please check the document version of this publication:

 A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

 The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these riahts.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.umlib.nl/taverne-license

Take down policy

If you believe that this document breaches copyright please contact us at:

repository@maastrichtuniversity.nl

providing details and we will investigate your claim.

NEW INSIGHTS IN PULMONARY CACHEXIA: A MULTIMODAL IMAGING APPROACH



© Karin Sanders, Maastricht, the Netherlands, 2020.

ISBN: 978 94 6421 012 5

Cover design: Evelien Jagtman Karin Sanders Layout: Production: Ipskamp

The research described in this thesis was performed at the Department of Respiratory Medicine, NUTRIM School of Nutrition and Translational Research in Metabolism at Maastricht University Medical Centre+, Maastricht, The Netherlands. The research conducted in this doctoral thesis was supported by a MUMC2020 grant.

NEW INSIGHTS IN PULMONARY CACHEXIA:

A MULTIMODAL IMAGING APPROACH

DISSERTATION

To obtain the degree of doctor at Maastricht University, on the authority of the Rector Magnificus, prof. dr. Rianne M. Letschert in accordance with the decision of the Board of Deans, to be defended in public on Thursday 5 November 2020 at 14:00 hours.

by

Karin Johanna Christina Sanders

Promotors

Prof. dr. A.M.W.J. Schols Prof. dr. A-M.C. Dingemans Prof. dr. F.M. Mottaghy

Assessment Committee

Prof. dr. K.H.C. Dejong (chair)
Prof. dr. M. Fallon (University of Edinburgh, United Kingdom)
Dr. F.M.E. Franssen
Prof. dr. M.A.E. de van der Schueren (HAN University of Applied Sciences, Nijmegen and Wageningen University and Research)
Prof. dr. J.E. Wildberger

CONTENTS

Chapter 1	General introduction and outline of the thesis	7
Chapter 2	Imaging approaches to understand disease complexity: chronic obstructive pulmonary disease as a clinical model	21
Chapter 3	Cross-sectional and longitudinal assessment of muscle from regular chest computed tomography scans: L1 and pectoralis muscle compared to L3 as reference in non-small cell lung cancer	39
Chapter 4	Cachexia in chronic obstructive pulmonary disease: new insights and therapeutic perspective	55
Chapter 5	Computed tomography derived muscle remodelling after bronchoscopic lung volume reduction in advanced emphysema	89
Chapter 6	Effect of bronchoscopic lung volume reduction in advanced emphysema on energy balance regulation	101
Chapter 7	Brown adipose tissue activation in relation to hypermetabolism in emphysematous chronic obstructive pulmonary disease patients	117
Chapter 8	Early weight loss during chemoradiotherapy has detrimental impact on outcome in non-small cell lung cancer	137
Chapter 9	The prognostic value of early onset CT derived loss of muscle and adipose tissue during chemotherapy in metastatic non-small cell lung cancer	153
Chapter 10	Can radiomics help to predict skeletal muscle response to chemotherapy in stage IV non-small cell lung cancer?	167
Chapter 11	Summary, general discussion, and future perspective	181
	Samenvatting	199
	Valorisation	205
	List of publications	211
	Acknowledgements	217
	Curriculum Vitae	223



1

General introduction and outline of the thesis

GENERAL INTRODUCTION

Despite well-known health risks, a substantial proportion of people continue to customarily smoke tobacco [1]. Worldwide, over 900 million people smoke, with prevalence rates increasing every year [2]. While there is a gradual downward trend in smoking in the Netherlands, in 2017 still 23.1% of the adults smoked [3]. Of the profuse smoking-related diseases, chronic obstructive pulmonary disease (COPD) is most common and lung cancer is the most deadly [4].

COPD is an airway and lung disease with persistent airflow obstruction, which is the result of small airway remodelling and loss of elastic recoil. The diagnosis is based on the presence and severity of airflow limitation measured by post-bronchodilator spirometry, combined with respiratory symptoms and past exacerbations. Clinically, the disease expresses in a variable combination of thickening and obstruction of the airways (chronic bronchitis) and destruction of the air sacs (emphysema) [5]. Although cigarette smoke is acknowledged as the most important risk factor for COPD, epidemiologic studies show that non-smokers may also develop COPD [6], arguing for other risk factors involved. The list of risk factors for COPD include genetic susceptibility for airway disease, processes occurring during gestation, birth and childhood affecting lung development, high intake of unhealthy or western diet and environmental exposures to particles such as toxic gasses, air pollution and biomass fuel [5,7,8]. In the Netherlands, an estimated 600,000 people suffer from COPD, and it is the sixth leading cause of death [9,10].

Lung cancer is less common than COPD, but accounted for over 10,000 deaths in the Netherlands in 2017 [9]. Based on biological behaviour, lung cancer can be subdivided in two major subtypes, i.e. non-small cell lung cancer (NSCLC) and small cell lung cancer. NSCLC occurs approximately in 80% of new lung cancer cases [11]. The majority (85%) of lung cancer cases are due to smoking [12], however only 10% of the smokers will develop the disease. Other identified risk factors are occupational exposure to asbestos, nickel and chromium, air pollution and genetic determinants [12,13]. In the Netherlands, an estimated 26,000 people suffer from lung cancer, which tops the list of cancer-related deaths [14,15].

COPD and lung cancer are interrelated. While they share environmental risk factors, epidemiological studies demonstrated that lung cancer is up to five times more likely to occur in smokers with airflow obstruction than those with normal lung function [16]. This risk increases with progressive decline in airflow obstruction [17]. Besides being each other's comorbid disease, the diseases also share extrapulmonary manifestations including unintended weight loss and muscle wasting [18-20].

Cachexia

Although the lung is the major organ of pathology in both COPD and lung cancer, it is increasingly recognized that systemic manifestation of the disease contribute to symptom burden and health status. As illustrated by the following cases unintended weight loss occurs both in COPD and NSCLC.

Case 1: COPD

Patient I, a 58 years old female, has a forced expiratory volume in 1 sec, residual volume and diffusion capacity for carbon monoxide of 28%, 274% and 38% of predicted, respectively. This indicates she suffers from emphysema and severe hyperinflation. Furthermore, she has an arterial oxygen tension of 7.3 kPa, with a carbon dioxide pressure of 4.6 kPa, consistent with arterial hypoxaemia without respiratory insufficiency. Last year she had been admitted to the hospital twice for an acute exacerbation. She is therefore labelled COPD GOLD D. After having smoked 40 pack years she succeeded quitting. During the past years she has lost over 10 kg of body weight and currently she weighs only 36.7 kg resulting in a body mass index of 15.1 kg/m². Despite optimal pharmacological treatment, she experiences increased dyspnea upon exertion, which limits her ability to walk to the supermarket. She is therefore homebound and spends most of the day sedentary. Currently, she is referred to the pulmonologist to evaluate whether she might be an appropriate candidate for bronchoscopic lung volume reduction.

Case 2: NSCLC

Patient II is a 63 years old female and recently diagnosed with stage III NSCLC for which she started with concomitant chemoradiotherapy 3 weeks ago. Currently, she is hospitalized due to symptoms related to treatment side effects, including neutropenic fever and grade 2 esophagitis. Due to these treatment side effects, the next chemotherapy cycle is postponed. During hospitalization it was noticed that she had lost more than 5% of her body weight since start of treatment. Unlike the patient, her daughter is very distressed about the weight loss. She is worried that malnutrition worsens her mother's condition and affects prognosis.

These cases illustrate two patients, with different pulmonary diseases, who both suffer unintentional weight loss. The subjects differ in weight loss progression, being slower in patient I. Progressive unintentional weight loss is a characteristic of cachexia due to wasting of skeletal muscle and adipose tissue [21]. Cachexia is prevalent in a wide range of illnesses, including COPD and lung cancer. Prevalences ranging from 15 to 40% in COPD and NSCLC depending on definition, disease and disease stage [22-25].

Cachexia associates with a multitude of morbidities affecting clinical outcome. In COPD, patients experience loss of skeletal muscle function, reduced exercise capacity and decreased health status [26,27]. Importantly, cachexia also increases mortality, independent of airflow obstruction [23,28]. Cachectic patients with lung cancer experience diminished quality of life [19], intolerance to anticancer treatment [29] and shorter survival [30,31]. Although the exact contribution of cachexia to mortality is difficult to establish, it is estimated that 20-30% of cancer-related deaths are a direct result of cachexia [32].

Diagnosis of muscle and adipose tissue loss in COPD and NSCLC

The negative impact of cachexia on clinical outcome and healthcare utilization [23,31,33,34], emphasizes the importance of early identification. Weight loss history in a predefined time period or low body mass index are most commonly used in routine clinical practice to assess cachexia [35]. However, low muscle mass is not restricted to those with lean stature [31] and may be masked by adiposity or fluid retention [30,36,37]. Currently, the role of medical imaging is evolving from a primarily diagnostic tool towards a more central role in characterization of patient heterogeneity. In that context there is a growing number of opportunities to combine the imaging for medical purpose with assessment of body composition (reviewed in chapter 2).

Furthermore, computed tomography is increasingly used for single-slice assessment of muscle mass. The third lumbar level is considered as reference for muscle, but chest scans reach generally not further than the first lumbar level. For this reason, in COPD research pectoralis muscle has been proposed as alternative [38,39]. However, to date no studies compared the method with appropriate reference methods or explored alternative locations on chest scans.

Pathophysiology of cachexia

The underlying mechanisms responsible for cachexia in COPD and lung cancer are considered complex and multifactorial, and may differ between the two diseases, between different pulmonary phenotypes and related to disease severity.

The onset of low muscle mass in COPD can arise either from failure to attain a normal skeletal muscle structure and function during early childhood or from an accelerated decline of muscle mass in later life. There is evidence that some COPD patients failed to achieve the normal spirometric plateau because of previous exposures to childhood respiratory infections [40], which might also adversely affect maturation of skeletal muscle. In line with this hypothesis, a constitutionally lower fat-free mass was found among older aged subjects with mild-to-moderate obstructive lung disease compared to smokers without obstructive lung disease. Seven-year follow-up revealed no differences in pattern and progression of age associated changes in fat-free mass [41]. Further, a genetic predisposition for lean stature in COPD patients was found [42] and this lean stature has also been suggested to be a risk factor for development of COPD [43].

There is evidence that a subgroup of COPD patients has a steeper decline of fat-free mass, associated with the number of exacerbations [44,45]. Furthermore, dyspnea upon exertion partially contributes to physical inactivity, which accelerates the downward spiral of deconditioning of the skeletal muscle. However, this does not fully explain the high prevalence and extensiveness of low muscle mass. Additionally, individual differences in the amount of muscle wasting complicate the understanding of the underlying mechanisms.

Great progress has been made in understanding molecular alterations underlying muscle decline. Muscle maintenance is tightly regulated by the balance between muscle cell synthesis and growing on one hand, and muscle breakdown mediated by proteolysis, autophagy and apoptosis on the other hand. In muscle-wasted COPD patients, myogenic signalling and muscle protein turnover is elevated [46]. Recent network analyses also highlight a coordinated regulation of muscle mass and metabolic plasticity [47], being disturbed in COPD [48]. Together, these molecular alterations suggest ongoing muscle repair and remodelling, which might be insufficient for muscle maintenance in COPD.

Muscle wasting commonly occurs in COPD [23] and occurs more frequently in the emphysematous phenotype [18,26] and in advanced disease [23]. During acute disease exacerbations this gradual loss might be accelerated [44,45]. Muscle wasting can occur with and without decrease in body mass index [23], suggesting that different pathophysiological mechanisms of impaired protein turnover versus whole body energy metabolism affect muscles. However, research on whole body energy regulation and fat metabolism has been underexposed.

The classical description of "the pink puffer" by Filley et al. [49] and more recent research using advanced body composition methods, indicate that in particular the emphysematous COPD patient is prone to cachexia. Results from a 3-year follow-up study revealed that patients with more emphysema exhibited an accelerated decline in both lung function and fat-free mass [50]. The emphysema phenotype is hallmarked by a reduction in lung elastic recoil and progressive hyperinflation, resulting in elevated airway resistance and contributing to impaired lung mechanics [51]. Although increased breathing workload in COPD is well documented [52,53], the effect of hyperinflation and emphysema on energy metabolism and muscle mass regulation is unclear.

Activation of brown adipose tissue (BAT) or browning of white adipose tissue has been proposed as putative trigger for hypermetabolism. By dissipating energy stored in triglycerides as heat [54], BAT activation contributes to cold-induced energy expenditure [54,55]. In healthy lean men, energy expenditure increased on average with 17.2% [56]. This implies that BAT thermogenesis may be a significant component of whole body energy expenditure and thereby play a role in regulation of body weight. Indeed, studies employing cold-induced BAT activation showed that BAT volume and activity are lower in individuals who are obese and BAT activity is inversely correlated with body weight [57-59]. BAT has been primarily researched as a potential target for mitigating obesity as it may facilitate weight loss. Yet few research attempts have been made at the other side of the spectrum, why patients with COPD and lung cancer unintentionally lose weight. Controlled prospective human studies investigating the role of sustained BAT activation in development of cachexia are lacking [60-62].

In cancer cachexia, the tumour is thought to be a potential driver of tissue wasting, partly mediated by tumour and host elicited inflammation. Cachectic NSCLC patients exhibited a prominent systemic inflammation, whereas lower systemic inflammation was apparent in precachectic patients. Furthermore, plasma transfer experiments revealed that plasma of both cachectic and precachectic patients induced inflammatory signalling in skeletal muscle, with increased proteolysis and an impaired anabolic signalling [63]. Cancer therapy aiming to treat the tumour also affects other tissues. Concurrent chemoradiotherapy, which is the standard treatment for patients with unresectable locally advanced NSCLC [64] frequently induces radiation esophagitis and associated dysphagia [65]. It is therefore assumed that weight loss results from radiation esophagitis due to impaired dietary intake. Whether weight loss also occurs before onset of radiation esophagitis and whether this weight loss contributes to prognosis requires further investigation.

In addition, it has been suggested that chemotherapy agents itself also affect cachexia. While some known side effects including nausea, diarrhoea and anorexia may contribute to weight loss, these symptoms do not fully explain muscle wasting. Evidence from experimental rodent cancer models suggest a direct effect of chemotherapeutics on muscle wasting and lipolysis [66, 67]. However, the extent to which muscle mass and adipose tissue mass are affected in a clinical setting is unknown.

Prediction of future muscle loss

There are currently no biomarkers available to identify patients at risk for development of cachexia. In oncology, radiomics is a growing era of interest. Radiomics is a method that quantitatively extracts features including shape, size, intensity and texture that are related to tumour pathophysiology from standard-of-care medical images [68]. In oncologic patients extracted tumour features aid in prognosis prediction [69-72]. Radiomic analysis of muscle features has never been used to predict future muscle loss.

AIMS AND OUTLINE OF THIS THESIS

Prevention and timely treatment of cachexia in COPD and NSCLC requires adequate and early identification of patients at risk and detailed understanding of the pathophysiology involved. Comparative clinical research between COPD and NSCLC may shed new light on pulmonary cachexia. Two specific aims were addressed:

- To evaluate opportunities to combine standard-of-care medical images with assessment and prediction of tissue wasting
- To provide insights in putative triggers involved in muscle wasting, including the contribution
 of impaired breathing mechanics and brown adipose tissue on energy metabolism and
 the impact of cancer treatment on muscle wasting.

In **Chapter 2**, we review the potential use of diagnostic medical images for characterization of COPD. Furthermore, we describe the potential implications for body composition assessment.

In **Chapter 3**, we investigate whether chest computed tomography scans made in daily clinical practice of respiratory patients can be used to assess cross-sectional and longitudinal body composition.

In Chapter 4, we review recent insights on the pathophysiology of cachexia in COPD.

Chapter 5 presents analysis of changes in skeletal muscle mass and adipose tissue in severe emphysematous COPD patients before and after treated with bronchoscopic lung volume reduction.

Chapter 6 reports a proof of concept clinical trial investigating the contribution of impaired lung mechanics on energy balance regulation in severe emphysematous COPD patients treated with bronchoscopic lung volume reduction.

Chapter 7 presents a controlled clinical study on brown adipose tissue activation and its potential contribution to whole body energy expenditure in COPD.

In cancer cachexia, cancer therapies aiming to treat the tumour might also affects other tissues, and thereby contributes to tissue wasting.

Chapter 8 reports the impact of weight loss within 3 weeks after initiation of concurrent administration of chemotherapy and radiotherapy on survival in stage III NSCLC patients.

Chapter 9 presents a longitudinal analysis of computed tomography-derived body composition during chemotherapy in therapy naïve stage IV NSCLC patients.

In **Chapter 10**, we use a radiomics approach to assess whether skeletal muscle radiomic features can predict future muscle loss.

Finally, the results of the studies described in the current dissertation will be discussed and placed in a broader context in **Chapter 11**.

REFERENCES

- 1. Peto R, Chen ZM, Boreham J. Tobacco--the growing epidemic. Nature medicine. 1999;5:15-17.
- Ng M, Freeman MK, Fleming TD, et al. Smoking prevalence and cigarette consumption in 187 countries, 1980-2012. Jama. 2014;311:183-192.
- Wood BP. Visual expertise. Radiology. 1999;211:1-3.
- Houghton AM. Mechanistic links between COPD and lung cancer. Nature reviews Cancer. 2013;13:233-245.
- Vogelmeier CF, Criner GJ, Martinez FJ, et al. Global Strategy for the Diagnosis, Management, and Prevention of Chronic Obstructive Lung Disease 2017 Report. GOLD Executive Summary. Am J Respir Crit Care Med. 2017;195:557-582.
- Lamprecht B, McBurnie MA, Vollmer WM, et al. COPD in never smokers: results from the population-based burden of obstructive lung disease study. Chest. 2011;139:752-763.
- Eisner MD, Anthonisen N, Coultas D, et al. An official American Thoracic Society public policy statement: Novel risk factors and the global burden of chronic obstructive pulmonary disease. American journal of respiratory and critical care medicine. 2010;182:693-718.
- Mount S, Cirillo E, Stewart K, et al. Network Analysis of Genome-Wide Association Studies for Chronic Obstructive Pulmonary Disease in the Context of Biological Pathways. American journal of respiratory and critical care medicine. 2019;200:1439-1441.
- CBS Overledenen, belangrijke doodsoorzaken.
 2018; https://opendata.cbs.nl/statline/#/CBS/nl/.
 Accessed February 9th, 2020.
- Nivel Zorgregistratie. Incidentie en prevalentie COPD. 2018; https://www.nivel.nl/nl/ Accessed February 9th, 2020.
- Molina JR, Yang P, Cassivi SD, Schild SE, Adjei AA. Non-small cell lung cancer: epidemiology, risk factors, treatment, and survivorship. Mayo Clinic proceedings. 2008;83:584-594.
- Planchard D, Popat S, Kerr K, et al. Metastatic non-small cell lung cancer: ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann Oncol. 2018;29:iv192-iv237.
- Alberg AJ, Brock MV, Ford JG, Samet JM, Spivack SD. Epidemiology of lung cancer: Diagnosis and management of lung cancer, 3rd ed: American College of Chest Physicians evidence-based clinical

practice guidelines. Chest. 2013;143:e1S-e29S.

- 14. Cox K. Perceiving clinical evidence. Medical Education. 2002;36:1189-1195.
- CBS Overledenen, belangrijke doodsoorzaken nieuwvormingen. https://opendata.cbs. nl/statline/#/CBS/nl/dataset/7052_95/ table?dl=E608. Accessed February 9th, 2020.
- Young RP, Hopkins RJ, Christmas T, Black PN, Metcalf P, Gamble GD. COPD prevalence is increased in lung cancer, independent of age, sex and smoking history. The European respiratory journal. 2009;34:380-386.
- Wasswa-Kintu S, Gan WQ, Man SF, Pare PD, Sin DD. Relationship between reduced forced expiratory volume in one second and the risk of lung cancer: a systematic review and meta-analysis. Thorax. 2005;60:570-575.
- Engelen MP, Schols AM, Lamers RJ, Wouters EF. Different patterns of chronic tissue wasting among patients with chronic obstructive pulmonary disease. Clinical nutrition. 1999;18:275-280.
- Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. Journal of cachexia, sarcopenia and muscle. 2017;8:759-767.
- Baracos VE, Reiman T, Mourtzakis M, Gioulbasanis I, Antoun S. Body composition in patients with non-small cell lung cancer: a contemporary view of cancer cachexia with the use of computed tomography image analysis. The American journal of clinical nutrition. 2010;91:1133S-1137S.
- Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus. The Lancet Oncology. 2011;12:489-495.
- Schols AM, Soeters PB, Dingemans AM, Mostert R, Frantzen PJ, Wouters EF. Prevalence and characteristics of nutritional depletion in patients with stable COPD eligible for pulmonary rehabilitation. The American review of respiratory disease. 1993;147:1151-1156.
- 23. Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. American journal of respiratory and critical care medicine. 2006;173:79-83.
- 24. Dewys WD, Begg C, Lavin PT, et al. Prognostic effect of weight loss prior to chemotherapy in cancer

patients. Eastern Cooperative Oncology Group. The American journal of medicine. 1980;69:491-497.

- 25. Laviano A, Meguid MM. Nutritional issues in cancer management. Nutrition. 1996;12:358-371.
- Engelen MP, Schols AM, Baken WC, Wesseling GJ, Wouters EF. Nutritional depletion in relation to respiratory and peripheral skeletal muscle function in out-patients with COPD. The European respiratory journal. 1994;7:1793-1797.
- Mostert R, Goris A, Weling-Scheepers C, Wouters EF, Schols AM. Tissue depletion and health related quality of life in patients with chronic obstructive pulmonary disease. Respiratory medicine. 2000;94:859-867.
- Schols AM, Broekhuizen R, Weling-Scheepers CA, Wouters EF. Body composition and mortality in chronic obstructive pulmonary disease. The American journal of clinical nutrition. 2005;82:53-59.
- 29. Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapy-induced haematological toxicity in advanced non-small cell lung cancer. Lung cancer. 2015;90:85-91.
- Martin L, Birdsell L, Macdonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2013;31:1539-1547.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. The Lancet Oncology. 2008;9:629-635.
- von Haehling S, Anker SD. Cachexia as a major underestimated and unmet medical need: facts and numbers. Journal of cachexia, sarcopenia and muscle. 2010;1:1-5.
- Decramer M, Gosselink R, Troosters T, Verschueren M, Evers G. Muscle weakness is related to utilization of health care resources in COPD patients. The European respiratory journal. 1997;10:417-423.
- Shachar SS, Deal AM, Weinberg M, et al. Skeletal Muscle Measures as Predictors of Toxicity, Hospitalization, and Survival in Patients with Metastatic Breast Cancer Receiving Taxane-Based Chemotherapy. Clin Cancer Res. 2017;23:658-665.
- Chailleux E, Laaban JP, Veale D. Prognostic value of nutritional depletion in patients with COPD treated by long-term oxygen therapy: data from

the ANTADIR observatory. Chest. 2003;123:1460-1466.

- 36. Gioulbasanis I, Martin L, Baracos VE, Thezenas S, Koinis F, Senesse P. Nutritional assessment in overweight and obese patients with metastatic cancer: does it make sense? Annals of oncology : official journal of the European Society for Medical Oncology. 2015;26:217-221.
- Cederholm T, Barazzoni R, Austin P, et al. ESPEN guidelines on definitions and terminology of clinical nutrition. Clinical nutrition. 2017;36:49-64.
- Diaz AA, Zhou L, Young TP, et al. Chest CT measures of muscle and adipose tissue in COPD: genderbased differences in content and in relationships with blood biomarkers. Acad Radiol. 2014;21:1255-1261.
- McDonald ML, Diaz AA, Ross JC, et al. Quantitative computed tomography measures of pectoralis muscle area and disease severity in chronic obstructive pulmonary disease. A cross-sectional study. Ann Am Thorac Soc. 2014;11:326-334.
- Stocks J, Hislop A, Sonnappa S. Early lung development: lifelong effect on respiratory health and disease. Lancet Respir Med. 2013;1:728-742.
- van den Borst B, Koster A, Yu B, et al. Is agerelated decline in lean mass and physical function accelerated by obstructive lung disease or smoking? Thorax. 2011;66:961-969.
- Wan ES, Cho MH, Boutaoui N, et al. Genomewide association analysis of body mass in chronic obstructive pulmonary disease. American journal of respiratory cell and molecular biology. 2011;45:304-310.
- Harik-Khan RI, Fleg JL, Wise RA. Body mass index and the risk of COPD. Chest. 2002;121:370-376.
- Rutten EP, Spruit MA, McDonald ML, et al. Continuous fat-free mass decline in COPD: fact or fiction? The European respiratory journal. 2015;46:1496-1498.
- 45. Hopkinson NS, Tennant RC, Dayer MJ, et al. A prospective study of decline in fat free mass and skeletal muscle strength in chronic obstructive pulmonary disease. Respiratory research. 2007;8:25.
- Kneppers AEM, Langen RCJ, Gosker HR, et al. Increased Myogenic and Protein Turnover Signaling in Skeletal Muscle of Chronic Obstructive Pulmonary Disease Patients With Sarcopenia. J Am Med Dir Assoc. 2017;18:637 e631-637 e611.
- 47. Kneppers A, Leermakers P, Pansters N, et al.

Coordinated regulation of skeletal muscle mass and metabolic plasticity during recovery from disuse. FASEB J. 2019;33:1288-1298.

- Tenyi A, Cano I, Marabita F, et al. Network modules uncover mechanisms of skeletal muscle dysfunction in COPD patients. J Transl Med. 2018;16:34.
- Filley GF, Beckwitt HJ, Reeves JT, Mitchell RS. Chronic obstructive bronchopulmonary disease.
 II. Oxygen transport in two clinical types. The American journal of medicine. 1968;44:26-38.
- Celli BR, Locantore N, Tal-Singer R, et al. Emphysema and extrapulmonary tissue loss in COPD: a multi-organ loss of tissue phenotype. The European respiratory journal. 2018;51.
- Criner GJ, Cordova F, Sternberg AL, Martinez FJ. The National Emphysema Treatment Trial (NETT): Part I: Lessons learned about emphysema. American journal of respiratory and critical care medicine. 2011;184:763-770.
- Sridhar MK, Carter R, Lean ME, Banham SW. Resting energy expenditure and nutritional state of patients with increased oxygen cost of breathing due to emphysema, scoliosis and thoracoplasty. Thorax. 1994;49:781-785.
- Shindoh C, Hida W, Kikuchi Y, et al. Oxygen consumption of respiratory muscles in patients with COPD. Chest. 1994;105:790-797.
- Heaton GM, Wagenvoord RJ, Kemp A, Jr., Nicholls DG. Brown-adipose-tissue mitochondria: photoaffinity labelling of the regulatory site of energy dissipation. Eur J Biochem. 1978;82:515-521.
- Yoneshiro T, Aita S, Matsushita M, et al. Brown adipose tissue, whole-body energy expenditure, and thermogenesis in healthy adult men. Obesity (Silver Spring). 2011;19:13-16.
- Claessens-van Ooijen AM, Westerterp KR, Wouters L, Schoffelen PF, van Steenhoven AA, van Marken Lichtenbelt WD. Heat production and body temperature during cooling and rewarming in overweight and lean men. Obesity (Silver Spring). 2006;14:1914-1920.
- Cypess AM, Lehman S, Williams G, et al. Identification and importance of brown adipose tissue in adult humans. N Engl J Med. 2009;360:1509-1517.
- Wang Q, Zhang M, Xu M, et al. Brown adipose tissue activation is inversely related to central obesity and metabolic parameters in adult human.

PLoS One. 2015;10:e0123795.

- Lee P, Greenfield JR, Ho KK, Fulham MJ. A critical appraisal of the prevalence and metabolic significance of brown adipose tissue in adult humans. Am J Physiol Endocrinol Metab. 2010;299:E601-606.
- Tsoli M, Moore M, Burg D, et al. Activation of thermogenesis in brown adipose tissue and dysregulated lipid metabolism associated with cancer cachexia in mice. Cancer Res. 2012;72:4372-4382.
- Petruzzelli M, Schweiger M, Schreiber R, et al. A switch from white to brown fat increases energy expenditure in cancer-associated cachexia. Cell Metab. 2014;20:433-447.
- Kir S, White JP, Kleiner S, et al. Tumour-derived PTHrelated protein triggers adipose tissue browning and cancer cachexia. Nature. 2014;513:100-104.
- 63. Op den Kamp CM, Langen RC, Snepvangers FJ, et al. Nuclear transcription factor kappa B activation and protein turnover adaptations in skeletal muscle of patients with progressive stages of lung cancer cachexia. The American journal of clinical nutrition. 2013;98:738-748.
- 64. Auperin A, Le Pechoux C, Rolland E, et al. Metaanalysis of concomitant versus sequential radiochemotherapy in locally advanced nonsmall-cell lung cancer. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2010;28:2181-2190.
- Govaert SL, Troost EG, Schuurbiers OC, et al. Treatment outcome and toxicity of intensitymodulated (chemo) radiotherapy in stage III non-small cell lung cancer patients. Radiat Oncol. 2012;7:150.
- Chen JA, Splenser A, Guillory B, et al. Ghrelin prevents tumour- and cisplatin-induced muscle wasting: characterization of multiple mechanisms involved. Journal of cachexia, sarcopenia and muscle. 2015;6:132-143.
- Garcia JM, Scherer T, Chen JA, et al. Inhibition of cisplatin-induced lipid catabolism and weight loss by ghrelin in male mice. Endocrinology. 2013;154:3118-3129.
- Kumar V, Gu Y, Basu S, et al. Radiomics: the process and the challenges. Magn Reson Imaging. 2012;30:1234-1248.
- Mattonen SA, Tetar S, Palma DA, Louie AV, Senan S, Ward AD. Imaging texture analysis for automated prediction of lung cancer recurrence

- Coroller TP, Agrawal V, Narayan V, et al. Radiomic phenotype features predict pathological response in non-small cell lung cancer. Radiother Oncol. 2016;119:480-486.
- Aerts HJ, Velazquez ER, Leijenaar RT, et al. Decoding tumour phenotype by noninvasive imaging using a quantitative radiomics approach. Nat Commun. 2014;5:4006.
- Lambin P, Leijenaar RTH, Deist TM, et al. Radiomics: the bridge between medical imaging and personalized medicine. Nat Rev Clin Oncol. 2017;14:749-762.

General introduction



2

Imaging approaches to understand disease complexity: chronic obstructive pulmonary disease as a clinical model

Karin J.C. Sanders*, Samuel Y. Ash*, George R. Washko, Felix M. Mottaghy, Annemie M.W.J. Schols. *Authors contributed equally

Journal of Applied Physiology. 2018; 124(2):512-20.

ABSTRACT

The clinical manifestations of chronic obstructive pulmonary disease (COPD) reflect an aggregate of multiple pulmonary and extra-pulmonary processes. It is increasingly clear that full assessment of these processes is essential to characterize disease burden and to tailor therapy. Medical imaging has advanced such that it is now possible to obtain in vivo insight in the presence and severity of lung disease associated features. In this review, we have assembled data from multiple disciplines of medical imaging research to review the role of imaging in characterization of COPD. Topics include imaging of the lungs, body composition and extra-pulmonary tissue metabolism. The primary focus is on imaging modalities which are widely available in clinical care settings and which potentially contribute to describing COPD heterogeneity and enhance our insight in underlying pathophysiological processes and their structural and functional effects.

INTRODUCTION

The clinical manifestations of chronic obstructive pulmonary disease (COPD) reflect an aggregate of multiple pulmonary and extra-pulmonary processes. It is increasingly clear that full assessment of these processes is essential to characterize disease burden and to tailor therapy [1,2]. Examples of these COPD associated features include emphysematous destruction and fibrosis of the lung parenchyma, cardiovascular disease, sarcopenia, abdominal adiposity, and osteoporosis [2-6]. Highly effective therapies are available for many of these conditions and yet their prevalence and role in the mortality and morbidity of those with chronic respiratory diseases is often underappreciated [1,2,7].

While no imaging study is critical to the initial diagnosis and management of COPD, many patients with COPD undergo imaging for other reasons such as lung cancer screening, screening for osteoporosis, or acute changes in their clinical status. Although these medical images are often used principally for the indication that prompted their acquisition, they also provide the opportunity to obtain in vivo insight into the presence and severity of these and other conditions [4,8-11]. Exciting new areas of imaging research are focused on singular aspects of these comorbidities, however such specialization often causes us to lose a larger view of efforts in this field. Thus, while we do not advocate the routine use of imaging in COPD without another indication, extraction of morphological information and comorbidity patterns from already available medical images, acquired in routine clinical care, may provide new opportunities to better characterize patients and individualize treatment in the future.

In this review, we have brought together the available information on structural and metabolic imaging with regard to its potential usefulness in improving the understanding of COPD and with particular emphasis on quantitative techniques. More specifically, we focus on the structural information obtained using routine computed tomography (CT) imaging as well as on the information on body composition from both CT imaging and dual-energy X-ray absorptiometry (DXA). Furthermore, we briefly discuss the potential of metabolic molecular imaging using the standard positron emission tomography (PET) radiopharmaceutical fluordeoxyglucose (FDG) in COPD when available for additional scientific purpose. These three imaging modalities allow the noninvasive characterization of many of the underlying pathophysiological processes in COPD and their structural and functional effects. Of note, while magnetic resonance imaging and optical coherence tomography have been and continue to be investigated as potential imaging techniques in COPD, none have yet found a place in routine clinical care and so are beyond the scope of this review. In addition, while ultrasound is increasingly useful in the diagnosis of acute chest diseases such as pneumothorax, and endobronchial ultrasound has been widely adopted for procedures such as lymph node biopsy, neither has been routinely used in COPD and so have also been excluded.

LUNG STRUCTURE

Functional, structural and molecular imaging of the lungs, using a variety of modalities, has revealed new information regarding the complexity of COPD. This discussion will primarily focus on the role of imaging in understanding COPD, but it should be noted that there is extensive work in this area on lung cancer, interstitial lung disease, asthma, pulmonary hypertension, and other chronic lung diseases.

Computed Tomography

Contributions to phenotyping: CT imaging utilizes x-rays to acquire a three dimensional (3D) attenuation image of the body. The images themselves are comprised of voxels, each of which has a density, typically measured in Hounsfield units (HU) that distinguishes water from air, enabling the characterization of various tissue types to that particular area. These voxels are then reconstructed to using various grey values to create the overall CT image. Although CT imaging is not indicated for the routine diagnosis or clinical staging of COPD, due to the overlapping risk factors and clinical manifestations of COPD with other chest diseases such as lung cancer, it is frequently obtained for other reasons in these patients [12,13]. There are a multitude of processes evident in CT imaging of the lungs, which may be grouped by anatomic compartment, namely those that affect the parenchyma, the airways and the vasculature.

With regard to the lung parenchyma, emphysema is the most prominent anatomic and radiologic COPD associated process (figure 1a), and the volume, distribution and subtype of emphysema on CT have yielded insights into the clinical management and physiology of COPD [14,15]. For instance, in the National Emphysema Treatment Trial, 1200 COPD patients with hyperinflation were randomized to either lung volume reduction surgery (LVRS) or optimal medical therapy. While there was no overall survival benefit to LVRS, a non-pre-specified subgroup analysis based on the combination of baseline CT emphysema distribution and exercise testing identified a subset of patients who responded differently to LVRS compared to medical treatment. Subjects with upper lobe or upper zone predominant emphysema and low exercise capacity had significantly lower mortality than subjects randomized to medical treatment [16]. Subsequent work suggests that such clinical differences between those with upper lobe predominant or heterogeneous emphysema and more homogeneous emphysema may reflect not only anatomic differences but physiologic ones as well. For instance, those with homogenous emphysema have been shown to have a greater degree of dynamic hyperinflation during exercise than those with upper lobe predominant disease suggesting potential differences in their lung mechanics and differences in the underlying pathophysiology of their disease [17]. Perhaps more importantly clinically is the role of CT in identifying patients for endobronchial lung volume reduction procedures, which are bronchoscopic alternatives to surgical LVRS including endobronchial valves, coils, glue and other devices. The effectiveness of endobronchial valves in particular may be limited by collateral ventilation in the setting of incomplete interlobular fissures.

Collateral ventilation prevents the valves from fully deflating the target area thus limiting their effectiveness. Small studies have suggested that CT imaging may be sensitive, though not necessarily specific, for the detection of the interlobular fissures, and new automated techniques may improve its performance in this area making it an important tool for the preprocedure planning of patients undergoing endobronchial lung volume reduction [18-20].

While less immediately clinically applicable, the CT identification of emphysema subtypes provide a non-invasive window into the pathophysiology of the disease, and thus open the door to additional treatment options. Smoking, the most frequent cause of emphysema, is most commonly associated with centrilobular disease, and, to a lesser, but still significant extent, with paraseptal emphysema [21-23]. Additional subtypes include panlobular disease which tends to be rare except in patients with alpha-1-antitrypsin deficiency or who have used certain intravenous drugs [21]. Recent investigations suggest that these subtypes, including both the lobular distribution as well as severity, may have different clinical associations [24,25]. For instance, while nearly all emphysema subtypes are associated with respiratory function, dyspnea, physical capacity and annual number of exacerbations in COPD, paraseptal emphysema may be a marker of more severe clinical manifestations of the disease [25]. Further CT based investigation has shown that there may be genetic and pathophysiologic reasons for such disease heterogeneity. For instance, emphysema matrix-metalloproteinases (MMPs), which are proteolytic enzymes that can break down the extracellular matrix, have been shown to be potentially important in the development of emphysema. While some MMPs, such as MMP-3 and MMP-10 are associated with most emphysema subtypes, others, such as MMP-3 and MMP-10 have been shown to be associated with both paraseptal emphysema and more severe centrilobular disease [26]. In addition, genome wide association studies have suggested that the each emphysema subtypes may have a somewhat unique set of genetic determinants [24]. Taken together, these findings suggest the importance of CT imaging for the understanding of both the clinical manifestations and pathophysiology of the emphysematous component of COPD, especially in terms of identifying specific phenotypes of the disease.

It is increasingly clear, in large part thanks to CT imaging based studies, that tobacco smoke exposure results in not only emphysematous but also interstitial changes in the lung, such as interstitial lung disease. In the extreme, such changes manifest radiologically and clinically as pulmonary fibrosis, but more subtle changes, often termed interstitial lung abnormalities (ILA) are highly prevalent in ever smokers [27-31] (figure 1b). Clinically, smokers with ILA tend to have less COPD, greater respiratory impairment, shorter 6 minute walk distances, and less emphysema as measured by standard densitometry [29,30,32,33]. Furthermore, these abnormalities are associated with a decline in lung function and are strongly predictive of all cause and respiratory specific mortality [31,34]. Finally, ILA have been shown to be associated with a specific single nucleotide polymorphism in the promoter region of the gene

encoding mucin 5B, which has been strongly linked to the presence of more advanced idiopathic pulmonary fibrosis [35,36]. These clinical and genetic associations do not necessarily imply that all ILA will progress to idiopathic pulmonary fibrosis, but rather highlight the potential role of CT in identifying those COPD patients at the greatest risk for adverse events as well as its importance for understanding the complexity of smoking related lung disease. Further work in this area could ultimately lead to the use of novel therapies such as anti-fibrotic agents in those patients with combined emphysema and early stage fibrosis.

The other lung compartment classically affected in COPD is the airways. While usually described as being the location of chronic bronchitis, the distal small airways are likely the site of airflow limitation for COPD more generally [37,38]. These airways are below the resolution of standard clinical CT imaging, however, quantitative CT analysis of more central airways in smokers has revealed that those with smaller airway caliber and thicker airway walls tend to have lower lung function and more frequent exacerbations [39-42]. While these findings are not yet clinically applicable, they may prove especially important in future phenotyping of COPD. For instance, recent work by Lange et al. has shown that some individuals with COPD developed airflow limitation in early adulthood followed by a slow decline in lung function rather than due to a rapid decline from a higher or more normal peak lung function, and it may be that those smokers who have a smaller airway caliber on CT in young adulthood are those more likely to develop COPD [43,44].

The last lung compartment with evident changes on the CT scans of smokers is the intraparenchymal pulmonary vasculature. Pruning of the vasculature and drop out in regions of severe emphysema have long been recognized on angiographic studies, and improved CT scan resolution has made it possible to detect and quantify such changes on CT [45,46]. These changes typically include pruning of the distal vasculature and dilation of the more central vessels [47-49]. While not yet clinically available, 3D analysis of the distal pulmonary vasculature has shown that pruning of the distal vessels is associated with increased respiratory symptoms, reduced exercise capacity, impaired diffusing capacity and multi-component predictors of mortality [47]. Further work is now ongoing to determine how metrics of pulmonary vascular morphology may be further integrated into clinical care to predict outcomes such as patient response to lung volume reduction. Of more immediate clinical importance is the appearance of the central vasculature on CT in patients with COPD. More specifically, Wells et al. has shown that pulmonary artery enlargement as defined by a ratio of the diameter of the pulmonary artery to the diameter of the aorta of greater than 1 is associated with an increased risk for severe respiratory exacerbations in patients with COPD, potentially identifying patients who will benefit the most from strategies to reduce exacerbation risk [49].

Advantages: CT imaging is almost universally available in clinical care settings. It provides high resolution insight into lung structure and there are a multitude of methods published on the standards of image interpretation.

Limitations: The greatest limitation to the broad utilization of CT imaging is ionizing radiation. There is an ongoing debate in the medical community about what is a safe level of such exposure. Because of this, CT manufacturers have invested a tremendous amount of resources to reduce the dose of radiation while maintaining or even improving image quality.

Clinical applications: CT is increasingly becoming a common component to the clinical care of patients with COPD. Despite the reason for image acquisition, the biomedical community is obliged to obtain as much data as possible from these images to improve our understanding of disease and improve patient care.

BODY COMPOSITION

Alterations in body composition frequently occur in COPD, contributing to increased morbidity and mortality. In fact, up to 25% of patients with COPD have a significant loss of skeletal muscle, which has been associated with impaired exercise performance and increased mortality independent of the primary lung impairment [50,51]. However, in overweight and obese subjects, low muscle mass may not be recognized without assessment of body composition [52]. Furthermore, Beijers et al. recently demonstrated in normal weight COPD patients a high prevalence of low muscle mass combined with abdominal obesity. Notably, these patients were characterized by an increased cardiometabolic risk [53]. Clinically available imaging modalities to visualize body composition include dual-energy X-ray absorptiometry (DXA) and CT.

DXA imaging

Contribution to phenotyping: Originally designed for osteoporosis assessment, DXA can also be utilized to measure body composition (figure 1e). The underlying concept for DXA estimation of body composition is based on assumptions regarding the difference in chemical constituents. Briefly, DXA utilizes the variable absorption or attenuation of high and low energy x-ray photons to estimate the fraction of tissue occupied by fat and fat free mass (FM and FFM respectively) [54,55]. While the details of how these measurements are made are beyond the scope of this review, it is important to note that increased tissue thickness is associated with a greater attenuation of low energy photons than high energy photons regardless of the tissue composition, the effect of which is a tendency for the underestimation of FM in obese subjects [56].

Recently, the European Respiratory Society taskforce on nutritional assessment and therapy in COPD has designated DXA as the most appropriate tool for combined screening of osteoporosis,

sarcopenia and adipose tissue redistribution [57]. While two compartment screening methods such as bioelectrical impedance analysis and skinfold thickness can discriminate whole body FM from FFM, DXA is able to analyse tissue depletion and redistribution at regional levels. This is particularly useful in COPD, in which appendicular FFM has been found to be more predictive for exercise performance than whole body FFM [52]. Moreover, DXA analyses have also provided insight into some of the clinical complexity of COPD. For example, despite similar decreases appendicular FFM, whole body and trunk FFM reductions have been found to be more altered in COPD patients with CT based emphysema compared to those without [58,59].

In addition to differences in truncal and whole body distribution, recent advances in DXA imaging have allowed for the differentiation between and the ratio of subcutaneous adipose tissue (SAT) and visceral adipose tissue (VAT) [60-62]. SAT and VAT differ in metabolic and inflammatory characteristics and the corresponding metabolic risk profile [63-65]. Smoking cigarettes is a main cause of both COPD and cardiac disease but other causes may increase the risk. For instance, cardiac disease in COPD associates with VAT [66,67]. DXA measurements of SAT and VAT have thus far only been validated in patients without COPD. Other limitations of DXA include the insensitivity of appendicular FFM for the detection of small loss of muscle fibre CSA, which may limit its role in monitoring subtle changes after intervention [68].

Advantages: DXA is particularly attractive due to low ionizing radiation, which is comparable to one day of normal background radiation [69]. With approximately 10 minutes scan time, DXA is very time-efficient and convenient. It enables non-invasive insight into body composition and its potential metabolic associations.

Limitations: Although the errors are relatively small, one has to keep in mind that the results may be influenced by conditions in which the ratio of extracellular water and intracellular water varies (e.g., severe malnutrition, oedema, diuretics, ageing) [70]. Furthermore, this technique only provides 2D projections of the body, thereby being unable to gain information about muscle groups or quantification of fat depots in the muscle. Regarding DXA-based VAT analysis more validation studies are warranted, as there seems to be a tendency to overestimation of VAT [60,62].

Clinical applicability: DXA is used in clinical routine care for evaluation of osteoporosis in COPD.

CT imaging

Contribution to phenotyping: As discussed above, CT imaging provides a wealth of information regarding lung structure and its implications for lung function in patients with COPD. Extrapulmonary findings on CT in COPD are of great scientific utility as well. Similar to DXA imaging, CT can also be utilized beyond its "field" to measure body composition in addition to the more traditional lung measures discussed above. This is done by measuring muscle cross sectional area (CSA) on CT imaging, and while the standard site to measure CSA is the third lumbar vertebra (L3) this is typically outside the field of view of clinically acquired CT imaging in patients with chest diseases [51,53,71,72] (figure 1d).

Therefore, other levels, such as L1, have been evaluated. Although single slice CSA L3 and L1 result in significantly different whole body estimates of muscle mass, the latter may be useful to detect changes during the disease course or after intervention [72]. Other sites that have been evaluated include the pectoralis muscles, where muscle CSA assessed by CT has been correlated with both objective and subjective measures of COPD severity, and mid-thigh muscle CSA which has been found to be strongly related to increased mortality risk [11,73]. Although such measures are not yet routinely clinically available, advances in CT segmentation technology that allows the automated detection of specific radiographic features may soon enable their more routine use.

CT is also able to quantify and evaluate the distribution of fat. For instance, CT-based analysis at the L4-L5 level revealed significantly increased VAT in elderly with obstructive lung disease than non-obstructive subjects despite a comparable SAT and BMI. This adipose tissue distribution also correlated to higher interleukin 6 levels in elderly with obstructive lung disease [74], suggesting that VAT might contribute to increased plasma IL-6, which was also shown to be a strong predictor of all-cause and respiratory mortality.

Finally, patients with chronic diseases such as COPD are known to have ectopic depots of fat beyond the major subcutaneous and visceral fat storage locations, such as intramuscular adipose tissue (IMAT). While, data concerning muscle lipid content in elderly and patients with muscle wasting disease have been initially relied on muscle biopsy studies, improved resolution of CT scanners makes it now possible to non-invasively quantify IMAT [75-77]. CT derived data concerning IMAT have shown fat infiltration in intercostal muscles and mid-thigh muscle in normal to overweight COPD patients. This infiltration in intercostal muscles in particular is correlated with COPD severity and patients with high mid-thigh IMAT tended to have lower physical activity levels [78,79].

Advantages: An advantage of CT is that information about body composition can be collected at the time of routine clinical imaging.

Limitations: Muscle CSA and adipose tissue quantification may be subject to a multitude of factors such as manufacturer, slice thickness and pixel spacing, although their exact influence on body composition quantification is unknown. Furthermore, the radiation exposure of CT limits longitudinal research applications.

Clinical applicability: Although CT imaging is not independently indicated for the diagnosis and management of isolated COPD, images are widely available due to their implementation in excluding other underlying pathology. Measurements of muscle CSA and adipose tissue are not yet routinely performed but new automated techniques may enable their introduction into clinical care in the near future.

TISSUE METABOLISM

Metabolic alterations are not only reflected by body composition but also by underlying changes in tissue metabolism. 18F-fluorodeoxyglucose positron emission tomography (18F-FDG-PET) may improve understanding of biological processes of tissues.

18F-FDG-PET imaging

Contribution to phenotyping: While DXA and CT are informative in tissue mass quantification and distribution, they are limited in monitoring tissue metabolic activity. Standard PET imaging uses a radiolabelled glucose derivate fluordeoxyglucose (FDG) to give a 3D information on glucose metabolism in the body enabling evaluation of local or tissue specific metabolic features [80]. There is a wide range of classical and novel radiopharmaceuticals (tracers) evaluated in (pre)clinical and experimental settings using PET. However, here we focus on FDG-PET because of its universal availability in clinical care settings. Although PET imaging is not routinely used in the workup of COPD assessment, this method allowing quantification of tissue metabolism may be useful in scientific research to contribute to better understanding of different clinical phenotypes as illustrated below.

FDG reflects the metabolic rate of glucose, a process reinforced in inflamed tissue and this distinctive characteristic enables PET to evaluate adipose tissue activity in vivo. One study showed that COPD patients exhibited increased FDG uptake in VAT compared to subjects without COPD. Furthermore, FDG uptake in VAT was found to predict aortic wall FDG uptake, even when adjusted for sex, age, BMI and smoking (figure 1c) [81-83]. It is well known that smoking tobacco is the main risk factor for development of both COPD and vascular inflammatory processes. Nevertheless, these data suggest a possible contributing role of VAT in augmenting atherosclerotic processes in COPD [6]. Further research is however needed including histological confirmation of VAT biological activity.

Several studies confirmed the presence of metabolically active brown adipose tissue (BAT) in the neck and supraclavicular region, the mediastinum as well as along the spine in healthy adults using FDG-PET scanning and biopsy verification [84,85]. BAT activity is inversely correlated to body fat percentage in healthy adults [85]. In addition, in several BAT-associated pathological conditions, such as pheochromocytoma [86] and hyperthyreoidism [87], energy expenditure is increased. Recently, it has been shown from in vitro experiments that lactate stimulate browning

of adipocytes, mediated by intracellular redox modifications [88]. Given the hypermetabolic state observed in COPD in particular in the emphysematous phenotype [89], mitochondrial dysfunction [90,91] and early lactate acidosis during exercise [92], increased BAT activity may be involved in energy homeostasis in COPD. However, no clinical studies are yet available that assessed BAT activity in COPD.

Advantages: Whereas DXA and CT provide information about structures, PET is a unique imaging tool that provides molecular imaging information. Using FDG non-invasive quantitative information of metabolic activity of tissues in vivo is acquired.

Limitations: FDG-PET uses ionizing radiation, is not broadly available and is more expensive than the other modalities. It is important to mention that the radiation burden is lower than conventionally used whole body CT protocols. Additionally, factors influencing the amount of FDG uptake such as overexpression of glucose receptors, might induce false positive or false negative results [80].

Clinical applicability: Not integrated in standard clinical care of COPD patients



Figure 1. Imaging techniques. (a) Thoracic CT image of patient with emphysema; (b) thoracic CT image of patient with interstitial lung abnormalities; (c) PET image displaying FDG uptake in aortic wall [Image courtesy of Dr. Jan Bucerius]; (d) abdominal CT with different body composition compartments at L3; (e) DXA scan leg with different body composition compartments.

CONCLUSION

Over the past century, imaging has become critically important in the routine clinical care of patients and the technologies used have advanced dramatically. The use of new analytic techniques applied to traditional imaging modalities and the development of novel imaging techniques have greatly improved the clinical care and understanding of complex diseases like COPD. These approaches, including CT, DXA, PET or hybrid PET/CT already aid in the non-invasive characterization of COPD phenotypes and severity, and ongoing and future work has the potential to improve our understanding of and care for this disease and the patients it affects even more. We acknowledge that at least in the short term, some of the methods discussed above are limited to the research setting. They may improve phenotyping for clinical observational studies and clinical trials, and may contribute to better risk stratification and personalized medicine.

REFERENCES

- Hillas G, Perlikos F, Tsiligianni I, Tzanakis N. Managing comorbidities in COPD. Int J Chron Obstruct Pulmon Dis. 2015;10:95-109.
- Divo MJ, Casanova C, Marin JM, et al. COPD comorbidities network. Eur Respir J. 2015;46:640-650.
- Decramer M, Janssens W, Miravitlles M. Chronic obstructive pulmonary disease. Lancet. 2012;379:1341-1351.
- Muller NL, Staples CA, Miller RR, Abboud RT. "Density mask". An objective method to quantitate emphysema using computed tomography. Chest. 1988;94:782-787.
- Lehouck A, Boonen S, Decramer M, Janssens W. COPD, bone metabolism, and osteoporosis. Chest. 2011;139:648-657.
- Vanfleteren LE, Spruit MA, Groenen M, et al. Clusters of comorbidities based on validated objective measurements and systemic inflammation in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2013;187:728-735.
- Divo M, Cote C, de Torres JP, et al. Comorbidities and risk of mortality in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2012;186:155-161.
- Ash SY, Diaz AA. The role of imaging in the assessment of severe asthma. Curr Opin Pulm Med. 2017;23:97-102.
- Budoff MJ, Hamirani YS, Gao YL, et al. Measurement of thoracic bone mineral density with quantitative CT. Radiology. 2010;257:434-440.
- Williams MC, Murchison JT, Edwards LD, et al. Coronary artery calcification is increased in patients with COPD and associated with increased morbidity and mortality. Thorax. 2014;69:718-723.
- McDonald ML, Diaz AA, Ross JC, et al. Quantitative computed tomography measures of pectoralis muscle area and disease severity in chronic obstructive pulmonary disease. A cross-sectional study. Ann Am Thorac Soc. 2014;11:326-334.
- National Lung Screening Trial Research T, Aberle DR, Adams AM, et al. Reduced lung-cancer mortality with low-dose computed tomographic screening. The New England journal of medicine.365:395-409.
- Ash SY, Washko, G.R. Controversies in COPD: The Value of CT Scanning. European Respiratory Society Monograph. 2015;69:121-133.
- 14. Budd G. Remarks on emphysema of the lungs. Med Chir Trans. 1840;23:37-62.

- 15. Kerley P. Emphysema: (Section of Radiology). Proc R Soc Med. 1936;29:1307-1324.
- Fishman A, Martinez F, Naunheim K, et al. A randomized trial comparing lung-volume-reduction surgery with medical therapy for severe emphysema. N Engl J Med. 2003;348:2059-2073.
- Boutou AK, Zoumot Z, Nair A, et al. The Impact of Homogeneous Versus Heterogeneous Emphysema on Dynamic Hyperinflation in Patients With Severe COPD Assessed for Lung Volume Reduction. Copd. 2015;12:598-605.
- Storbeck B, Schroder TH, Oldigs M, Rabe KF, Weber C. Emphysema: Imaging for Endoscopic Lung Volume Reduction. Rofo. 2015;187:543-554.
- Reymond E, Jankowski A, Pison C, et al. Prediction of lobar collateral ventilation in 25 patients with severe emphysema by fissure analysis with CT. AJR American journal of roentgenology. 2013;201:W571-575.
- Ross JC, San Jose Estepar R, Kindlmann G, et al. Automatic lung lobe segmentation using particles, thin plate splines, and maximum a posteriori estimation. Medical image computing and computer-assisted intervention : MICCAI International Conference on Medical Image Computing and Computer-Assisted Intervention. 2010;13:163-171.
- Austin JH, Muller NL, Friedman PJ, et al. Glossary of terms for CT of the lungs: recommendations of the Nomenclature Committee of the Fleischner Society. Radiology. 1996;200:327-331.
- Heard BE, Khatchatourov V, Otto H, Putov NV, Sobin L. The morphology of emphysema, chronic bronchitis, and bronchiectasis: definition, nomenclature, and classification. J Clin Pathol. 1979;32:882-892.
- Leopold JG, Gough J. The centrilobular form of hypertrophic emphysema and its relation to chronic bronchitis. Thorax. 1957;12:219-235.
- Castaldi PJ, Cho MH, San Jose Estepar R, et al. Genome-wide association identifies regulatory Loci associated with distinct local histogram emphysema patterns. American journal of respiratory and critical care medicine. 2014;190:399-409.
- Castaldi PJ, San Jose Estepar R, Mendoza CS, et al. Distinct quantitative computed tomography emphysema patterns are associated with physiology and function in smokers. Am J Respir Crit Care Med. 2013;188:1083-1090.
- 26. Ostridge K, Williams N, Kim V, et al. Distinct em-

physema subtypes defined by quantitative CT analysis are associated with specific pulmonary matrix metalloproteinases. Respiratory research. 2016;17:92.

- Baumgartner KB, Samet JM, Stidley CA, Colby TV, Waldron JA. Cigarette smoking: a risk factor for idiopathic pulmonary fibrosis. Am J Respir Crit Care Med. 1997;155:242-248.
- Doyle TJ, Hunninghake GM, Rosas IO. Subclinical interstitial lung disease: why you should care. Am J Respir Crit Care Med. 2012;185:1147-1153.
- Washko GR, Hunninghake GM, Fernandez IE, et al. Lung volumes and emphysema in smokers with interstitial lung abnormalities. N Engl J Med. 2011;364:897-906.
- Washko GR, Lynch DA, Matsuoka S, et al. Identification of early interstitial lung disease in smokers from the COPDGene Study. Acad Radiol. 2010;17:48-53.
- Putman RK, Hatabu H, Araki T, et al. Association Between Interstitial Lung Abnormalities and All-Cause Mortality. JAMA. 2016;315:672-681.
- 32. Doyle TJ, Dellaripa PF, Batra K, et al. Functional impact of a spectrum of interstitial lung abnormalities in rheumatoid arthritis. Chest. 2014;146:41-50.
- Doyle TJ, Washko GR, Fernandez IE, et al. Interstitial lung abnormalities and reduced exercise capacity. Am J Respir Crit Care Med. 2012;185:756-762.
- Araki T, Putman RK, Hatabu H, et al. Development and Progression of Interstitial Lung Abnormalities in the Framingham Heart Study. Am J Respir Crit Care Med. 2016;194:1514-1522.
- Hunninghake GM, Hatabu H, Okajima Y, et al. MUC5B promoter polymorphism and interstitial lung abnormalities. The New England journal of medicine. 2013;368:2192-2200.
- Seibold MA, Wise AL, Speer MC, et al. A common MUC5B promoter polymorphism and pulmonary fibrosis. The New England journal of medicine. 2011;364:1503-1512.
- 37. Blue bloater: pink puffer. Br Med J. 1968;2:677.
- Hogg JC, Macklem PT, Thurlbeck WM. Site and nature of airway obstruction in chronic obstructive lung disease. N Engl J Med. 1968;278:1355-1360.
- Nakano Y, Muro S, Sakai H, et al. Computed tomographic measurements of airway dimensions and emphysema in smokers. Correlation with lung function. Am J Respir Crit Care Med. 2000;162:1102-1108.
- 40. Niimi A, Matsumoto H, Amitani R, et al. Airway wall

thickness in asthma assessed by computed tomography. Relation to clinical indices. Am J Respir Crit Care Med. 2000;162:1518-1523.

- 41. Van Tho N, Ogawa E, Trang le TH, et al. A mixed phenotype of airway wall thickening and emphysema is associated with dyspnea and hospitalization for chronic obstructive pulmonary disease. Ann Am Thorac Soc. 2015;12:988-996.
- Han MK, Kazerooni EA, Lynch DA, et al. Chronic obstructive pulmonary disease exacerbations in the COPDGene study: associated radiologic phenotypes. Radiology. 2011;261:274-282.
- 43. Lange P, Celli B, Agusti A, et al. Lung-Function Trajectories Leading to Chronic Obstructive Pulmonary Disease. The New England journal of medicine. 2015;373:111-122.
- Ash SY, Washko GR. New Lung Imaging Findings in Chronic Respiratory Diseases. BRN Rev. 2017;3:121-135.
- Scarrow GD. The pulmonary angiogram in chronic bronchitis and emphysema. Proc R Soc Med. 1965;58:684-687.
- Jacobson G, Turner AF, Balchum OJ, Jung R. Vascular changes in pulmonary emphysema. The radiologic evaluation by selective and peripheral pulmonary wedge angiography. Am J Roentgenol Radium Ther Nucl Med. 1967;100:374-396.
- Estepar RS, Kinney GL, Black-Shinn JL, et al. Computed tomographic measures of pulmonary vascular morphology in smokers and their clinical implications. Am J Respir Crit Care Med. 2013;188:231-239.
- 48. Wells JM, Iyer AS, Rahaghi FN, et al. Pulmonary artery enlargement is associated with right ventricular dysfunction and loss of blood volume in small pulmonary vessels in chronic obstructive pulmonary disease. Circ Cardiovasc Imaging. 2015;8.
- Wells JM, Washko GR, Han MK, et al. Pulmonary arterial enlargement and acute exacerbations of COPD. N Engl J Med. 2012;367:913-921.
- Shrikrishna D, Patel M, Tanner RJ, et al. Quadriceps wasting and physical inactivity in patients with COPD. Eur Respir J. 2012;40:1115-1122.
- Swallow EB, Reyes D, Hopkinson NS, et al. Quadriceps strength predicts mortality in patients with moderate to severe chronic obstructive pulmonary disease. Thorax. 2007;62:115-120.
- van de Bool C, Rutten EP, Franssen FM, Wouters EF, Schols AM. Antagonistic implications of sarcopenia and abdominal obesity on physical performance

in COPD. Eur Respir J. 2015;46:336-345.

- 53. Beijers R, van de Bool C, van den Borst B, Franssen FME, Wouters EFM, Schols A. Normal Weight but Low Muscle Mass and Abdominally Obese: Implications for the Cardiometabolic Risk Profile in Chronic Obstructive Pulmonary Disease. J Am Med Dir Assoc. 2017;18:533-538.
- Jebb SA. Measurement of soft tissue composition by dual energy X-ray absorptiometry. Br J Nutr. 1997;77:151-163.
- Pietrobelli A, Formica C, Wang Z, Heymsfield SB. Dual-energy X-ray absorptiometry body composition model: review of physical concepts. Am J Physiol. 1996;271:E941-951.
- Goodsitt MM. Evaluation of a new set of calibration standards for the measurement of fat content via DPA and DXA. Med Phys. 1992;19:35-44.
- Schols AM, Ferreira IM, Franssen FM, et al. Nutritional assessment and therapy in COPD: a European Respiratory Society statement. Eur Respir J. 2014;44:1504-1520.
- Engelen MP, Schols AM, Does JD, Wouters EF. Skeletal muscle weakness is associated with wasting of extremity fat-free mass but not with airflow obstruction in patients with chronic obstructive pulmonary disease. Am J Clin Nutr. 2000;71:733-738.
- Engelen MP, Schols AM, Lamers RJ, Wouters EF. Different patterns of chronic tissue wasting among patients with chronic obstructive pulmonary disease. Clin Nutr. 1999;18:275-280.
- Kaul S, Rothney MP, Peters DM, et al. Dual-energy X-ray absorptiometry for quantification of visceral fat. Obesity (Silver Spring). 2012;20:1313-1318.
- Micklesfield LK, Goedecke JH, Punyanitya M, Wilson KE, Kelly TL. Dual-energy X-ray performs as well as clinical computed tomography for the measurement of visceral fat. Obesity (Silver Spring). 2012;20:1109-1114.
- Lin H, Yan H, Rao S, et al. Quantification of visceral adipose tissue using lunar dual-energy X-ray absorptiometry in Asian Chinese. Obesity (Silver Spring). 2013;21:2112-2117.
- Sin DD, Jones RL, Man SF. Obesity is a risk factor for dyspnea but not for airflow obstruction. Archives of internal medicine. 2002;162:1477-1481.
- Fimognari FL, Scarlata S, Pastorelli R, Antonelli-Incalzi R. Visceral obesity and different phenotypes of COPD. American journal of respiratory and critical care medicine. 2009;180:192-193; author reply

193.

- Leone N, Courbon D, Thomas F, et al. Lung function impairment and metabolic syndrome: the critical role of abdominal obesity. American journal of respiratory and critical care medicine. 2009;179:509-516.
- Lima MM, Pareja JC, Alegre SM, et al. Visceral fat resection in humans: effect on insulin sensitivity, beta-cell function, adipokines, and inflammatory markers. Obesity (Silver Spring). 2013;21:E182-189.
- Diaz AA, Young TP, Kurugol S, et al. Abdominal Visceral Adipose Tissue is Associated with Myocardial Infarction in Patients with COPD. Chronic Obstr Pulm Dis (Miami). 2015;2:8-16.
- 68. Op den Kamp CM, Langen RC, Snepvangers FJ, et al. Nuclear transcription factor kappa B activation and protein turnover adaptations in skeletal muscle of patients with progressive stages of lung cancer cachexia. Am J Clin Nutr. 2013;98:738-748.
- Blake GM, Naeem M, Boutros M. Comparison of effective dose to children and adults from dual X-ray absorptiometry examinations. Bone. 2006;38:935-942.
- Pietrobelli A, Wang Z, Formica C, Heymsfield SB. Dual-energy X-ray absorptiometry: fat estimation errors due to variation in soft tissue hydration. Am J Physiol. 1998;274:E808-816.
- Shen W, Punyanitya M, Wang Z, et al. Total body skeletal muscle and adipose tissue volumes: estimation from a single abdominal cross-sectional image. J Appl Physiol (1985). 2004;97:2333-2338.
- Schweitzer L, Geisler C, Pourhassan M, et al. What is the best reference site for a single MRI slice to assess whole-body skeletal muscle and adipose tissue volumes in healthy adults? Am J Clin Nutr. 2015;102:58-65.
- Marquis K, Debigare R, Lacasse Y, et al. Midthigh muscle cross-sectional area is a better predictor of mortality than body mass index in patients with chronic obstructive pulmonary disease. American journal of respiratory and critical care medicine. 2002;166:809-813.
- van den Borst B, Gosker HR, Koster A, et al. The influence of abdominal visceral fat on inflammatory pathways and mortality risk in obstructive lung disease. Am J Clin Nutr. 2012;96:516-526.
- Forsberg AM, Nilsson E, Werneman J, Bergstrom J, Hultman E. Muscle composition in relation to age and sex. Clin Sci (Lond). 1991;81:249-256.
- Jones DA, Round JM, Edwards RH, Grindwood SR, Tofts PS. Size and composition of the calf and quadriceps muscles in Duchenne muscular dystrophy. A tomographic and histochemical study. J Neurol Sci. 1983;60:307-322.
- Goodpaster BH, Kelley DE, Thaete FL, He J, Ross R. Skeletal muscle attenuation determined by computed tomography is associated with skeletal muscle lipid content. J Appl Physiol (1985). 2000;89:104-110.
- Park MJ, Cho JM, Jeon KN, et al. Mass and fat infiltration of intercostal muscles measured by CT histogram analysis and their correlations with COPD severity. Acad Radiol. 2014;21:711-717.
- Maddocks M, Shrikrishna D, Vitoriano S, et al. Skeletal muscle adiposity is associated with physical activity, exercise capacity and fibre shift in COPD. The European respiratory journal. 2014;44:1188-1198.
- Basu S, Kwee TC, Surti S, Akin EA, Yoo D, Alavi A. Fundamentals of PET and PET/CT imaging. Ann N Y Acad Sci. 2011;1228:1-18.
- Vanfleteren LE, van Meerendonk AM, Franssen FM, et al. A possible link between increased metabolic activity of fat tissue and aortic wall inflammation in subjects with COPD. A retrospective 18F-FDG-PET/CT pilot study. Respiratory medicine. 2014;108:883-890.
- Bucerius J, Vijgen GH, Brans B, et al. Impact of bariatric surgery on carotid artery inflammation and the metabolic activity in different adipose tissues. Medicine (Baltimore). 2015;94:e725.
- Bucerius J, Mani V, Wong S, et al. Arterial and fat tissue inflammation are highly correlated: a prospective 18F-FDG PET/CT study. Eur J Nucl Med Mol Imaging. 2014;41:934-945.
- Cypess AM, Lehman S, Williams G, et al. Identification and importance of brown adipose tissue in adult humans. N Engl J Med. 2009;360:1509-1517.
- van Marken Lichtenbelt WD, Vanhommerig JW, Smulders NM, et al. Cold-activated brown adipose tissue in healthy men. N Engl J Med. 2009;360:1500-1508.
- Wang Q, Zhang M, Ning G, et al. Brown adipose tissue in humans is activated by elevated plasma catecholamines levels and is inversely related to central obesity. PLoS One. 2011;6:e21006.
- Lopez M, Varela L, Vazquez MJ, et al. Hypothalamic AMPK and fatty acid metabolism mediate thyroid regulation of energy balance. Nat Med.

2010;16:1001-1008.

- Carriere A, Jeanson Y, Berger-Muller S, et al. Browning of white adipose cells by intermediate metabolites: an adaptive mechanism to alleviate redox pressure. Diabetes. 2014;63:3253-3265.
- Schols AM, Fredrix EW, Soeters PB, Westerterp KR, Wouters EF. Resting energy expenditure in patients with chronic obstructive pulmonary disease. Am J Clin Nutr. 1991;54:983-987.
- Gifford JR, Trinity JD, Layec G, et al. Quadriceps exercise intolerance in patients with chronic obstructive pulmonary disease: the potential role of altered skeletal muscle mitochondrial respiration. J Appl Physiol (1985). 2015;119:882-888.
- Yue L, Yao H. Mitochondrial dysfunction in inflammatory responses and cellular senescence: pathogenesis and pharmacological targets for chronic lung diseases. Br J Pharmacol. 2016;173:2305-2318.
- 92. Engelen MP, Schols AM, Does JD, Gosker HR, Deutz NE, Wouters EF. Exercise-induced lactate increase in relation to muscle substrates in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2000;162:1697-1704.

Imaging to understand disease complexity



3

Cross-sectional and longitudinal assessment of muscle from regular chest computed tomography scans: L1 and pectoralis muscle compared to L3 as reference in non-small cell lung cancer

Karin J.C. Sanders, Juliette H.R.J. Degens, Anne-Marie C. Dingemans, Annemie M.W.J. Schols.

International Journal of Chronic Obstructive Pulmonary Disease. 2019;14:781-89.

ABSTRACT

Background

Computed tomography (CT) is increasingly used in clinical research for single-slice assessment of muscle mass to correlate with clinical outcome and evaluate treatment efficacy. The third lumbar level (L3) is considered as reference for muscle, but chest scans generally do not reach beyond the first lumbar level (L1). This study investigates if pectoralis muscle and L1 are appropriate alternatives for L3.

Methods

CT scans of 115 stage IV non-small cell lung cancer patients were analysed before and during tumour therapy. Skeletal muscle assessed at pectoralis and L1 muscle was compared to L3 at baseline. Furthermore, the prognostic significance of changes in muscle mass determined at different locations was investigated.

Results

Pearson's correlation coefficient between skeletal muscle at L3 and L1 was stronger (r=0.90, P<0.001) than between L3 and pectoralis muscle (r=0.71, P<0.001). Cox regression analysis revealed that L3 (HR 0.943, 95% CI: 0.92-0.97, p<0.001) and L1 muscle loss (HR 0.954, 95% CI: 0.93-0.98, p<0.001) predicted overall survival, whereas pectoralis muscle loss did not.

Conclusion

L1 is a better alternative than pectoralis muscle to substitute L3 for analysis of muscle mass from regular chest CT scans.

INTRODUCTION

It is now widely recognized that loss of skeletal muscle mass adversely impacts clinical outcome and increases healthcare utilization in patients facing cancer or a chronic disease like Chronic Obstructive Pulmonary Disease (COPD) [1-7]. Since low muscle mass is not restricted to those with lean stature [5], there is a growing interest to assess muscle mass with use of techniques that are easily applicable in the clinic or are already being used in regular clinical care for other diagnostic purposes. Whole body skeletal muscle mass can be estimated in clinical settings by measuring fat-free mass using bioimpedance analysis and dual-energy X-ray absorptiometry [8]. Since these modalities are not typically incorporated in routine clinical care, other strategies to evaluate muscle mass have been employed. Computed tomography (CT) is increasingly used as research tool for local quantification of skeletal muscle. These medical images provide the opportunity to obtain information about skeletal muscle quantity and quality. CT-derived skeletal muscle at the third lumbar level (L3) is considered as reference because muscle crosssectional area (CSA) at L3 is linearly related to whole-body muscle mass assessed by magnetic resonance imaging [9]. As most chest CT scans do not reach beyond the first lumbar level (L1), attempts have been made to assess muscles at other levels. For this reason, in COPD research pectoralis muscle has been proposed as alternative [10-12]. However it is unclear whether this pectoralis muscle reflects a reliable representation of whole body muscle area, as to date no studies compared the method with appropriate reference methods. The first aim of this study is therefore to investigate the association between pectoralis muscle CSA and L3 muscle CSA.

The advantage of L3 and L1 as opposed to pectoralis is that besides skeletal muscle, adipose tissue CSA and distribution can be quantified. Recently, skeletal muscle cut-off values for sarcopenia at L1 have been proposed based on CT analysis in a healthy American population. The authors reported good correlations regarding skeletal muscle CSA at L1 compared to L3. However, pectoralis muscle was not assessed and data comparing adipose tissue between both regions are lacking [13]. The second aim of the study is therefore to compare skeletal muscle and adipose tissue analysis between L1 and L3.

The added value of CT-derived analysis of body composition may not only be to phenotype patients with respiratory disease but even more to detect treatment induced subtle changes in muscle CSA that may adversely affect disease progression or treatment outcome. The third aim of the study is to evaluate the association between changes in skeletal muscle CSA assessed from pectoralis muscle, L1 and L3 after tumour treatment to overall survival.

METHODS

Study subjects

To address the research questions, CT scans derived from a randomized clinical trial involving patients with non-small cell lung cancer (NSCLC) were analysed. This multicentre randomized phase II trial investigated nitroglycerin patches added to paclitaxel-carboplatin-bevacizumab in 223 therapy-naive patients with stage IV non-squamous NSCLC. As part of standard work-up, patients received a 18F-fluorodeoxyglucose positron emission tomography scan combined with CT scan at baseline. The study protocol pre-specified a second 18F-fluorodeoxyglucose positron emission tomography/CT between day 22 and 24 (after second chemotherapy infusion). The study was performed in accordance with the provisions of the Declaration of Helsinki. The study was approved by the medical ethical committee of University Medical Centre Groningen in the Netherlands (METC 2010.241) and registered at clinicaltrial.gov (NCT01171170). All patients provided written informed consent before performing any study-related activities. Adding nitroglycerin to first-line carboplatin-paclitaxel-bevacizumab did not improve progression-free survival and overall survival in this cohort. The detailed methodology of this study has been published previously [14].

Image analysis

CT scans were made at baseline and after the second chemotherapy infusion for response assessment. Cross-sectional measurements of skeletal muscle areas (CSA in cm²) were made on transverse images. Three images were selected for each patient. The pectoralis muscle slice was selected by scrolling towards the apex of the lungs and identifying the first axial image above the aortic arch. Bilaterally, the area of pectoralis major and minor muscles was measured. For slices at lumbar levels, the first image at the third and first lumbar level with both vertebral transverse processes clearly visible was used in the analysis. The measurements included the psoas, erector spinae, quadratus lumborum, transversus abdominis, external and internal oblique, and rectus abdominis muscles.

Additionally, adipose tissue compartments including subcutaneous adipose tissue (SAT) CSA and visceral adipose tissue (VAT) CSA were analysed at L3 and L1. Proportional changes in CSA between CT scans were normalized for the time interval between scans. We therefore divided the proportional CSA change by the number of days between scans and multiplied by 100 days to standardise for all patients (percentage change per 100 days).

All analyses were performed with Slice-O-Matic software v5.0 (Tomovision, Montreal, Canada). CSA of these structures were quantified on the basis of pre-established thresholds of Hounsfield units (skeletal muscle-29 to 150, IMAT-190 to-30, SAT-190 to-30, and VAT-150 to-50). Boundaries were corrected manually when necessary.

Statistical analyses

Patients were included if CT scans both at baseline and follow-up were available and contained images of pectoralis, L1 and L3. Descriptive statistics of demographic and clinical variables were obtained. Means and standard deviations (SD) were provided for continuous normally distributed variables, median (range) for continuous not-normally distributed variables and percentages were shown for categorical variables. Comparisons within groups were performed with paired t-test or Wilcoxon signed rank test.

To measure the strength of the linear association of different assay methods (pectoralis, L1 and L3 at baseline and during follow-up) Pearson's correlation coefficient (r) was used. Additionally, a Bland-Altman plot was drawn to investigate the existence of systematic bias and to identify possible outliers. Differences between assay methods are expressed as percentage of the values on the Y-axis. L3 was used as reference method and therefore plotted on the X-axis. If the mean value of the difference between assay methods is different from 0, it indicates a systematic difference. If there is proportional bias, the Bland-Altman plot shows whether or not this bias is constant for all the measures of CSA's.

To evaluate if pectoralis, L1 and L3 measured the same construct (i.e. internal consistency) Cronbach's coefficient alpha was calculated. Cronbach's coefficient alpha is expressed as a number between 0 and 1, and values closer to 1 indicate good reliability among assay methods.

Kaplan meier survival analysis was performed to assess the contribution of skeletal muscle loss measured at different slices to overall survival. Overall survival was defined as the interval from randomization to death from any cause. A mean coefficient of variation between observers of 1.3% for skeletal muscle CSA in a random sample of 15 patients was observed, which is in line with the variation of 0-2% in other studies [5,15]. Therefore, a measurement error of 1.3% was adopted for L1 and L3. Changes of equal or larger than-1.3% were considered as 'loss of tissue', while changes of smaller than-1.3% were considered 'maintenance of tissue'. A variation regarding pectoralis muscle is not described in literature, and therefore the median was chosen as cut-off. The hazard ratio indicates the effect of increased muscle mass on overall survival. In addition, multivariate Cox regression analysis was performed with change in muscle CSA at the level of pectoralis muscle, L1 and L3, gender, age and Charlson comorbidity index as independent variables. The WHO performance score was between 0 and 1 in the majority of the patients and therefore not included in the regression analysis.

All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, version 24.0, IBM, Armonk, NY). Results with two-sided exact p values (<0.05) were considered statistically significant.

RESULTS

Patients and characteristics

In total, 223 patients were enrolled in the randomized controlled trial. One hundred and three patients were excluded due to unavailability of one or both CT scans, three patients were excluded because pectoralis muscle or L3 was not evaluable, one patient was excluded due to lacking OS data and one was excluded for insufficient quality of the scans. After exclusion, CT scans from 115 patients were eligible. Sixty-four patients (56%) were male, with mean age of 61 years and body mass index (mean \pm SD) of 25.1 \pm 4.2 kg/m². Patient characteristics, treatment arm, Charlson comorbidity index and overall survival were not different between patients included (n=115) and patients excluded (n=108) (data not shown).

Skeletal muscle

Mean \pm SD skeletal muscle CSA (in cm²) of pectoralis, L1 and L3 for pre-chemotherapy scans were respectively 36.0 ± 10.1 (figure 1, table 1), 116.1 ± 24.5 , and 134.8 ± 28.0 , respectively. Baseline r between pectoralis and L3 was 0.71 (p<0.001) (supplemental figure 1a). The associating Bland-Altman plot shows that mean difference between pectoralis muscle and L3 muscle was $116.3 \pm 13.6\%$, which was significantly different from 0 (p<0.001). The plot shows that the bias is relatively constant (supplemental figure 1b). Cronbach's alpha was 0.622.

Correlations, Bland-Altman plots and Cronbach's alpha were also performed for postchemotherapy scans, showing similar results (data not shown).



Figure 1. Skeletal muscle area on transverse CT images at (a) pectoralis, (b) first lumbar level and (c) third lumbar level.

	Baseline (cm²)	Follow-up (cm²)	Delta absolute (cm²/100 days)	Delta relative (%/100 days)	P value
Pectoralis muscle	36.0 ± 10.1	33.9 ± 9.8	-4.5 ± 12.9	-4.3 ± 13.3	<0.001
L1 muscle	116.1 ± 24.5	112.8 ± 22.7	-5.7 ± 20.1	-2.3 ± 7.7	<0.001
L3 muscle	134.8 ± 28.0	130.6 ± 27.3	-7.6 ± 18.8	-2.9 ± 6.7	<0.001

Table 1. Computed Tomography measurements	s ot	muscle	2
--	------	--------	---

Data are expressed as mean ± standard deviation. Abbreviations: L1: first lumbar vertebra; L3: third lumbar vertebra.

Baseline Pearson's r between L1 and L3 was 0.90 (p<0.001) (supplemental figure 1c). The corresponding Bland-Altman plot shows that mean difference between L1 and L3 measuring muscle was $14.9 \pm 9.0\%$, which was significantly different from 0 (p<0.001) (supplemental figure 1d). The plot reveals that there is a bias which is almost constant for all measured CSA's. The muscle CSA of L3 is larger than L1. Cronbach's alpha was 0.859, indicating a high internal consistency of skeletal muscle CSA measurement at L1 opposed to L3.

Adipose tissue

Mean \pm SD SAT CSA (in cm²) of L1 and L3 for pre-chemotherapy scans were respectively 107.0 \pm 73.5 and 160.8 \pm 85.1 (table 2).

	Baseline (cm²)	Follow-up (cm²)	Delta absolute (cm²/100 days)	Delta relative (%/100 days)	p value
L1 SAT	107.0 ± 73.5	104.5 ± 73.5	-3.6 ± 33.5	-1.4 ± 25.5	0.076
L3 SAT	160.8 ± 85.1	155.3 ± 84.9	-9.6 ± 61.8	-2.1 ± 19.9	0.021
L1 VAT	100.1 ± 65.5	100.1 ± 69.8	-1.3 ± 78.1	4.9 ± 33.2	0.979
L3 VAT	106.9 ± 70.4	104.8 ± 70.5	-5.5 ± 56.9	2.6 ± 29.0	0.386

Table 2. Computed Tomography measurements of adipose tissue compartments.

Data are expressed as mean ± standard deviation. Abbreviations: L1: first lumbar vertebra; L3: third lumbar vertebra; SAT: subcutaneous adipose tissue; VAT: visceral adipose tissue.

Baseline r was 0.93 (p<0.001) (supplemental figure 2a). The corresponding Bland-Altman plot shows that mean difference between L1 and L3 measuring SAT was 48.2 \pm 26.5%, which was significantly different from 0 (p<0.001). The plot shows that the bias is not constant, as the difference is proportional to the magnitude of L3 muscle (supplemental figure 2b). Cronbach's alpha was 0.959, indicating a high internal consistency of SAT CSA measurement at L1 compared to that at L3.

Mean \pm SD VAT CSA (in cm²) of L1 and L3 for pre-chemotherapy scans were respectively 100.1 \pm 65.5 and 106.9 \pm 70.4 (table 2). Baseline r was 0.90 (p<0.001) (supplemental figure 2c). The corresponding Bland-Altman plot shows that mean difference between L1 and L3 measuring SAT was 5.4 \pm 31.5%, which was significantly different from 0 (p<0.001). The plot shows that the bias is almost constant, with the exception of very low values of L3 VAT. (supplemental figure 2d). Cronbach's alpha was 0.940, indicating a high internal consistency of VAT CSA measurement at L1 opposed to L3. Correlations, Bland-Altman plots and Cronbach's alpha were also performed for post-chemotherapy scans, showing similar results.

Longitudinal changes

Table 1 and 2 display changes in body composition during the course of chemotherapy. Within two cycles of chemotherapy, muscle CSA significantly declined at the level of pectoralis, L1 and

L3. SAT CSA at L3 decreased respectively by-2.1%/100 days \pm 19.9 (p=0.021), while no significant changes in SAT CSA at L1 were observed. No changes were observed with respect to VAT.

Survival analysis

Kaplan-Meier survival analysis showed that skeletal muscle loss at L1 (HR 1.6, 95% CI: 1.08-2.36, p=0.020) and skeletal muscle loss at L3 (HR 1.9, 95% CI: 1.27-2.83, p=0.002), but not pectoralis



Figure 2. Kaplan Meier overall survival curve for patients with loss of muscle mass compared to patients without loss of muscle mass at the level of (a) pectoralis, (b) L1 and (c) L3.

muscle loss (HR 1.2, 95% CI: 0.79-1.70, p=0.463), were associated with overall survival (figure 2). On multivariate analysis correcting for gender, age and comorbidities, skeletal muscle loss at L3 and L1 levels were associated with overall survival, while pectoralis muscle loss was not (table 3).

	В	HR (95% CI)	P value		В	HR (95% CI)	P value		В	HR (95% CI)	P value
Gender	0.44	1.6 (0.90-2.69)	0.112	Gender	0.48	1.6 (0.94-2.80)	0.085	Gender	0.53	1.7 (0.97-2.98)	0.064
Age	0.01	1.0 (0.98-1.04)	0.577	Age	0.01	1.0 (0.98-1.04)	0.621	Age	0.01	1.0 (0.98-1.04)	0.762
Charlson comor- bidity index	0.12	1.1 (0.96-1.33)	0.144	Charlson comor- bidity index	0.11	1.1 (0.95-1.31)	0.169	Charlson comor- bidity index	60.0	1.1 (0.93-1.29)	0.293
Delta L3 muscle	-0.05	0.9 (0.91-0.98)	0.005	Delta L1 muscle	-0.06	0.9 (0.91-0.98)	0.002	Delta Pectoralis muscle	-0.02	1.0 (0.96-1.01)	0.126

Table 3. Multivariate analysis for predictors of overall survival.

DISCUSSION

The main objective of this study was to investigate whether pectoralis muscle and L1 could be suitable alternatives for L3 to assess muscle CSA. Crosssectional and longitudinal comparison revealed that L1 is stronger associated to L3 than pectoralis muscle.

To date, several studies have investigated CT-derived muscle to predict clinical outcome. The majority of studies conducted in patients with cancer involve quantification of L3. However, in COPD, a singlemuscle approach has been used more frequently. Güerri et al. showed in 20 COPD patients that those with a history of more frequent exacerbations exhibited smaller intercostal muscles, independent of age, gender and body mass index. The authors failed to verify these results among other evaluated muscles including upper limb muscles and psoas muscle [16]. Furthermore, attempts have been made to derive whole body fat-free mass from CT-based pectoralis muscle area by comparing pectoralis muscle with fat-free mass assessed by bioelectrical impedance analysis (BIA), but as BIA is a double indirect measure of fat-free mass [17], this cannot be considered as appropriate reference method such as muscle mass derived by magnetic resonance imaging. In addition, the unadjusted correlation between pectoralis and BIA derived fat-free mass was low (R²=0.38) [18]. Also in other patient populations, similar attempts have been made to measure one muscle as representative for lumbar muscle CSA, including psoas muscle [19,20]. However, Rutten et al. found that change in psoas muscle area was not representative of total muscle area change [21]. Overall, the idea that an individual muscle as opposed to total lumbar muscle reflects whole body muscle mass or fat-free mass has therefore not yet been appropriately validated.

Our results are in line with a previously published study performed in 90 small-cell lung cancer patients reporting similar baseline correlations between L3 and L1 muscle and between L3 and pectoralis muscle. Longitudinal data and adipose tissue CSA were not assessed [10]. The advantage of lumbar assessment of muscle mass is that also adipose tissue can be assessed. Based on whole body magnetic resonance imaging, single slice adipose tissue quantification at different levels in the abdominal region (L4-5, 5 cm above and below) correlated well with total body adipose tissue [9]. Analysing adipose tissue at the level of L1 might be informative as normal-weight COPD patients exhibited more VAT compared to smoking controls despite similar SAT [22]. Additionally, the presence of excessive VAT contributed to increased plasma interleukin-6 and is associated with increased mortality in persons with obstructive lung disease [23], emphasizing the importance of body composition in chronic disease risk assessment.

The advantage of this study is that it included a well-defined randomized patient cohort. Limitations are that the CT scans were performed in different medical centres, which might have resulted in variation in acquisition and scanning procedures. However, the body composition data derived from pre- and post-chemotherapy scans were made in the same centre, which implies that it will not have a major impact on the outcome of this research. Furthermore, not all CT scans could be retrieved and therefore only a subset of the scans has been analysed. However, the data we presented here are very consistent. We therefore feel confident that this patient sample is representative for the whole study cohort.

In conclusion, the present study shows that L1 but not pectoralis muscle can substitute L3 to determine body composition from regular chest CT scans for cross-sectional and longitudinal purposes.

REFERENCES

- Schols AM, Broekhuizen R, Weling-Scheepers CA, Wouters EF. Body composition and mortality in chronic obstructive pulmonary disease. Am J Clin Nutr. 2005;82:53-59.
- Shoup R, Dalsky G, Warner S, et al. Body composition and health-related quality of life in patients with obstructive airways disease. Eur Respir J. 1997;10:1576-1580.
- Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. Am J Respir Crit Care Med. 2006;173:79-83.
- Decramer M, Gosselink R, Troosters T, Verschueren M, Evers G. Muscle weakness is related to utilization of health care resources in COPD patients. Eur Respir J. 1997;10:417-423.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008;9:629-635.
- Malietzis G, Currie AC, Athanasiou T, et al. Influence of body composition profile on outcomes following colorectal cancer surgery. Br J Surg. 2016;103:572-580.
- Shachar SS, Deal AM, Weinberg M, et al. Skeletal Muscle Measures as Predictors of Toxicity, Hospitalization, and Survival in Patients with Metastatic Breast Cancer Receiving Taxane-Based Chemotherapy. Clin Cancer Res. 2017;23:658-665.
- Rozenberg D, Martelli V, Vieira L, et al. Utilization of non-invasive imaging tools for assessment of peripheral skeletal muscle size and composition in chronic lung disease: A systematic review. Respir Med. 2017;131:125-134.
- Shen W, Punyanitya M, Wang Z, et al. Total body skeletal muscle and adipose tissue volumes: estimation from a single abdominal cross-sectional image. J Appl Physiol (1985). 2004;97:2333-2338.
- Diaz AA, Zhou L, Young TP, et al. Chest CT measures of muscle and adipose tissue in COPD: genderbased differences in content and in relationships with blood biomarkers. Acad Radiol. 2014;21:1255-1261.
- McDonald ML, Diaz AA, Ross JC, et al. Quantitative computed tomography measures of pectoralis muscle area and disease severity in chronic

obstructive pulmonary disease. A cross-sectional study. Ann Am Thorac Soc. 2014;11:326-334.

- Tanimura K, Sato S, Fuseya Y, et al. Quantitative Assessment of Erector Spinae Muscles in Patients with Chronic Obstructive Pulmonary Disease. Novel Chest Computed Tomography-derived Index for Prognosis. Ann Am Thorac Soc. 2016;13:334-341.
- Derstine BA, Holcombe SA, Ross BE, Wang NC, Su GL, Wang SC. Skeletal muscle cutoff values for sarcopenia diagnosis using T10 to L5 measurements in a healthy US population. Sci Rep. 2018;8:11369.
- Dingemans AM, Groen HJ, Herder GJ, et al. A randomized phase II study comparing paclitaxelcarboplatin-bevacizumab with or without nitroglycerin patches in patients with stage IV nonsquamous nonsmall-cell lung cancer: NVALT12 (NCT01171170)dagger. Ann Oncol. 2015;26:2286-2293.
- Martin L, Birdsell L, Macdonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol. 2013;31:1539-1547.
- Guerri R, Gayete A, Balcells E, et al. Mass of intercostal muscles associates with risk of multiple exacerbations in COPD. Respir Med. 2010;104:378-388.
- Lukaski HC, Johnson PE, Bolonchuk WW, Lykken GI. Assessment of fat-free mass using bioelectrical impedance measurements of the human body. Am J Clin Nutr. 1985;41:810-817.
- McDonald MN, Diaz AA, Rutten E, et al. Chest computed tomography-derived low fat-free mass index and mortality in COPD. Eur Respir J. 2017;50.
- Amini N, Spolverato G, Gupta R, et al. Impact Total Psoas Volume on Short- and Long-Term Outcomes in Patients Undergoing Curative Resection for Pancreatic Adenocarcinoma: a New Tool to Assess Sarcopenia. J Gastrointest Surg. 2015;19:1593-1602.
- Kuroki LM, Mangano M, Allsworth JE, et al. Preoperative assessment of muscle mass to predict surgical complications and prognosis in patients with endometrial cancer. Ann Surg Oncol. 2015;22:972-979.
- Rutten IJG, Ubachs J, Kruitwagen R, Beets-Tan RGH, Olde Damink SWM, Van Gorp T. Psoas muscle area is not representative of total skeletal muscle area

in the assessment of sarcopenia in ovarian cancer. J Cachexia Sarcopenia Muscle. 2017;8:630-638.

- Furutate R, Ishii T, Wakabayashi R, et al. Excessive visceral fat accumulation in advanced chronic obstructive pulmonary disease. Int J Chron Obstruct Pulmon Dis. 2011;6:423-430.
- van den Borst B, Gosker HR, Koster A, et al. The influence of abdominal visceral fat on inflammatory pathways and mortality risk in obstructive lung disease. Am J Clin Nutr. 2012;96:516-526.

L1 and pectoralis muscle compared to L3

SUPPLEMENTALS



Supplemental figure 1. Scatter plot and Bland-Altman plot of skeletal muscle with data from pre-chemotherapy scans. (a) Intermeasurement correlation of pectoralis and L3, (b) Blant-Altman plot of pectoralis and L3, (c) intermeasurement correlation of L1 and L3, and (d) Blant-Altman plot of L1 and L3. Abbreviations: L1: first lumbar vertebra; L3: third lumbar vertebra.



Supplemental figure 2. Scatter plot and Bland-Altman plot of adipose tissue with data from prechemotherapy scans. (a) Intermeasurement correlation of SAT at L1 and L3, (b) Blant-Altman plot of SAT at L1 and L3, (c) intermeasurement correlation of VAT at L1 and L3, and (d) Blant-Altman plot of VAT at L1 and L3. Abbreviations: L1: first lumbar vertebra; L3: third lumbar vertebra.

3



4

Cachexia in chronic obstructive pulmonary disease: new insights and therapeutic perspective

Karin J.C. Sanders*, Anita E.M. Kneppers*, Coby van de Bool, Ramon C.J. Langen, Annemie M.W.J. Schols. * Authors contributed equally

Journal of Cachexia, Sarcopenia and Muscle. 2016;7(1):5-22.

ABSTRACT

Cachexia and muscle wasting are well recognized as common and partly reversible features of chronic obstructive pulmonary disease (COPD) adversely affecting disease progression and prognosis. This argues for integration of weight and muscle maintenance in patient care. In this review, recent insights are presented in the diagnosis of muscle wasting in COPD, the pathophysiology of muscle wasting and putative mechanisms involved in a disturbed energy balance as cachexia driver. We discuss the therapeutic implications of these new insights for optimizing and personalizing management of cachexia in COPD.

INTRODUCTION

Chronic obstructive pulmonary disease (COPD) is one of the leading causes of death worldwide. It has been estimated that COPD related mortality rates will even increase in the coming decades. This increase is not only related to the prevalence of smoking, but also ageing and reduced mortality from other common causes of death play a role [1]. Additionally, COPD is a major contributor to global disease burden, accounting for 43.3 million disability adjusted life years in 2010 [2]. The disease is characterised by persistent airflow obstruction resulting from inflammation and remodelling of the airways, and may include development of emphysema. Furthermore, systemic disease manifestations and acute exacerbations influence disease burden and mortality risk [3]. Extending the classical descriptions of the "pink puffer" and "blue bloater", recent unbiased statistical approaches [4,5] support the concept that body weight and body composition discriminate pulmonary phenotypes, and are predictors of outcome. Extra-pulmonary degenerative manifestations that may occur in COPD, include osteoporosis and muscle wasting [6]. The prevalence of muscle wasting is relatively high in COPD, 15–40% depending on definition and disease stage [7,8]. Importantly, muscle wasting not only contributes to diminished skeletal muscle function, reduced exercise capacity and decreased health status [9,10], but is also a determinant of mortality in COPD, independent of airflow obstruction [8,11].

Muscle wasting in COPD has been demonstrated by decreases in fat-free mass (FFM) at whole body level, but also specifically at level of the extremities [12]. Whole body and trunk FFM reduction are more pronounced in the emphysematous phenotype, whereas reduced FFM in extremities is not different between the pulmonary phenotypes [13,14]. In addition, muscle wasting is apparent as a decrease in the size of individual muscle fibres, and this muscle fibre atrophy in COPD seems selective for type II fibres in peripheral muscle [15,16], which is in line with other chronic diseases prone to cachexia such as chronic heart failure [17]. Furthermore, a shift in muscle fibre composition from type I (oxidative) to type II (glycolytic), accompanied by a decrease in oxidative capacity culminates in reduced muscle endurance [18]. This not only contributes to reduced exercise capacity [19], but may also affect muscle mass in COPD [20], since type I and II fibres display different responses to anabolic and catabolic signals [21,22].

While unintended weight loss was initially considered to be an indicator of inevitable and terminal progression of the disease process, there is now convincing evidence that it is an independent determinant of survival, arguing for weight maintenance in patient care. There are indications that the pathophysiology of unintended weight loss is different between clinically stable COPD and during acute flare ups of the disease. To date, data in acute exacerbations of COPD is however very limited. Therefore, lung cancer is used as a comparative acute pulmonary cachexia model.

A recent unbiased statistical approach suggests that not all COPD patients, but only the emphysematous phenotype is prone to cachexia [4], although the informative value of available clinical studies is limited by a cross-sectional study design. The last two decades have also yielded insight in the impairments of the processes governing muscle mass and identified putative triggers of muscle wasting in COPD. However, it remains unclear to what extent acute flare ups of COPD may accelerate chronic wasting of muscle mass and whether muscle wasting involves similar mechanisms as in other chronic diseases or in lung cancer cachexia. In this review, we present recent insights in the pathophysiology of a disturbed energy balance as important driver of cachexia, that may lead to novel targets for clinical management of cachexia in COPD.

RECENT DEVELOPMENTS IN IDENTIFYING MUSCLE WASTING IN COPD

Incorporation of body composition into nutritional assessment has been a major step forward in understanding systemic COPD pathophysiology, since changes in weight and classification of body mass index (BMI) do not account for (hidden) body compositional shifts in fat mass (FM), FFM and bone mineral density. In clinical research bioelectrical impedance analysis is commonly used to identify cachexia. Traditionally, reference values for FFM index in COPD were developed based on age- and gender-specific 10th percentile values [8]. These reference values were defined as abnormally low, based on well-established adverse effects of low FFM index on physical performance and survival in normal to underweight COPD patients [7,11]. However, this might underestimate low muscle mass in the increasing proportion of overweight to obese COPD patients [23]. The recent European Respiratory Society statement on nutritional assessment and therapy in COPD [24] proposed dual energy x-ray absorptiometry as most appropriate method for body composition analysis in COPD, mostly because it combines screening for osteoporosis with assessment of FM and FFM at regional level in addition to whole body level. Consequently, body composition assessed by dual energy x-ray absorptiometry also allows measurement of appendicular skeletal muscle mass, which has been demonstrated to be stronger related to physical functioning than total FFM [23]. Moreover, we recently showed that this particularly applies to overweight and obese COPD patients [23,25].

Whereas low muscle mass is prevalent in $\pm 15\%$ of well-functioning elderly in the general aged population [26], a higher prevalence can be expected in COPD as a reflection of accelerated aging [25]. Indeed, Van de Bool et al. recently identified low appendicular skeletal muscle mass in even 87% of Dutch COPD patients eligible for pulmonary rehabilitation, along with a high persisting prevalence across all BMI categories [23]. The appendicular skeletal muscle mass-wasted phenotype was not only associated with impaired strength, but in men also with decreased endurance capacity. Coexistent abdominal obesity was identified in 78% of muscle wasted patients, which appeared to have a protective effect on physical functioning [23]. Physiological alterations in terms of less hyperinflation and a larger inspiratory capacity in obese COPD patients contribute to certain advantages during physical activity [27]. In addition, mortality rates in advanced COPD are the lowest among obese subjects [28]. This prognostic advantage typically reflects the 'obesity paradox', because obesity on the other hand is also associated with increased risk of cardiovascular and metabolic diseases.

Although clinically useful estimates can be derived by dual energy x-ray absorptiometry, more precise distinction between muscle mass, visceral and subcutaneous adipose tissue require more advanced imaging technologies. This could be relevant in COPD as Van den Borst et al. and Furutate et al. reported a higher visceral adipose tissue in older age patients with COPD compared to age matched healthy controls, despite comparable subcutaneous adipose tissue and BMI [29,30].

McDonald et al. [31] demonstrated that CT-derived pectoralis muscle area provides a more clinically relevant measure of COPD-related outcomes in comparison to BMI, as lower pectoralis muscle area was associated with more severe expiratory airflow obstruction, lower quality of life, and impaired exercise capacity. Since gender differences have been documented in body composition and its functional implications [23], Diaz et al. explored gender differences in computed tomography (CT)-derived pectoralis muscle area and observed lower pectoralis muscle area in women compared to men [32]. CT scans are often used to exclude other underlying illnesses, and therefore chest CT derived analysis of body composition may be an attractive diagnostic tool to combine screening for pulmonary and systemic pathology. However, it first needs to be properly validated against reference methods of whole body and regional body composition to allow use in clinical practice. Furthermore, due to the radiation exposure, use of CT scans for body composition assessment is only admissible when scans are already performed to screen for pulmonary pathology.

NEW INSIGHTS IN THE PATHOPHYSIOLOGY OF MUSCLE WASTING IN COPD

The loss of muscle mass and cross sectional area in COPD patients as determined by imaging techniques has been confirmed at the cellular level, i.e. a reduction in muscle fibre cross sectional area [16]. As reviewed by Langen et al. [33] and Remels et al. [20] triggers of muscle wasting include hypoxemia, oxidative stress, inflammation, impaired growth factor signalling, oral glucocorticoids, disuse and malnutrition, some of which are influenced by smoking [34]. Wasting of skeletal muscle is due to a net catabolic state, which may result from an imbalance in muscle protein synthesis and breakdown (protein turnover), as well as from an imbalance in myonuclear accretion and loss (myonuclear turnover).

Protein turnover

To get insight in the rate of (muscle) protein turnover in COPD, information on (muscle)

protein synthesis as well as breakdown is required. Both increased and normal rates of whole body protein turnover have been reported in patients with COPD [35,36] but the relative contribution of muscle versus other tissues to protein turnover is unknown. Rutten et al. observed an increase in myofibrillar protein breakdown in cachectic COPD patients compared to non-cachectic patients and controls [35], but no data are available regarding muscle protein synthesis rate, except for a small study showing depressed muscle protein synthesis rates in malnourished patients with emphysema [37]. Numerous studies however have addressed molecular regulation of anabolic and catabolic pathways in the quadriceps muscle of COPD patients, which provides some insight in altered muscle protein turnover in muscle wasted COPD patients.

Proteolytic signalling

Several environmental triggers can lead to catabolic signalling in the skeletal muscle mediated by transcriptional regulators including nuclear factor kappa-light-chain-enhancer of activated B cells (NF- κ B) and forkhead box O transcription factors (FOXO)s. NF- κ B activity is increased in COPD patients compared to controls [38,39] and in cachectic COPD patients compared to non-cachectic COPD patients [38,40]. Furthermore, limb muscle NF- κ B activity is increased in patients with lung cancer cachexia [41]. FOXO mRNA and protein expression are increased in patients with COPD [38,39,42-45], seemingly independent of body composition, although it is noticeable that in all studies the patient group showed signs of emphysema. In the COPD patient group, FOXO-1 protein expression was higher in limb muscles than in respiratory muscles, while this difference was not found in controls [46]. Interestingly, the respiratory muscles of COPD patients show an opposite fibre type shift compared to limb muscles, i.e. towards more type I fibres [47,48]. This will have implications for the expression levels of constituents of atrophy signalling pathways [22,49]. Increased catabolic signalling through FOXO and NF- κ B can induce gene expression of key factors in both the ubiquitin proteasome system (UPS) [50,51] and the autophagy lysosome pathway [33,52].

Ubiquitin proteasome-mediated degradation

The ubiquitin 26S-proteasome pathway consists of coordinated actions of the ubiquitin conjugating and ligating enzymes, that link ubiquitin chains onto proteins to mark them for degradation by the proteasome [53,54]. These enzymes include tripartite motif containing 63, E3 ubiquitin protein ligase (TRIM63, referred to as MURF1), F-box protein 32 (FBXO32, referred to as ATROGIN1) and neural precursor cell expressed, developmentally down regulated 4, E3 ubiquitin protein (NEDD4).

MURF1 limb- and respiratory muscle mRNA and protein expression appear unaltered in COPD patients compared to controls [38,40,55-57], although one study reported increased MURF1 protein expression in the limb muscles of cachectic COPD patients [39]. In COPD patients,

MURF1 protein expression is relatively increased in limb muscle than in the respiratory muscle, while this difference was not found in controls [46]. Furthermore, cachectic COPD patients show increased limb muscle mRNA expression of MURF1 compared to a control population [44]. ATROGIN1 mRNA and protein expression are increased in limb muscles [38-40,44,56,57], but unaltered in respiratory muscles of patients with COPD [55]. Similarly, ATROGIN1 mRNA expression is increased in the limb muscles of smokers [58]. Furthermore, Doucet et al. found that COPD patients display a higher ATROGIN1 protein expression in limb muscles than in respiratory muscles, while this difference was not found in controls [46]. Additionally, limb muscle NEDD4 protein expression is increased in patients with COPD [57]. Total polyubiquitinated protein is increased in limb muscles of COPD patients compared to healthy controls [39,56], and in cachectic COPD patients compared to non-cachectic COPD patients [38].

Taken together, the majority of the literature suggests that wasting in COPD is accompanied by an increase in UPS activation. The increase in catabolic signalling in cachectic COPD patients is site-specific. This may reflect disuse atrophy of the limb muscle with maintained or increased respiratory muscle activity, or it may result from an interaction between inactivity and other triggers of atrophy, such as smoking.

Autophagy-lysosome-mediated degradation

The autophagy-lysosome pathway is a protein degradation pathway, which recently gained interest in the context of COPD-associated muscle dysfunction. Upon activation, autophagosomes form and mature to subsequently fuse with lysosomes. The autophago-lysosomes degrade the cargo and release amino-acids for de novo protein synthesis or other metabolic fates [59].

Signalling through v-akt murine thymoma viral oncogene (AKT) regulates mechanistic target of rapamycin (serine/threonine kinase) complex 1 (MTORC1) activity and downstream of MTORC1, unc-51 like autophagy activating kinase 1 (ULK1) activity, thereby regulating autophagy initiation [60,61]. Inhibitory MTORC1 mediated ULK1 phosphorylation is decreased in limb muscles of COPD patients compared to controls [45], which may implicate an increase in autophagic flux induction. The increase of FOXO mRNA and protein expression in COPD patients may induce the transcription of autophagy related genes. However, it should be taken into account that FOXO transcriptional activity is also regulated by post-translational modifications. Plant et al. found that the mRNA expression of autophagy-related genes beclin-1 and microtubule-associated protein 1 light chain 3 alpha (MAP1LC3A, referred to as LC3) is unaltered in the limb muscles of COPD patients compared to controls [57]. Limb muscle sequestosome 1 (SQSTM1, referred to as P62) mRNA expression, however, is increased in COPD patients [45]. Although the mRNA expression of autophagy related genes and the activation of ULK1 may give some insight in the level of autophagy initiation, it does not directly reflect the level of autophagic flux. In muscle biopsies, the conversion of LC3BI to LC3BI can be used as a measure for autophagic

flux. Furthermore, P62 is used as a marker for autophagic flux since it is broken down by the lysosome. In the limb muscles of COPD patients, the LC3BII/I ratio is increased [39,45], pointing to an increase in the autophagic flux. In contrast, P62 protein expression is increased, pointing to a decrease in autophagic flux [45]. However, it cannot be excluded that the increase in P62 protein expression is due to the increase in P62 transcription. The number of autophagosomes was found to be increased in the limb muscle of COPD patients [39,45], which suggests an increase in autophagic flux. However, it is only possible to speculate on the level of autophagic flux in the limb muscles of patients with COPD based on the currently applied markers, as these incompletely cover autophagic flux, autophagy induction and autophagic-lysosomal degradation. Therefore, autophagic flux markers should be analysed coupled to the activity status of upstream regulators such as MTORC1 and AMP-activated protein kinase (AMPK) in muscle biopsies. Moreover, besides its role in protein breakdown, autophagy also acts as a quality control mechanism for proteins and intracellular components [62,63]. Therefore, an impaired autophagic flux in COPD patients may have consequences for the integrity and function of intracellular components such as the nucleus and mitochondria.

It currently is unknown if the autophagic-lysosome pathway activity is altered during acute exacerbations of COPD, since most studies were conducted in stable COPD patients. However, in lung cancer cachexia LC3BII protein expression and BCL2/ adenovirus E1B 19kDa interacting protein 3 mRNA expression are induced [41], pointing to an increase in autophagy. From this, autophagy induction in skeletal muscle might be anticipated during acute stages of COPD wasting.

Protein synthesis signalling

A major anabolic pathway is the insulin-like growth factor 1 (IGF1)/phosphatidylinositol-4,5bisphosphate 3-kinase, catalytic subunit alpha (PIK3CA referred to as PI3K)/AKT pathway. Most studies found an increase in IGF1 mRNA expression in the limb muscle of COPD patients compared to controls [42,43,64], although Crul et al. found a decrease in IGF1 mRNA expression in stable COPD patients [65]. Unfortunately, this study did not provide body composition data. Cachectic COPD patients seem to have a lower limb muscle IGF1 mRNA and protein expression than non-cachectic COPD patients [40]. Furthermore, during an acute exacerbation, muscle IGF1 mRNA expression is lower in COPD patients than in controls, although IGF1 protein expression remains unaltered [65].

Even though IGF1 mRNA expression is increased in limb muscles of COPD patients, AKT activation remains unaltered [39,45,57]. AKT activity is relatively increased in cachectic patients compared to non-cachectic patients and healthy controls [40,44], while the decrease in IGF1 mRNA expression in this group would generally implicate a decrease of IGF1/AKT signalling. Interestingly, an increase in AKT activation is also observed in patients with lung cancer related

cachexia [41], suggesting it may be a common feature pulmonary cachexia. The discrepancy in IGF1 mRNA expression and AKT activation suggests altered regulation at the IGF- or IGF-receptor protein expression level [66].

Signalling through AKT inhibits the upstream inhibitor of MTORC1, thereby inducing MTORC1 activation and subsequent phosphorylation of its downstream targets eukaryotic translation initiation factor 4E binding protein 1 (EIF4EBP1 also called 4EBP1) and ribosomal protein S6 kinase, 70kDa, polypeptide 1 (PRPS6KB1 also called p70S6K) [67]. The increased AKT activation in the limb muscle of cachectic patients compared to non-cachectic COPD patients is paralleled by an increase in phosphorylation of the downstream targets 4E-BP1 and p70S6K [44]. P70S6K phosphorylation is unaltered in COPD patients compared to controls [44], while ribosomal protein S6 (RPS6 referred to as S6) phosphorylation was even decreased in COPD patients [45]. Together, these studies show an increase in protein synthesis signalling in the limb muscles of cachectic COPD patients compared to non-cachectic COPD patients, but no alteration in the general COPD population. In patients with lung cancer related cachexia, AKT activation is increased without concurrent activation of MTOR or its downstream targets [41]. This may indicate that, although impaired AKT signalling is found in lung cancer cachexia, AKT signalling is largely intact in COPD induced muscle wasting. However, one limitation of these studies concerns the evaluation of the activation status of protein synthesis signalling solely in a basal state. Although the current data may suggest that the protein synthesis pathway is a promising target for the treatment of COPD-induced muscle wasting, the integrity of the anabolic response should be further addressed.

Only limited data is available on anabolic signalling in respiratory muscles of COPD patients, and although the results also point to an increase in anabolic signalling, it remains unclear if this is different between cachectic and non-cachectic COPD patients. Martinez-Llorens et al. found an increase in IGF1 mRNA expression in the intercostal muscles of patients with COPD [68]. Doucet et al. compared the ratio of quadriceps to diaphragm AKT activation in COPD patients with controls and found a lower ratio in COPD [20]. This implicates that the AKT activation is relatively higher in the diaphragm than in the quadriceps. In line with this, the p70S6K phosphorylation is relatively higher in the diaphragm, while 4E-BP1 phosphorylation is higher in the quadriceps [20].

Interestingly, Tannerstedt et al. showed a difference in anabolic response between type-I and type-II muscle fibres, with increased AKT phosphorylation and downstream pathway activation in the type-II fibres [69]. In contrast, in COPD, the respiratory muscle with a shift towards more type-I fibres displayed a larger AKT activation and downstream signalling than the limb muscles with a shift towards more type-II fibres. Therefore, the shift in fibre type does not explain the variation in AKT phosphorylation and downstream signalling between the limb muscle and

respiratory muscle. Other discriminating factors, such as the muscle activity level, may be more closely linked to differences in AKT phosphorylation.

Taken together, anabolic signalling is increased in the skeletal muscle of patients with COPD, with an even larger increase in the diaphragm than the limb muscles. One may speculate that the increased activation of AKT signalling in the respiratory muscles is an attempt to preserve respiratory function by compensating catabolic triggers, although it may also reflect intrinsic alterations in muscle fibre composition.

Myonuclear turnover

Besides the turnover of proteins, the turnover of myonuclei appears essential for muscle regeneration [70-72]. Furthermore, although at a lower rate, myonuclear turnover might be indispensable for the maintenance of skeletal muscle mass. The regulated loss of a nucleus may involve apoptosis. Increased apoptosis, as determined by an increase in terminal deoxynucleotidyl transferase dUTP nick end labelling (TUNEL) staining and poly (ADP-ribose) polymerase cleavage in the skeletal muscle of cachectic COPD patients has been reported [73]. In the diaphragm of COPD patients with muscle wasting, elevated caspase-3 levels indicated an increase in apoptosis [55]. No difference in apoptosis measured by TUNEL staining or caspase-3 was found in COPD patients with maintained muscle mass compared to controls [74]. A possible alteration in apoptosis in the muscle of COPD patients remains inconclusive due to the lack of studies and the use of different markers. Furthermore, the possible role of apoptotic signalling in the skeletal muscle atrophy remains obscure [75], and multinucleated muscle fibres may utilize other mechanisms for selective myonuclear loss, such as autophagy [76].

To replenish the myonuclear pool, satellite cells are essential [72]. In contrast to muscle wasting in aging [77,78], the number of satellite cells per muscle fibre is unaltered in the limb and respiratory muscles of patients with COPD compared to controls [68,79-81]. Furthermore, no differences in satellite cell number have been reported between muscle wasted and non-muscle wasted COPD patients [80].

Upon activation, satellite cells proliferate, differentiate and fuse with myofibres. Activation and proliferation of satellite cells in the limb muscles does not seem to differ between COPD patients and age matched controls based on number of satellite cells 24h after a resistance-exercise bout [81]. However, molecular markers of satellite cell activation may be more sensitive than satellite cell number to quantify the satellite cell response.

In a basal condition, myogenic factor 5 mRNA expression is unaltered in the limb muscles of patients with COPD compared to controls [57]. Myogenic differentiation 1 (MYOD1) mRNA expression [40] and protein expression appear similar in the limb muscles of COPD patients

and healthy controls [57]. Furthermore, myogenin (MYOG) mRNA expression is unaltered in COPD patients [57]. However, in cachectic COPD patients, MYOD1 [40] and MYOG [38,39] protein expression are reduced compared to controls, while no alteration in MYOG protein expression was found in the limb muscles of cachectic compared to non-cachectic COPD patients [38]. It should be considered that in different populations and disease stages, the course of the satellite cell response might be altered, which may have implications for the timing of the measurements.

A negative regulator of myogenesis is myostatin (MSTN) [82]. Limb muscle MSTN mRNA expression is increased in COPD compared to controls [57,83], while no difference seems to be present between cachectic and non-cachectic COPD patients [40]. The increased MSTN mRNA expression in COPD patients may be partially explained by smoking status, as this increase is also found in smokers [58]. MSTN protein expression is unaltered in the limb muscle of COPD patients, independent of the pulmonary phenotype [38-40], while the circulatory level of MSTN is increased in COPD compared to control subjects [84]. It should be noted that Snijders et al. showed a delayed response in MSTN protein levels upon a single bout of resistance exercise in elderly, while basal levels did not seem to differ [85]. This implies that in absence of a myogenic trigger, intrinsic alterations in satellite cell plasticity responses in muscle of COPD patients may be masked.

Centrally localized myonuclei in myofibres are considered derived from newly fused satellite cells prior to their final location peripherally in the myofibre against the sarcolemma. COPD patients with preserved muscle mass have higher amounts of central nuclei in the limb muscle than muscle wasted COPD patients and controls [80]. This could be interpreted as an attempt to counteract atrophic signalling in COPD patients, which may be essential to preserve muscle mass. However, as there is only indirect evidence that central myonuclei reflect recent regenerative events, central nuclei could also reflect an increase in myonuclear turnover to compensate for increased loss of myonuclei.

To gain further insight in the regulation of myonuclear turnover and possible defects in COPDinduced skeletal muscle wasting, it is essential to incorporate satellite cell activation stimuli and sensitive techniques to monitor myonuclear accretion and turnover in the study design. This will require further development of new techniques, in parallel with novel ex vivo and in vitro approaches to monitor myonuclear accretion and possibly myonuclear loss, and assessment of the role of alterations in myonuclear turnover in muscle atrophy.

Loss of muscle oxidative phenotype

Besides the importance of the muscle quantity for muscle function, also the quality of the muscle should be considered. This is highlighted by the finding that muscle mass-specific

muscle strength and endurance are reduced in patients with COPD [86-88]. A well-established qualitative alteration in the skeletal muscle of COPD patients is the loss of oxidative phenotype (OXPHEN) characterized by a muscle fibre type I to type II shift and a loss of oxidative capacity [20,88,89]. The loss of OXPHEN is associated with increased oxidative stress [88,90], which may render the muscle more susceptible to muscle atrophy [38]. In addition, type II fibres are generally more susceptible to atrophy stimuli including e.g. inflammation [21] and hypoxia [22]. Therefore, the loss of OXPHEN in COPD may accelerate the loss of muscle mass, thereby linking muscle quality to muscle quantity. This is supported by the fact that non-symptomatic smokers already exhibit reduced mitochondrial capacity and a similar fibre type shift [91]. Although less extensively investigated, striking similarities are reported regarding muscle oxidative metabolism in chronic heart failure [92]. As these patients also share other systemic features and lifestyle characteristics (e.g. muscle wasting, low physical activity level) comparative analyses between well phenotyped patients with COPD and chronic heart failure may provide more insight in common and disease specific denominators and mechanisms.

Therapeutic perspective

Since muscle wasting may result from alterations in the protein and myonuclear turnover, targeting key pathways in these processes will be required to combat muscle wasting.

UPS activity is increased in the muscles of cachectic COPD patients, which implicates the atrogenes MURF1 and ATROGIN1 as targets to normalize UPS activity. This is supported by the finding that in a cell culture model and in a mouse model of muscle disuse, MURF1 inhibition and knockout respectively, prevented muscle fibre atrophy [93,94]. Pharmacological inhibitors that target specific ubiquitin-conjugating and deconjugating enzymes are being developed to treat cancer, neurodegenerative disorders and autoimmune diseases [95], but may also be highly relevant for the treatment of COPD-induced muscle wasting. Furthermore, exercise training may attenuate MURF1 expression, as was observed in the skeletal muscle of chronic heart failure patients [50,96]. In contrast to exercise training, one bout of exercise leads to an increase in MURF1 expression, albeit blunted in COPD [97,98], while the increase in proteolytic signalling is reduced by branched-chain amino acid supplementation in a healthy population [97].

Autophagy is disturbed in patients with COPD, although it remains unclear whether there is an increased induction of autophagy or an inhibition of autophagic-lysosomal degradation. Low amino acid availability can activate autophagy by inhibition of MTOR [99]. In line with this, branched-chain amino acid supplementation leads to an inhibition of autophagy by activation of MTOR [100]. Furthermore, overall low energy status, DNA damage and hypoxia can inhibit MTOR through AMPK and hypoxia inducible factor 1, alpha subunit (HIF1- α) [101,102] and thereby induce autophagy. Interestingly, exercise targets these factors, and exercise training results in elevated levels of basal autophagy [103]. However, exercise leads to an increase in muscle mass and strength in COPD patients [98]. Moreover, autophagy is required for muscle adaptations to training [103,104]. The counterintuitive effect of exercise on autophagy may therefore be more tightly linked to its function in quality control.

A more speculative thought is that autophagy may play a role in selective removal of damaged myonuclei. Besides causing mitochondrial DNA damage, oxidative stress also causes nuclear DNA damage. COPD patients display elevated levels of oxidative stress, which may lead to increased DNA damage and requires increased removal of damaged nuclei. Although exercise induces oxidative stress, which is even accentuated in COPD patients [105], exercise also triggers myogenesis [106]. In COPD, this myogenic response to exercise seems intact, although it is specifically impaired in cachectic COPD [40, 98]. Exercise induced satellite cell activation is mediated by IGF signalling [107]. In contrast, MSTN signalling inhibits satellite cell activation [108]. Since MSTN expression is increased in COPD, pharmacological inhibition of MSTN might be beneficial to prevent COPD-induced muscle wasting. This idea is supported by the finding that in mice with chronic kidney failure, pharmacological inhibition of MSTN blocks muscle atrophy [109], and that pharmacological inhibition of the ActII-receptor, which mediates MSTN signalling, prevents glucocorticoid-induced muscle atrophy [110]. In vitro, glutamine reduced the tumour necrosis factor alpha dependent increase of MSTN [111]. This implicates that availability of amino acids is important for normal satellite cell function in COPD, and that restoration of normal amino acid levels may be required for muscle maintenance. Taken together, the dual function and differential regulation of UPS and autophagy in the maintenance of muscle mass and quality reflects a highly interactive signalling network that is regulated by several upstream pathways. The effect of specific modulation of UPS and autophagy mediators may therefore transcend catabolic signalling and may affect a range of other cellular processes, yielding it difficult to predict long-term side effects. So far, exercise seems to be the only intervention that can target UPS and autophagy leading to improved quantity, as well as an improved quality of the muscle in COPD patients. One prerequisite is that COPD patients, and specifically cachectic COPD patients, have maintained responsiveness to exercise stimuli, which remains to be established. Exercise capacity in COPD may be limited by impaired pulmonary function, leading to incapability to supply a sufficiently strong exercise trigger to the muscles. In this case pharmacological or nutritional activators of AMPK, sirtuin 1 and peroxisome proliferator-activated receptors (PPARs) such as metformin, resveratrol, rosglitazone and poly unsaturated fatty acids could be used as exercise mimetics and may help sensitize the muscle to a following exercise bout. Furthermore, anabolic steroids could be considered in the treatment of COPD induced muscle wasting, although a recent meta-analysis showed that exercise capacity of COPD patients was not improved [112]. It should also be considered that an appropriate nutritional status is necessary for the beneficial effects of exercise, and that exercise (in particular endurance type of exercise) in a malnourished state could even have detrimental effects by worsening the energy imbalance. Taken together, a multi-modal approach may be required to combat COPD-induced muscle wasting, in which exercise training is central. However, to establish such intervention, further research is crucial to determine whether the response to exercise is intact, or if specific defects occur, in cachectic patients with COPD.

PUTATIVE MECHANISMS INVOLVED IN A DISTURBED ENERGY BALANCE IN COPD

Specific loss of muscle mass in weight stable COPD patients has been observed, which may reflect a tissue specific sensitivity to an overall catabolic state. A net catabolic state may also result from an imbalance in energy expenditure and energy availability.

Increased energy expenditure

It is well established that components of whole body energy expenditure may be increased in patients with COPD [113]. Total daily energy expenditure (TEE) is the sum of resting metabolic rate (RMR), activity induced energy expenditure and diet-induced thermogenesis. Assessment of TEE requires sophisticated methodology including a respiration chamber [114] or doubly labelled water to allow assessment of TEE in free living conditions [115,116]. Data on TEE in COPD are scarce and sometimes contradicting which may be related to the use of different methodology or patient characteristics. Based on the available evidence it seems that in particular the emphysematous phenotype is prone to increased TEE, as high values are more often observed in patients with low carbon monoxide diffusing capacity [117-119]. This is in line with a positive effect of lung volume reduction surgery on body weight in emphysema although surgical intervention might not only decrease energy requirements but also improve dietary intake by alleviating dyspnoea [120]. More research is needed to assess putative differences in whole body energy metabolism and its components among COPD phenotypes under comparable standardized circumstances.

Numerous studies have shown that RMR is raised [121-123]. This is more prevalent in emphysema [124,125], during acute exacerbations [126], and appears inversely correlated with forced expiratory volume in 1 second when comparing different studies [118,119,122,127]. Highest values are found among weight losing patients [122]. This is in contrast with non-pathology-induced malnutrition, where subjects with low BMI have lower RMR due to hypometabolic adjustments [128]. The same results are found for non-small cell lung cancer (NSCLC), where RMR is found to be upregulated in 74% of primary lung cancer patients [129-131]. Hypermetabolism at rest was also found to be more pronounced in weight-losing compared to weight-stable lung cancer patients [132]. Thus increased RMR is a consistent feature of chronic and more acute cachexia and seems to be more pronounced in the emphysematous subtype.

Activity-induced energy expenditure is the most variable component of TEE, and it has been postulated that COPD patients reduce physical activity to compensate for dyspnoea severity

or to anticipate to breathlessness. Indeed lower physical activity levels are seen in COPD [133]. Physical inactivity is associated with advanced disease stage [134-136], exacerbations [137,138] and degree of emphysema [139]. In addition, lung volume reduction surgery in patients with severe emphysema improved exercise performance due to reduced lung hyperinflation, less dyspnoea severity and less energy cost of breathing [140]. However, it did not cause augmentation of physical activity level, implying other factors play a role, including motivation or anxiety [141]. There are several indications that when COPD patients perform physical activities, they require more energy. For examples, Lahaije et al. found a higher daily activity-related oxygen consumption assessed by a face mask measuring ventilatory and metabolic demand in COPD patients compared to healthy controls [142], while Vaes et al. found an increase in FFM adjusted oxygen consumption in COPD patients compared to controls, although total oxygen consumption was not altered [143]. Indirect evidence for altered activity induced energy expenditure is a rise in plasma ammonia in COPD patients during low intensity walking, which is an indicator of muscle ATP depletion and metabolic stress [144]. These collective data may indicate that COPD patients use oxygen less efficient and exhibit an altered energy metabolism during physical activity. This is not surprising in view of the shift in lower limb muscle fibre type composition in COPD towards less oxidative fibres, which appears to be more pronounced in the emphysematous phenotype [18]. The opposite shift in muscle fibre type of the diaphragm relative to the limb muscle [47,48] indicates an adaptation to chronic increase in work of breathing. Together with hyperinflation-induced mechanical inefficiency, this muscle fibre type shift could contribute to enhanced oxygen cost of breathing, illustrated by the effects of lung volume reduction surgery [145] and non-invasive positive-pressure ventilation therapy [146]. For comparison, in lung cancer, physical activity level assessed by accelerometry is also reduced [147,148], but no specific data about activity induced energy expenditure is available.

Diet-induced thermogenesis represents metabolic oxygen cost for processing of ingested nutrients. Green and colleagues [125], reported enhanced diet-induced thermogenesis in emphysematous COPD patients but this was not confirmed by other authors [149,150]. These differences may be due to different test meal composition and portion size. Although oxygen desaturations during meals were noticed in severe COPD patients [152], it is unknown whether this is diet-induced thermogenesis-related or not. Therefore, the thermic effect of dietary intake remains unclear. Taken together, it indicates that energy requirements are increased in COPD and there is certainly no adaptive reduced energy demand.

In addition to the hypermetabolic state, early clinical trials have shown that enhanced systemic inflammation is a contributing factor to elevated RMR, both in COPD [153] and in lung cancer [154], the source of which is yet unclear. Besides pulmonary inflammation [155], also adipose tissue has been suggested to contribute to a higher inflammatory gene expression in adipose tissue, as has been reported in malnourished patients with advanced COPD [156].

Adipose tissue metabolism

In cachexia muscle wasting is accompanied by loss of adipose tissue [14,157]. In fact, in cancer-induced cachexia adipose tissue is often one of the first affected organs, illustrated by decreasing fat cell volume and upregulation of fatty acid metabolism [158]. Regarding COPD, low BMI [11] and fat mass depletion [14] particularly occur in those with advanced disease and in the emphysematous phenotype.

Schols et al. observed low leptin levels in blood of patients with emphysema compared to chronic bronchitis in line with a lower BMI and fat mass [159]. After adjustment for FM and oral corticosteroid use as possible confounders, leptin was associated with systemic inflammation, in particular in the emphysematous patients. More recently, Brusik et al. investigated serum levels and adipose tissue expression of leptin and adiponectin in patients with COPD and reported an association between decreased serum and tissue leptin levels, decreased serum adiponectin and increased RMR adjusted for body weight in underweight patients [160]. In adjpose tissue two cell types can be distinguished: white adipose tissue and brown adipose tissue (BAT). BAT is differentiated from white adipose tissue by the presence of cold-induced thermogenesis. Thermogenesis is facilitated by BAT-specific uncoupling proteins that dissipate the proton gradient in mitochondria in order to generate heat [161]. High amounts of mitochondria are responsible for the brown colour of BAT [162]. Additionally, white adipose tissue can be converted in BAT, called browning [163]. BAT activation negatively correlates with BMI, as demonstrated by decreased BAT activation in obese subjects [164] and during aging [165]. Cachexia on the other hand is characterized by fat mass depletion. This raises the question whether there is a role of BAT activity in the hypermetabolic state, as seen in pulmonary cachexia.

No studies are performed to determine BAT activity in COPD patients. With respect to lung cancer, results are conflicting and scarce. Despite negative results of BAT activation reported by some authors [166,167], Shellock and colleagues provided evidence for BAT activation as a cachexia mediator. Autopsy reports of cachectic cancer patients revealed high incidence of BAT in this group compared to age-matched controls [168]. Furthermore, a correlation between BAT activity and neoplastic status has been suggested [169], although the authors also reported high amount of BAT activation in non-malignancy subjects.

There is indirect evidence that BAT activation might be a potential cachexia driver in COPD as well. Hypoxia [170] and hypermetabolism [122] are hallmarks of COPD. In response to hypoxia, cells can produce vascular endothelial growth factor (VEGF) in order to restore oxygen supply [171]. This has been established by Van de Borst et al., who found an upregulation of the VEGF gene in adipose tissue in response to chronic hypoxia in mice. Congruently, adipose tissue showed a brown appearance. This browning of adipose tissue was established by increased

expression of uncoupling protein 1 [172], which proposes a link between hypoxia-induced VEGF activation and browning. Indeed, Sun et al. revealed upregulation of uncoupling protein 1, the main characteristic of BAT, in a VEGF overexpressing mouse model [173]. In addition, recently, increased thermogenesis and energy expenditure were found in mice with VEGF overexpression in BAT and white adipose tissue [174].

Another hypoxia alignment occurs in the form of lactate. In peripheral muscle of COPD patients increased glycolysis metabolism is observed [175], which in turn causes rising lactate levels [176,177]. Lactate is indeed increasingly released by adipocytes in a hypoxic environment [178], which in turn is able to control the expression of uncoupling protein 1. The uncoupling protein 1 regulation is independent of HIF-1 α and thereby also promotes browning under normoxic circumstances [179].

Another possible browning factor is beta-adrenergic stimulation mediated by norepinephrine [180, 181]. Emphysematous COPD patients indeed exhibit increased plasma norepinephrine levels [182], indicating a possible activation of the autonomic nervous system. However, Schiffelers et al. showed a blunted beta-adrenoceptor-mediated lipolytic and thermogenic response [183], suggesting desensitization. Additionally, in 10 lean healthy men BAT activity in response to a systemic non-selective beta-agonist was not enhanced [184]. In contrast, blocking the receptors by propranolol decreases BAT activity [185]. Therefore, activation of brown fat through beta-adrenergic stimulation remains disputable.

It can be concluded that there is some indirect evidence pointing towards a role of BAT in pulmonary cachexia, but this area requires more research to identify therapeutic potential.

Compromised dietary intake

In order to compensate for increased energy requirements in COPD, patients should be able to adapt their dietary intake. Systematic analyses of dietary intake in COPD patients are rare. In terms of caloric content, dietary intake was found to be normal compared to healthy controls, but inadequate for measured energy expenditure [118,186-188]. During severe acute exacerbations the gap between energy intake and energy expenditure becomes even wider, which slowly decreases upon recovery [126,189]. To our knowledge, no human studies have systematically investigated the relation between dietary intake and disease severity or putative differences between emphysematous and non-emphysematous patients. Advanced disease stages and acute exacerbations are often characterized by chronic or acute hypoxemia [170]. It is well established that mice under chronic hypoxic circumstances experience weight loss, which is partly due to temporarily decreased dietary intake [172].
Anorexia

It is acknowledged that apparently normal dietary intake in COPD patients may be insufficient to meet elevated energy requirements but reduced food intake may also be caused by anorexia, i.e. loss of appetite [190]. A few underlying causes have been mentioned, including nicotine use [191], physical discomfort, such as dyspnoea and increased breathing effort [192], depression and anxiety [193], seen in COPD [194] as well as in NSCLC [195,196].

Besides pulmonary and psychological symptoms, COPD patients often experience pain. In a Norwegian study, which controlled for age and gender, 45% of the COPD patients reported chronic pain, compared to 34% of the general population [197]. Opioids are commonly used to combat pain in COPD [198]. Side-effects of opioids occur regularly, and opioids are able to cause gastrointestinal motility disorders [199], of which constipation is the most common [200]. People suffering from constipation often present with anorexia [201], probably due to early satiety. Separate from use of pain medication, early satiety and abdominal bloating is highly prevalent in COPD [202].

Chemosensory alterations

Food intake is regulated by taste and smell [203,204], and chemosensory dysfunction could influence dietary intake. Nordén et al. showed that 21 out of 169 stable COPD patients reported taste changes, which contributed to a decreased energy intake [205]. In addition, Dewan et al. compared 20 COPD subjects with long-term oxygen therapy to 20 COPD patients without oxygen therapy and 20 healthy elderly controls. They found reduced smell and taste test scores among COPD patients compared to controls, independent of oxygen supply [206]. Also Wardwell and colleagues found that healthy elderly tended to be able to identify more different tastes correctly than COPD patients, although not statistically significant [207]. Both authors did not report medication use, and therefore the influence of treatment is unknown. Although data are scarce and methodological quality of the studies is limited, these data suggest that COPD or its treatment could modify taste and smell detection.

Food reward system

Fullness is regulated by gastrointestinal hormones, including leptin and ghrelin [208], and their secretion is affected by dietary intake and nutritional status. Clinically stable emphysematous COPD patients exhibit low leptin levels compared to the chronic bronchitis subtype [159]. During acute flare ups these plasma levels rise temporary [189], as seen in NSCLC [209,210]. Likewise, enhanced plasma ghrelin levels are noticed in COPD [211] and NSCLC [212], and are related to cachectic status.

The peripheral hormonal satiety system closely interacts with the central nervous system in order to regulate food intake. Brain imaging studies have revealed reward-specific brain regions

related to food reward [213], and activation of these regions correlate with food rewarding [214]. Different orexigenic and anorexigenic peptides and hormones can stimulate neurons in these specific cerebral regions [208,215]. For instance, leptin inhibits neurons, causing reduced food intake and increased energy expenditure [216]. Ghrelin, considered to be a leptin counterpart, can induce food intake mediated by stimulation of neurons in this area [217].

There is surprisingly no human study available that explored the role of central dysregulation in food reward in patients with COPD. In relation to lung cancer cachexia only one study is performed identifying brain activity in anorectic and non-anorectic patients while receiving pleasant and unpleasant food cues [218]. In contrast to non-anorectic patients, anorectic patients showed no brain activity differences in response to pleasant versus unpleasant pictures. This implies an overall blunted response in the perceptual and motivational system that could also be involved in COPD but requires further investigation.

Therapeutic perspective

The importance of nutritional status is not only emphasized by adverse effects on muscle function and exercise performance, but also by detrimental effects of malnutrition on lung tissue. These effects have mostly been studied in animal models. Following the clinical phenotyping of the pink puffer and the blue bloater in the 1960s, Sahebjami et al. found reinforcement of pre-existing emphysematous processes due to caloric food deprivation in rats [219], which was more pronounced in young rats [220]. These deleterious effects could partly be reversed by refeeding [221]. In contrast, Bishai et al. found no alveolar size changes in calorie restricted mice, although the lungs became stiffer and lung capacity was decreased [222]. Supplementary evidence was provided by emphysema-like changes present in anorexia nervosa patients, which underscores the impact of chronic malnutrition on alveoli [223]. In addition to lung tissue, respiratory muscles also contribute to breathing. Weight loss does not spare the respiratory muscles, since weight loss is related to diminished diaphragm weight [224] and decreased function [225] in experimental models and in humans.

As proof of concept, Efthimiou et al. conducted a randomized controlled trial to investigate the effect of nutritional support on respiratory and peripheral muscle function in malnourished COPD patients. They reported improvement in respiratory muscle strength and hand grip strength, accompanied by less dyspnoea and enhanced distance in 6 minute walk test. Importantly, these effects diminished after quitting the dietary supplementation [226]. The positive effects of dietary support on body weight was verified by Weekes et al., who found weight gain in the intervention group, whereas the control group continued to lose weight. Addition of dietary counselling to dietary support has been shown to maintain weight loss after cessation of intervention [227].

Initially, the focus was primarily on caloric intake to balance energy requirements, but more recent proof of concept experiments also highlighted the importance of optimizing protein intake [228,229]. Low intake of other essential nutrients is identified, including vitamin D and calcium [230], which are also relevant in the context of osteoporosis as clustering comorbid condition.

One should keep in mind that dietary intake does not reflect actual availability of ingested micronutrients. There are indications that intestinal function is impaired in COPD, illustrated by splanchnic hypoperfusion and reduced intestinal permeability [231]. Altered intestinal function translates into reduced splanchnic extraction of amino acids derived from nutritional intake [36,232], but amino acid uptake in the skeletal muscle of clinically stable COPD patients appears increased [233]. Thus the significance for clinical applications remains ambiguous.

Both dietary intake and nutrient availability are controlled by gastrointestinal hormones. By binding to the growth hormone secretagogue receptor, ghrelin can induce secretion of growth hormone [234]. This leads to modulation of the growth hormone/IGF1 axis, which is an important anabolic pathway in human skeletal muscle [235]. Furthermore, ghrelin can induce food intake, mediated by stimulation of specific neurons in the food reward centre [217]. Due to its orexigenic property, ghrelin analogues have been proposed for clinical application in cachexia. One clinical trial with ghrelin analogues have been conducted in COPD patients. They reported improvements of ventilatory efficiency at peak exercise, reflected by increased peak oxygen uptake [236]. However, it did not translate in improved 6-minute walk distance and no data are available about body composition or food intake. Clinical trials in cancer cachexia [237], including lung cancer [238] demonstrate an enhanced lean body mass and quality of life. Hence, ghrelin analogues warrant further investigation in COPD. Besides dietary and pharmacological interventions, cognitive behavioural interventions are relatively underexplored in the management of cachexia in COPD. Although results from different functional neuroimaging studies are inconsistent and sometimes conflicting [239], there might be altered reactivity in the brain reward system in response to perceived food stimuli in people with altered eating patterns, including anorexia nervosa and obesity [240,241]. Therefore, cognitive behavioural therapy may serve as a treatment for patients with an eating disorder like anorexia nervosa [242]. Recently, a randomized controlled trial was conducted in obese subjects, receiving behavioural therapy for 6 months, in order to reduce weight. Analysis of functional magnetic resonance imaging revealed changes in reward system activity in the intervention group versus controls. Further research has to identify whether it is possible to enhance neuroplasticity in the food reward centre in order to increase successfulness for eating disorder treatment [243]. This opens up new insights and therapeutic opportunities for suspected nutritional therapy resistant cachectic COPD patients, if disturbances in the central food reward system are indeed identified.



Figure 1. Pulmonary and extra-pulmonary crosstalk in COPD cachexia. (a) Altered brain responses to food stimuli; (b) muscle fibre type shifting and oxidative metabolism; (c) altered adipose tissue metabolism; (d) adipose tissue wasting; (e) limb muscle dysfunction; (f) respiratory muscle dysfunction; (g) osteoporosis; (h) altered gut integrity and reduced splanchnic extraction.

CONCLUSION

It is well established that the prevalence and related disease burden of cachexia is high in COPD and likely to increase in the near future given the high and increasing prevalence of the disease in an ageing population. Nevertheless, cachexia management is still poorly implemented in clinical practice. In 2014 the European Respiratory Society published a statement on nutritional assessment and therapy in COPD including a nutritional risk stratification diagram based on assessment of BMI, weight changes and body composition, that could be useful in patient counselling [24]. In order to increase overall survival and compress morbidity a multi-modal intervention approach is needed, which should target the discussed factors involved in cachexia (figure 1). Such multi-modal intervention approach, encompassing exercise training and improvement of energy balance and nutrient availability is currently feasible as supported by recent statements and meta-analyses, possibly improved in the near future by targeted pharmacological interventions and cognitive behavioural therapy to sensitize patients to anabolic stimuli.

REFERENCES

- Mathers CD, Loncar D. Projections of global mortality and burden of disease from 2002 to 2030. PLoS medicine. 2006;3:e442.
- Prince MJ, Wu F, Guo Y, et al. The burden of disease in older people and implications for health policy and practice. Lancet. 2014.
- Vestbo J, Hurd SS, Agusti AG, et al. Global strategy for the diagnosis, management, and prevention of chronic obstructive pulmonary disease: GOLD executive summary. Am J Respir Crit Care Med. 2013;187:347-365.
- Vanfleteren LE, Spruit MA, Groenen M, et al. Clusters of comorbidities based on validated objective measurements and systemic inflammation in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2013;187:728-735.
- Burgel PR, Paillasseur JL, Peene B, et al. Two distinct chronic obstructive pulmonary disease (COPD) phenotypes are associated with high risk of mortality. PLoS One. 2012;7:e51048.
- Romme EA, Murchison JT, Phang KF, et al. Bone attenuation on routine chest CT correlates with bone mineral density on DXA in patients with COPD. J Bone Miner Res. 2012;27:2338-2343.
- Schols AM, Soeters PB, Dingemans AM, Mostert R, Frantzen PJ, Wouters EF. Prevalence and characteristics of nutritional depletion in patients with stable COPD eligible for pulmonary rehabilitation. Am Rev Respir Dis. 1993;147:1151-1156.
- Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. Am J Respir Crit Care Med. 2006;173:79-83.
- Mostert R, Goris A, Weling-Scheepers C, Wouters EF, Schols AM. Tissue depletion and health related quality of life in patients with chronic obstructive pulmonary disease. Respir Med. 2000;94:859-867.
- Engelen MP, Schols AM, Baken WC, Wesseling GJ, Wouters EF. Nutritional depletion in relation to respiratory and peripheral skeletal muscle function in out-patients with COPD. Eur Respir J. 1994;7:1793-1797.
- Schols AM, Broekhuizen R, Weling-Scheepers CA, Wouters EF. Body composition and mortality in chronic obstructive pulmonary disease. The

American journal of clinical nutrition. 2005;82:53-59.

- Bernard S, LeBlanc P, Whittom F, et al. Peripheral muscle weakness in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 1998;158:629-634.
- Engelen MP, Schols AM, Does JD, Wouters EF. Skeletal muscle weakness is associated with wasting of extremity fat-free mass but not with airflow obstruction in patients with chronic obstructive pulmonary disease. The American journal of clinical nutrition. 2000;71:733-738.
- Engelen MP, Schols AM, Lamers RJ, Wouters EF. Different patterns of chronic tissue wasting among patients with chronic obstructive pulmonary disease. Clin Nutr. 1999;18:275-280.
- Caron MA, Debigare R, Dekhuijzen PN, Maltais F. Comparative assessment of the quadriceps and the diaphragm in patients with COPD. J Appl Physiol (1985). 2009;107:952-961.
- Gosker HR, Engelen MP, van Mameren H, et al. Muscle fiber type IIX atrophy is involved in the loss of fat-free mass in chronic obstructive pulmonary disease. The American journal of clinical nutrition. 2002;76:113-119.
- Mancini DM, Coyle E, Coggan A, et al. Contribution of intrinsic skeletal muscle changes to 31P NMR skeletal muscle metabolic abnormalities in patients with chronic heart failure. Circulation. 1989;80:1338-1346.
- Gosker HR, van Mameren H, van Dijk PJ, et al. Skeletal muscle fibre-type shifting and metabolic profile in patients with chronic obstructive pulmonary disease. Eur Respir J. 2002;19:617-625.
- Maltais F, LeBlanc P, Whittom F, et al. Oxidative enzyme activities of the vastus lateralis muscle and the functional status in patients with COPD. Thorax. 2000;55:848-853.
- Remels AH, Gosker HR, Langen RC, Schols AM. The mechanisms of cachexia underlying muscle dysfunction in COPD. J Appl Physiol (1985). 2013;114:1253-1262.
- Files DC, D'Alessio FR, Johnston LF, et al. A critical role for muscle ring finger-1 in acute lung injuryassociated skeletal muscle wasting. Am J Respir Crit Care Med. 2012;185:825-834.
- de Theije CC, Langen RC, Lamers WH, Gosker HR, Schols AM, Koehler SE. Differential sensitivity of oxidative and glycolytic muscles to hypoxiainduced muscle atrophy. J Appl Physiol (1985).

2014:jap 00624 02014.

- van de Bool C, Rutten EP, Franssen FM, Wouters EF, Schols AM. Antagonistic implications of sarcopenia and abdominal obesity on physical performance in COPD. Eur Respir J. 2015.
- Schols AM, Ferreira IM, Franssen FM, et al. Nutritional assessment and therapy in COPD: a European Respiratory Society statement. Eur Respir J. 2014.
- Jones SE, Maddocks M, Kon SS, et al. Sarcopenia in COPD: prevalence, clinical correlates and response to pulmonary rehabilitation. Thorax. 2015.
- Cherin P, Voronska E, Fraoucene N, de Jaeger
 C. Prevalence of sarcopenia among healthy ambulatory subjects: the sarcopenia begins from 45 years. Aging Clin Exp Res. 2014;26:137-146.
- O'Donnell DE, Ciavaglia CE, Neder JA. When obesity and chronic obstructive pulmonary disease collide. Physiological and clinical consequences. Annals of the American Thoracic Society. 2014;11:635-644.
- Guenette JA, Jensen D, O'Donnell DE. Respiratory function and the obesity paradox. Curr Opin Clin Nutr Metab Care. 2010;13:618-624.
- van den Borst B, Gosker HR, Koster A, et al. The influence of abdominal visceral fat on inflammatory pathways and mortality risk in obstructive lung disease. The American journal of clinical nutrition. 2012;96:516-526.
- Furutate R, Ishii T, Wakabayashi R, et al. Excessive visceral fat accumulation in advanced chronic obstructive pulmonary disease. International journal of chronic obstructive pulmonary disease. 2011;6:423-430.
- McDonald ML, Diaz AA, Ross JC, et al. Quantitative computed tomography measures of pectoralis muscle area and disease severity in chronic obstructive pulmonary disease. A cross-sectional study. Annals of the American Thoracic Society. 2014;11:326-334.
- Diaz AA, Zhou L, Young TP, et al. Chest CT measures of muscle and adipose tissue in COPD: genderbased differences in content and in relationships with blood biomarkers. Academic radiology. 2014;21:1255-1261.
- Langen RC, Gosker HR, Remels AH, Schols AM. Triggers and mechanisms of skeletal muscle wasting in chronic obstructive pulmonary disease. The international journal of biochemistry & cell biology. 2013;45:2245-2256.
- 34. Zuo L, He F, Sergakis GG, et al. Interrelated role of

cigarette smoking, oxidative stress, and immune response in COPD and corresponding treatments. American journal of physiology Lung cellular and molecular physiology. 2014;307:L205-218.

- Rutten EP, Franssen FM, Engelen MP, Wouters EF, Deutz NE, Schols AM. Greater whole-body myofibrillar protein breakdown in cachectic patients with chronic obstructive pulmonary disease. The American journal of clinical nutrition. 2006;83:829-834.
- Engelen MP, Deutz NE, Wouters EF, Schols AM. Enhanced levels of whole-body protein turnover in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2000;162:1488-1492.
- Morrison WL, Gibson JN, Scrimgeour C, Rennie MJ. Muscle wasting in emphysema. Clinical science. 1988;75:415-420.
- Fermoselle C, Rabinovich R, Ausin P, et al. Does oxidative stress modulate limb muscle atrophy in severe COPD patients? Eur Respir J. 2012;40:851-862.
- Puig-Vilanova E, Rodriguez DA, Lloreta J, et al. Oxidative Stress, Redox Signaling Pathways, And Autophagy In Cachectic Muscles Of Male Patients With Advanced Copd And Lung Cancer. Free radical biology & medicine. 2014.
- Vogiatzis I, Simoes DC, Stratakos G, et al. Effect of pulmonary rehabilitation on muscle remodelling in cachectic patients with COPD. Eur Respir J. 2010;36:301-310.
- Op den Kamp CM, Langen RC, Snepvangers FJ, et al. Nuclear transcription factor kappa B activation and protein turnover adaptations in skeletal muscle of patients with progressive stages of lung cancer cachexia. The American journal of clinical nutrition. 2013;98:738-748.
- Debigare R, Cote CH, Maltais F. Ubiquitination and proteolysis in limb and respiratory muscles of patients with chronic obstructive pulmonary disease. Proceedings of the American Thoracic Society. 2010;7:84-90.
- Debigare R, Maltais F, Cote CH, et al. Profiling of mRNA expression in quadriceps of patients with COPD and muscle wasting. Copd. 2008;5:75-84.
- Doucet M, Russell AP, Leger B, et al. Muscle atrophy and hypertrophy signaling in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2007;176:261-269.
- 45. Guo Y, Gosker HR, Schols AM, et al. Autophagy

in locomotor muscles of patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2013;188:1313-1320.

- Doucet M, Dube A, Joanisse DR, et al. Atrophy and hypertrophy signalling of the quadriceps and diaphragm in COPD. Thorax. 2010;65:963-970.
- Levine S, Gregory C, Nguyen T, et al. Bioenergetic adaptation of individual human diaphragmatic myofibers to severe COPD. J Appl Physiol (1985). 2002;92:1205-1213.
- Levine S, Kaiser L, Leferovich J, Tikunov B. Cellular adaptations in the diaphragm in chronic obstructive pulmonary disease. The New England journal of medicine. 1997;337:1799-1806.
- Mofarrahi M, Guo Y, Haspel JA, et al. Autophagic flux and oxidative capacity of skeletal muscles during acute starvation. Autophagy. 2013;9:1604-1620.
- 50. Gielen S, Sandri M, Kozarez I, et al. Exercise training attenuates MuRF-1 expression in the skeletal muscle of patients with chronic heart failure independent of age: the randomized Leipzig Exercise Intervention in Chronic Heart Failure and Aging catabolism study. Circulation. 2012;125:2716-2727.
- Sandri M, Sandri C, Gilbert A, et al. Foxo transcription factors induce the atrophy-related ubiquitin ligase atrogin-1 and cause skeletal muscle atrophy. Cell. 2004;117:399-412.
- Mammucari C, Milan G, Romanello V, et al. FoxO3 controls autophagy in skeletal muscle in vivo. Cell metabolism. 2007;6:458-471.
- Glickman MH, Ciechanover A. The ubiquitinproteasome proteolytic pathway: destruction for the sake of construction. Physiological reviews. 2002;82:373-428.
- 54. Pickart CM. Back to the future with ubiquitin. Cell. 2004;116:181-190.
- Ottenheijm CA, Heunks LM, Li YP, et al. Activation of the ubiquitin-proteasome pathway in the diaphragm in chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2006;174:997-1002.
- Lemire BB, Debigare R, Dube A, Theriault ME, Cote CH, Maltais F. MAPK signaling in the quadriceps of patients with chronic obstructive pulmonary disease. J Appl Physiol (1985). 2012;113:159-166.
- Plant PJ, Brooks D, Faughnan M, et al. Cellular markers of muscle atrophy in chronic obstructive pulmonary disease. American journal of respiratory

cell and molecular biology. 2010;42:461-471.

- Petersen AM, Magkos F, Atherton P, et al. Smoking impairs muscle protein synthesis and increases the expression of myostatin and MAFbx in muscle. American journal of physiology Endocrinology and metabolism. 2007;293:E843-848.
- Yang Z, Klionsky DJ. An overview of the molecular mechanism of autophagy. Current topics in microbiology and immunology. 2009;335:1-32.
- Mizushima N. The role of the Atg1/ULK1 complex in autophagy regulation. Current opinion in cell biology. 2010;22:132-139.
- Mammucari C, Schiaffino S, Sandri M. Downstream of Akt: FoxO3 and mTOR in the regulation of autophagy in skeletal muscle. Autophagy. 2008;4:524-526.
- Mai S, Muster B, Bereiter-Hahn J, Jendrach M. Autophagy proteins LC3B, ATG5 and ATG12 participate in quality control after mitochondrial damage and influence lifespan. Autophagy. 2012;8:47-62.
- Rello-Varona S, Lissa D, Shen S, et al. Autophagic removal of micronuclei. Cell cycle. 2012;11:170-176.
- Lewis A, Riddoch-Contreras J, Natanek SA, et al. Downregulation of the serum response factor/miR-1 axis in the quadriceps of patients with COPD. Thorax. 2012;67:26-34.
- Crul T, Spruit MA, Gayan-Ramirez G, et al. Markers of inflammation and disuse in vastus lateralis of chronic obstructive pulmonary disease patients. European journal of clinical investigation. 2007;37:897-904.
- Adams GR. Invited Review: Autocrine/paracrine IGF-I and skeletal muscle adaptation. J Appl Physiol (1985). 2002;93:1159-1167.
- Hay N, Sonenberg N. Upstream and downstream of mTOR. Genes & development. 2004;18:1926-1945.
- Martinez-Llorens J, Casadevall C, Lloreta J, et al. [Activation of satellite cells in the intercostal muscles of patients with chronic obstructive pulmonary disease]. Archivos de bronconeumologia. 2008;44:239-244.
- Tannerstedt J, Apro W, Blomstrand E. Maximal lengthening contractions induce different signaling responses in the type I and type II fibers of human skeletal muscle. J Appl Physiol (1985). 2009;106:1412-1418.
- 70. Lepper C, Partridge TA, Fan CM. An absolute

requirement for Pax7-positive satellite cells in acute injury-induced skeletal muscle regeneration. Development. 2011;138:3639-3646.

- Relaix F, Zammit PS. Satellite cells are essential for skeletal muscle regeneration: the cell on the edge returns centre stage. Development. 2012;139:2845-2856.
- Adams GR, Caiozzo VJ, Haddad F, Baldwin KM. Cellular and molecular responses to increased skeletal muscle loading after irradiation. American journal of physiology Cell physiology. 2002;283:C1182-1195.
- Agusti AG, Sauleda J, Miralles C, et al. Skeletal muscle apoptosis and weight loss in chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2002;166:485-489.
- Gosker HR, Kubat B, Schaart G, van der Vusse GJ, Wouters EF, Schols AM. Myopathological features in skeletal muscle of patients with chronic obstructive pulmonary disease. Eur Respir J. 2003;22:280-285.
- Dupont-Versteegden EE. Apoptosis in skeletal muscle and its relevance to atrophy. World journal of gastroenterology : WJG. 2006;12:7463-7466.
- Mijaljica D, Devenish RJ. Nucleophagy at a glance. Journal of cell science. 2013;126:4325-4330.
- Verdijk LB, Snijders T, Drost M, Delhaas T, Kadi F, van Loon LJ. Satellite cells in human skeletal muscle; from birth to old age. Age. 2014;36:545-547.
- Snijders T, Verdijk LB, van Loon LJ. The impact of sarcopenia and exercise training on skeletal muscle satellite cells. Ageing research reviews. 2009;8:328-338.
- Eliason G, Abdel-Halim S, Arvidsson B, Kadi F, Piehl-Aulin K. Physical performance and muscular characteristics in different stages of COPD. Scandinavian journal of medicine & science in sports. 2009;19:865-870.
- Theriault ME, Pare ME, Maltais F, Debigare R. Satellite cells senescence in limb muscle of severe patients with COPD. PLoS One. 2012;7:e39124.
- Menon MK, Houchen L, Singh SJ, Morgan MD, Bradding P, Steiner MC. Inflammatory and satellite cells in the quadriceps of patients with COPD and response to resistance training. Chest. 2012;142:1134-1142.
- Rodriguez J, Vernus B, Chelh I, et al. Myostatin and the skeletal muscle atrophy and hypertrophy signaling pathways. Cellular and molecular life sciences : CMLS. 2014;71:4361-4371.

- Man WD, Natanek SA, Riddoch-Contreras J, et al. Quadriceps myostatin expression in COPD. Eur Respir J. 2010;36:686-688.
- Ju CR, Chen RC. Serum myostatin levels and skeletal muscle wasting in chronic obstructive pulmonary disease. Respir Med. 2012;106:102-108.
- Snijders T, Verdijk LB, Smeets JS, et al. The skeletal muscle satellite cell response to a single bout of resistance-type exercise is delayed with aging in men. Age. 2014;36:9699.
- Malaguti C, Nery LE, Dal Corso S, et al. Scaling skeletal muscle function to mass in patients with moderate-to-severe COPD. European journal of applied physiology. 2006;98:482-488.
- 87. Malaguti C, Napolis LM, Villaca D, Neder JA, Nery LE, Dal Corso S. Relationship between peripheral muscle structure and function in patients with chronic obstructive pulmonary disease with different nutritional status. Journal of strength and conditioning research / National Strength & Conditioning Association. 2011;25:1795-1803.
- van den Borst B, Slot IG, Hellwig VA, et al. Loss of quadriceps muscle oxidative phenotype and decreased endurance in patients with mildto-moderate COPD. J Appl Physiol (1985). 2013;114:1319-1328.
- Gosker HR, Zeegers MP, Wouters EF, Schols AM. Muscle fibre type shifting in the vastus lateralis of patients with COPD is associated with disease severity: a systematic review and meta-analysis. Thorax. 2007;62:944-949.
- Barreiro E, de la Puente B, Minguella J, et al. Oxidative stress and respiratory muscle dysfunction in severe chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2005;171:1116-1124.
- Degens H, Gayan-Ramirez G, van Hees HW. Smoking-induced skeletal muscle dysfunction: from evidence to mechanisms. Am J Respir Crit Care Med. 2015;191:620-625.
- 92. Gosker HR, Wouters EF, van der Vusse GJ, Schols AM. Skeletal muscle dysfunction in chronic obstructive pulmonary disease and chronic heart failure: underlying mechanisms and therapy perspectives. The American journal of clinical nutrition. 2000;71:1033-1047.
- Labeit S, Kohl CH, Witt CC, Labeit D, Jung J, Granzier H. Modulation of muscle atrophy, fatigue and MLC phosphorylation by MuRF1 as indicated by hindlimb suspension studies on MuRF1-KO mice. Journal of biomedicine & biotechnology.

2010;2010:693741.

- 94. Eddins MJ, Marblestone JG, Suresh Kumar KG, et al. Targeting the ubiquitin E3 ligase MuRF1 to inhibit muscle atrophy. Cell biochemistry and biophysics. 2011;60:113-118.
- 95. Edelmann MJ, Nicholson B, Kessler BM. Pharmacological targets in the ubiquitin system offer new ways of treating cancer, neurodegenerative disorders and infectious diseases. Expert reviews in molecular medicine. 2011;13:e35.
- 96. Gosker HR, Lencer NH, Franssen FM, van der Vusse GJ, Wouters EF, Schols AM. Striking similarities in systemic factors contributing to decreased exercise capacity in patients with severe chronic heart failure or COPD. Chest. 2003;123:1416-1424.
- Borgenvik M, Apro W, Blomstrand E. Intake of branched-chain amino acids influences the levels of MAFbx mRNA and MuRF-1 total protein in resting and exercising human muscle. American journal of physiology Endocrinology and metabolism. 2012;302:E510-521.
- Constantin D, Menon MK, Houchen-Wolloff L, et al. Skeletal muscle molecular responses to resistance training and dietary supplementation in COPD. Thorax. 2013;68:625-633.
- Hara K, Yonezawa K, Weng QP, Kozlowski MT, Belham C, Avruch J. Amino acid sufficiency and mTOR regulate p70 S6 kinase and eIF-4E BP1 through a common effector mechanism. The Journal of biological chemistry. 1998;273:14484-14494.
- Drummond MJ, Rasmussen BB. Leucine-enriched nutrients and the regulation of mammalian target of rapamycin signalling and human skeletal muscle protein synthesis. Curr Opin Clin Nutr Metab Care. 2008;11:222-226.
- Ditch S, Paull TT. The ATM protein kinase and cellular redox signaling: beyond the DNA damage response. Trends in biochemical sciences. 2012;37:15-22.
- Sabatini DM. mTOR and cancer: insights into a complex relationship. Nature reviews Cancer. 2006;6:729-734.
- 103. Lira VA, Okutsu M, Zhang M, et al. Autophagy is required for exercise training-induced skeletal muscle adaptation and improvement of physical performance. FASEB journal : official publication of the Federation of American Societies for Experimental Biology. 2013;27:4184-4193.

- 104. Sanchez AM, Bernardi H, Py G, Candau RB. Autophagy is essential to support skeletal muscle plasticity in response to endurance exercise. American journal of physiology Regulatory, integrative and comparative physiology. 2014;307:R956-969.
- 105. Couillard A, Maltais F, Saey D, et al. Exercise-induced quadriceps oxidative stress and peripheral muscle dysfunction in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2003;167:1664-1669.
- 106. Wilborn CD, Taylor LW, Greenwood M, Kreider RB, Willoughby DS. Effects of different intensities of resistance exercise on regulators of myogenesis. Journal of strength and conditioning research / National Strength & Conditioning Association. 2009;23:2179-2187.
- 107. Hill M, Goldspink G. Expression and splicing of the insulin-like growth factor gene in rodent muscle is associated with muscle satellite (stem) cell activation following local tissue damage. The Journal of physiology. 2003;549:409-418.
- 108. Snijders T, Verdijk LB, McKay BR, et al. Acute dietary protein intake restriction is associated with changes in myostatin expression after a single bout of resistance exercise in healthy young men. The Journal of nutrition. 2014;144:137-145.
- 109. Zhang L, Rajan V, Lin E, et al. Pharmacological inhibition of myostatin suppresses systemic inflammation and muscle atrophy in mice with chronic kidney disease. FASEB journal : official publication of the Federation of American Societies for Experimental Biology. 2011;25:1653-1663.
- Lach-Trifilieff E, Minetti GC, Sheppard K, et al. An antibody blocking activin type II receptors induces strong skeletal muscle hypertrophy and protects from atrophy. Molecular and cellular biology. 2014;34:606-618.
- 111. Bonetto A, Penna F, Minero VG, et al. Glutamine prevents myostatin hyperexpression and protein hypercatabolism induced in C2C12 myotubes by tumor necrosis factor-alpha. Amino acids. 2011;40:585-594.
- 112. Pan L, Wang M, Xie X, Du C, Guo Y. Effects of anabolic steroids on chronic obstructive pulmonary disease: a meta-analysis of randomised controlled trials. PLoS One. 2014;9:e84855.
- 113. Schols AM. Nutrition as a metabolic modulator in COPD. Chest. 2013;144:1340-1345.
- 114. Schoffelen PF, Westerterp KR, Saris WH, Ten

Hoor F. A dual-respiration chamber system with automated calibration. J Appl Physiol (1985). 1997;83:2064-2072.

- 115. Roberts SB. Use of the doubly labeled water method for measurement of energy expenditure, total body water, water intake, and metabolizable energy intake in humans and small animals. Canadian journal of physiology and pharmacology. 1989;67:1190-1198.
- Schoeller DA, Ravussin E, Schutz Y, Acheson KJ, Baertschi P, Jequier E. Energy expenditure by doubly labeled water: validation in humans and proposed calculation. The American journal of physiology. 1986;250:R823-830.
- 117. Hugli O, Schutz Y, Fitting JW. The daily energy expenditure in stable chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 1996;153:294-300.
- Slinde F, Ellegard L, Gronberg AM, Larsson S, Rossander-Hulthen L. Total energy expenditure in underweight patients with severe chronic obstructive pulmonary disease living at home. Clin Nutr. 2003;22:159-165.
- Baarends EM, Schols AM, Westerterp KR, Wouters EF. Total daily energy expenditure relative to resting energy expenditure in clinically stable patients with COPD. Thorax. 1997;52:780-785.
- 120. Mineo TC, Pompeo E, Mineo D, Ambrogi V, Ciarapica D, Polito A. Resting energy expenditure and metabolic changes after lung volume reduction surgery for emphysema. The Annals of thoracic surgery. 2006;82:1205-1211.
- 121. Kao CC, Hsu JW, Bandi V, Hanania NA, Kheradmand F, Jahoor F. Resting energy expenditure and protein turnover are increased in patients with severe chronic obstructive pulmonary disease. Metabolism: clinical and experimental. 2011;60:1449-1455.
- 122. Schols AM, Fredrix EW, Soeters PB, Westerterp KR, Wouters EF. Resting energy expenditure in patients with chronic obstructive pulmonary disease. The American journal of clinical nutrition. 1991;54:983-987.
- Goldstein S, Askanazi J, Weissman C, Thomashow B, Kinney JM. Energy expenditure in patients with chronic obstructive pulmonary disease. Chest. 1987;91:222-224.
- Cohen RI, Marzouk K, Berkoski P, O'Donnell CP, Polotsky VY, Scharf SM. Body composition and resting energy expenditure in clinically stable, non-

weight-losing patients with severe emphysema. Chest. 2003;124:1365-1372.

- Green JH, Muers MF. The thermic effect of food in underweight patients with emphysematous chronic obstructive pulmonary disease. Eur Respir J. 1991;4:813-819.
- 126. Vermeeren MA, Schols AM, Wouters EF. Effects of an acute exacerbation on nutritional and metabolic profile of patients with COPD. Eur Respir J. 1997;10:2264-2269.
- 127. Ryan CF, Road JD, Buckley PA, Ross C, Whittaker JS. Energy balance in stable malnourished patients with chronic obstructive pulmonary disease. Chest. 1993;103:1038-1044.
- Kurpad AV, Muthayya S, Vaz M. Consequences of inadequate food energy and negative energy balance in humans. Public health nutrition. 2005;8:1053-1076.
- 129. Fredrix EW, Staal-van den Brekel AJ, Wouters EF. Energy balance in nonsmall cell lung carcinoma patients before and after surgical resection of their tumors. Cancer. 1997;79:717-723.
- 130. Harvie MN, Campbell IT, Thatcher N, Baildam A. Changes in body composition in men and women with advanced nonsmall cell lung cancer (NSCLC) undergoing chemotherapy. Journal of human nutrition and dietetics : the official journal of the British Dietetic Association. 2003;16:323-326.
- 131. Staal-van den Brekel AJ, Schols AM, Dentener MA, ten Velde GP, Buurman WA, Wouters EF. Metabolism in patients with small cell lung carcinoma compared with patients with non-small cell lung carcinoma and healthy controls. Thorax. 1997;52:338-341.
- Cao DX, Wu GH, Zhang B, et al. Resting energy expenditure and body composition in patients with newly detected cancer. Clin Nutr. 2010;29:72-77.
- Van Remoortel H, Hornikx M, Demeyer H, et al. Daily physical activity in subjects with newly diagnosed COPD. Thorax. 2013;68:962-963.
- Andersson M, Slinde F, Gronberg AM, Svantesson U, Janson C, Emtner M. Physical activity level and its clinical correlates in chronic obstructive pulmonary disease: a cross-sectional study. Respiratory research. 2013;14:128.
- 135. Troosters T, Sciurba F, Battaglia S, et al. Physical inactivity in patients with COPD, a controlled multicenter pilot-study. Respir Med. 2010;104:1005-1011.
- 136. Pitta F, Troosters T, Spruit MA, Probst VS, Decramer

M, Gosselink R. Characteristics of physical activities in daily life in chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2005;171:972-977.

- 137. Ehsan M, Khan R, Wakefield D, et al. A longitudinal study evaluating the effect of exacerbations on physical activity in patients with chronic obstructive pulmonary disease. Annals of the American Thoracic Society. 2013;10:559-564.
- Pitta F, Troosters T, Probst VS, Spruit MA, Decramer M, Gosselink R. Physical activity and hospitalization for exacerbation of COPD. Chest. 2006;129:536-544.
- Waschki B, Spruit MA, Watz H, et al. Physical activity monitoring in COPD: compliance and associations with clinical characteristics in a multicenter study. Respir Med. 2012;106:522-530.
- 140. Ferguson GT, Fernandez E, Zamora MR, Pomerantz M, Buchholz J, Make BJ. Improved exercise performance following lung volume reduction surgery for emphysema. Am J Respir Crit Care Med. 1998;157:1195-1203.
- Hartman JE, Boezen HM, Heintzbergen S, et al. Daily physical activity after bronchoscopic lung volume reduction: a pilot study. Eur Respir J. 2012;40:1566-1567.
- Lahaije AJ, van Helvoort HA, Dekhuijzen PN, Heijdra YF. Physiologic limitations during daily life activities in COPD patients. Respir Med. 2010;104:1152-1159.
- 143. Vaes AW, Wouters EF, Franssen FM, et al. Taskrelated oxygen uptake during domestic activities of daily life in patients with COPD and healthy elderly subjects. Chest. 2011;140:970-979.
- Calvert LD, Steiner MC, Morgan MD, Singh SJ. Plasma ammonia response to incremental cycling and walking tests in COPD. Respir Med. 2010;104:675-681.
- Criner GJ, Belt P, Sternberg AL, et al. Effects of lung volume reduction surgery on gas exchange and breathing pattern during maximum exercise. Chest. 2009;135:1268-1279.
- Budweiser S, Heinemann F, Meyer K, Wild PJ, Pfeifer M. Weight gain in cachectic COPD patients receiving noninvasive positive-pressure ventilation. Respir Care. 2006;51:126-132.
- Granger CL, McDonald CF, Irving L, et al. Low physical activity levels and functional decline in individuals with lung cancer. Lung cancer. 2014;83:292-299.

- Hummler S, Thomas M, Hoffmann B, et al. Physical performance and psychosocial status in lung cancer patients: results from a pilot study. Oncology research and treatment. 2014;37:36-41.
- Hugli O, Frascarolo P, Schutz Y, Jequier E, Leuenberger P, Fitting JW. Diet-induced thermogenesis in chronic obstructive pulmonary disease. Am Rev Respir Dis. 1993;148:1479-1483.
- 150. Dore MF, Laaban JP, Orvoen-Frija E, Kouchakji B, Joubert M, Rochemaure J. Role of the thermic effect of food in malnutrition of patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 1997;155:1535-1540.
- Tang NL, Chung ML, Elia M, et al. Total daily energy expenditure in wasted chronic obstructive pulmonary disease patients. European journal of clinical nutrition. 2002;56:282-287.
- Schols A, Mostert R, Cobben N, Soeters P, Wouters E. Transcutaneous oxygen saturation and carbon dioxide tension during meals in patients with chronic obstructive pulmonary disease. Chest. 1991;100:1287-1292.
- Broekhuizen R, Wouters EF, Creutzberg EC, Schols AM. Raised CRP levels mark metabolic and functional impairment in advanced COPD. Thorax. 2006;61:17-22.
- Staal-van den Brekel AJ, Schols AM, ten Velde GP, Buurman WA, Wouters EF. Analysis of the energy balance in lung cancer patients. Cancer research. 1994;54:6430-6433.
- 155. Langen RC, Haegens A, Vernooy JH, et al. NFkappaB activation is required for the transition of pulmonary inflammation to muscle atrophy. American journal of respiratory cell and molecular biology. 2012;47:288-297.
- 156. Tkacova R, Ukropec J, Skyba P, et al. Increased adipose tissue expression of proinflammatory CD40, MKK4 and JNK in patients with very severe chronic obstructive pulmonary disease. Respiration; international review of thoracic diseases. 2011;81:386-393.
- 157. Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus. Lancet Oncol. 2011;12:489-495.
- Dahlman I, Mejhert N, Linder K, et al. Adipose tissue pathways involved in weight loss of cancer cachexia. British journal of cancer. 2010;102:1541-1548.
- 159. Schols AM, Creutzberg EC, Buurman WA, Campfield LA, Saris WH, Wouters EF. Plasma

leptin is related to proinflammatory status and dietary intake in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 1999;160:1220-1226.

- 160. Brusik M, Ukropec J, Joppa P, et al. Circulatory and adipose tissue leptin and adiponectin in relationship to resting energy expenditure in patients with chronic obstructive pulmonary disease. Physiological research / Academia Scientiarum Bohemoslovaca. 2012;61:469-480.
- Kajimura S, Saito M. A new era in brown adipose tissue biology: molecular control of brown fat development and energy homeostasis. Annu Rev Physiol. 2014;76:225-249.
- Cinti S. Transdifferentiation properties of adipocytes in the adipose organ. American journal of physiology Endocrinology and metabolism. 2009;297:E977-986.
- Petruzzelli M, Schweiger M, Schreiber R, et al. A switch from white to brown fat increases energy expenditure in cancer-associated cachexia. Cell metabolism. 2014;20:433-447.
- Vijgen GH, Bouvy ND, Teule GJ, Brans B, Schrauwen P, van Marken Lichtenbelt WD. Brown adipose tissue in morbidly obese subjects. PLoS One. 2011;6:e17247.
- 165. Yoneshiro T, Aita S, Matsushita M, et al. Age-related decrease in cold-activated brown adipose tissue and accumulation of body fat in healthy humans. Obesity. 2011;19:1755-1760.
- 166. Lee P, Greenfield JR, Ho KK, Fulham MJ. A critical appraisal of the prevalence and metabolic significance of brown adipose tissue in adult humans. American journal of physiology Endocrinology and metabolism. 2010;299:E601-606.
- Cypess AM, Lehman S, Williams G, et al. Identification and importance of brown adipose tissue in adult humans. The New England journal of medicine. 2009;360:1509-1517.
- Shellock FG, Riedinger MS, Fishbein MC. Brown adipose tissue in cancer patients: possible cause of cancer-induced cachexia. J Cancer Res Clin Oncol. 1986;111:82-85.
- 169. Huang YC, Chen TB, Hsu CC, et al. The relationship between brown adipose tissue activity and neoplastic status: an (18)F-FDG PET/CT study in the tropics. Lipids Health Dis. 2011;10:238.
- 170. Saure EW, Eagan TM, Jensen RL, et al. Explained variance for blood gases in a population with

COPD. The clinical respiratory journal. 2012;6:72-80.

- 171. Nakayama K, Frew IJ, Hagensen M, et al. Siah2 regulates stability of prolyl-hydroxylases, controls HIF1alpha abundance, and modulates physiological responses to hypoxia. Cell. 2004;117:941-952.
- 172. van den Borst B, Schols AM, de Theije C, et al. Characterization of the inflammatory and metabolic profile of adipose tissue in a mouse model of chronic hypoxia. J Appl Physiol (1985). 2013;114:1619-1628.
- 173. Sun K, Wernstedt Asterholm I, Kusminski CM, et al. Dichotomous effects of VEGF-A on adipose tissue dysfunction. Proceedings of the National Academy of Sciences of the United States of America. 2012;109:5874-5879.
- 174. Elias I, Franckhauser S, Ferre T, et al. Adipose tissue overexpression of vascular endothelial growth factor protects against diet-induced obesity and insulin resistance. Diabetes. 2012;61:1801-1813.
- 175. Franssen FM, Sauerwein HP, Ackermans MT, Rutten EP, Wouters EF, Schols AM. Increased postabsorptive and exercise-induced wholebody glucose production in patients with chronic obstructive pulmonary disease. Metabolism: clinical and experimental. 2011;60:957-964.
- 176. Engelen MP, Schols AM, Does JD, Deutz NE, Wouters EF. Altered glutamate metabolism is associated with reduced muscle glutathione levels in patients with emphysema. Am J Respir Crit Care Med. 2000;161:98-103.
- 177. Engelen MP, Schols AM, Does JD, Gosker HR, Deutz NE, Wouters EF. Exercise-induced lactate increase in relation to muscle substrates in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2000;162:1697-1704.
- 178. Perez de Heredia F, Wood IS, Trayhurn P. Hypoxia stimulates lactate release and modulates monocarboxylate transporter (MCT1, MCT2, and MCT4) expression in human adipocytes. Pflugers Archiv : European journal of physiology. 2010;459:509-518.
- Carriere A, Jeanson Y, Berger-Muller S, et al. Browning of white adipose cells by intermediate metabolites: an adaptive mechanism to alleviate redox pressure. Diabetes. 2014;63:3253-3265.
- Virtanen KA, Lidell ME, Orava J, et al. Functional brown adipose tissue in healthy adults. The New England journal of medicine. 2009;360:1518-1525.
- 181. Soderlund V, Larsson SA, Jacobsson H. Reduction

of FDG uptake in brown adipose tissue in clinical patients by a single dose of propranolol. European journal of nuclear medicine and molecular imaging. 2007;34:1018-1022.

- 182. Hofford JM, Milakofsky L, Vogel WH, Sacher RS, Savage GJ, Pell S. The nutritional status in advanced emphysema associated with chronic bronchitis. A study of amino acid and catecholamine levels. Am Rev Respir Dis. 1990;141:902-908.
- 183. Schiffelers SL, Blaak EE, Baarends EM, et al. beta-Adrenoceptor-mediated thermogenesis and lipolysis in patients with chronic obstructive pulmonary disease. American journal of physiology Endocrinology and metabolism. 2001;280:E357-364.
- 184. Vosselman MJ, van der Lans AA, Brans B, et al. Systemic beta-adrenergic stimulation of thermogenesis is not accompanied by brown adipose tissue activity in humans. Diabetes. 2012;61:3106-3113.
- 185. Parysow O, Mollerach AM, Jager V, Racioppi S, San Roman J, Gerbaudo VH. Low-dose oral propranolol could reduce brown adipose tissue F-18 FDG uptake in patients undergoing PET scans. Clinical nuclear medicine. 2007;32:351-357.
- 186. Lee H, Kim S, Lim Y, et al. Nutritional status and disease severity in patients with chronic obstructive pulmonary disease (COPD). Archives of gerontology and geriatrics. 2013;56:518-523.
- 187. Goris AH, Vermeeren MA, Wouters EF, Schols AM, Westerterp KR. Energy balance in depleted ambulatory patients with chronic obstructive pulmonary disease: the effect of physical activity and oral nutritional supplementation. Br J Nutr. 2003;89:725-731.
- Schols AM, Soeters PB, Mostert R, Saris WH, Wouters EF. Energy balance in chronic obstructive pulmonary disease. Am Rev Respir Dis. 1991;143:1248-1252.
- 189. Creutzberg EC, Wouters EF, Vanderhoven-Augustin IM, Dentener MA, Schols AM. Disturbances in leptin metabolism are related to energy imbalance during acute exacerbations of chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2000;162:1239-1245.
- 190. Bruera E. ABC of palliative care. Anorexia, cachexia, and nutrition. BMJ. 1997;315:1219-1222.
- 191. Pilhatsch M, Scheuing H, Kroemer N, et al. Nicotine administration in healthy non-smokers reduces appetite but does not alter plasma ghrelin. Human

psychopharmacology. 2014;29:384-387.

- 192. Eckerblad J, Todt K, Jakobsson P, et al. Symptom burden in stable COPD patients with moderate or severe airflow limitation. Heart & lung : the journal of critical care. 2014;43:351-357.
- Hilmarsen CW, Wilke S, Engan H, et al. Impact of symptoms of anxiety and depression on COPD Assessment Test scores. Eur Respir J. 2014;43:898-900.
- 194. Gronberg AM, Slinde F, Engstrom CP, Hulthen L, Larsson S. Dietary problems in patients with severe chronic obstructive pulmonary disease. Journal of human nutrition and dietetics : the official journal of the British Dietetic Association. 2005;18:445-452.
- 195. Sarna L, Lindsey AM, Dean H, Brecht ML, McCorkle R. Weight change and lung cancer: relationships with symptom distress, functional status, and smoking. Research in nursing & health. 1994;17:371-379.
- 196. Iyer S, Roughley A, Rider A, Taylor-Stokes G. The symptom burden of non-small cell lung cancer in the USA: a real-world cross-sectional study. Supportive care in cancer : official journal of the Multinational Association of Supportive Care in Cancer. 2014;22:181-187.
- 197. Bentsen SB, Rustoen T, Miaskowski C. Prevalence and characteristics of pain in patients with chronic obstructive pulmonary disease compared to the Norwegian general population. The journal of pain : official journal of the American Pain Society. 2011;12:539-545.
- 198. Roberts MH, Mapel DW, Hartry A, Von Worley A, Thomson H. Chronic pain and pain medication use in chronic obstructive pulmonary disease. A crosssectional study. Annals of the American Thoracic Society. 2013;10:290-298.
- Ketwaroo GA, Cheng V, Lembo A. Opioid-induced bowel dysfunction. Curr Gastroenterol Rep. 2013;15:344.
- 200. Portenoy RK, Ahmed E. Principles of opioid use in cancer pain. J Clin Oncol. 2014;32:1662-1670.
- Jamshed N, Lee ZE, Olden KW. Diagnostic approach to chronic constipation in adults. Am Fam Physician. 2011;84:299-306.
- 202. Tielemans MM, Jaspers Focks J, van Rossum LG, et al. Gastrointestinal symptoms are still prevalent and negatively impact health-related quality of life: a large cross-sectional population based study in The Netherlands. PLoS One. 2013;8:e69876.

- Veldhuizen MG, Albrecht J, Zelano C, Boesveldt S, Breslin P, Lundstrom JN. Identification of human gustatory cortex by activation likelihood estimation. Human brain mapping. 2011;32:2256-2266.
- Oliveira-Maia AJ, Roberts CD, Simon SA, Nicolelis MA. Gustatory and reward brain circuits in the control of food intake. Advances and technical standards in neurosurgery. 2011;36:31-59.
- Norden J, Gronberg AM, Bosaeus I, et al. Nutrition impact symptoms and body composition in patients with COPD. European journal of clinical nutrition. 2014.
- Dewan NA, Bell CW, Moore J, Anderson B, Kirchain W, O'Donohue WJ, Jr. Smell and taste function in subjects with chronic obstructive pulmonary disease. Effect of long-term oxygen via nasal cannulas. Chest. 1990;97:595-599.
- 207. Wardwell L, Chapman-Novakofski K, Brewer MS. Effects of age, gender and chronic obstructive pulmonary disease on taste acuity. International journal of food sciences and nutrition. 2009;60 Suppl 6:84-97.
- Janssen P, Vanden Berghe P, Verschueren S, Lehmann A, Depoortere I, Tack J. Review article: the role of gastric motility in the control of food intake. Aliment Pharmacol Ther. 2011;33:880-894.
- Karapanagiotou EM, Tsochatzis EA, Dilana KD, Tourkantonis I, Gratsias I, Syrigos KN. The significance of leptin, adiponectin, and resistin serum levels in non-small cell lung cancer (NSCLC). Lung cancer. 2008;61:391-397.
- Terzidis A, Sergentanis TN, Antonopoulos G, et al. Elevated serum leptin levels: a risk factor for nonsmall-cell lung cancer? Oncology. 2009;76:19-25.
- Itoh T, Nagaya N, Yoshikawa M, et al. Elevated plasma ghrelin level in underweight patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2004;170:879-882.
- 212. Karapanagiotou EM, Polyzos A, Dilana KD, et al. Increased serum levels of ghrelin at diagnosis mediate body weight loss in non-small cell lung cancer (NSCLC) patients. Lung cancer. 2009;66:393-398.
- 213. Sescousse G, Caldu X, Segura B, Dreher JC. Processing of primary and secondary rewards: a quantitative meta-analysis and review of human functional neuroimaging studies. Neurosci Biobehav Rev. 2013;37:681-696.
- 214. Murdaugh DL, Cox JE, Cook EW, 3rd, Weller RE.

fMRI reactivity to high-calorie food pictures predicts short- and long-term outcome in a weight-loss program. Neuroimage. 2012;59:2709-2721.

- 215. Patra SK, Arora S. Integrative role of neuropeptides and cytokines in cancer anorexia-cachexia syndrome. Clin Chim Acta. 2012;413:1025-1034.
- 216. Luo N, Marcelin G, Liu SM, Schwartz G, Chua S, Jr. Neuropeptide Y and agouti-related peptide mediate complementary functions of hyperphagia and reduced energy expenditure in leptin receptor deficiency. Endocrinology. 2011;152:883-889.
- 217. Briggs DI, Andrews ZB. Metabolic status regulates ghrelin function on energy homeostasis. Neuroendocrinology. 2011;93:48-57.
- 218. Sanchez-Lara K, Arrieta O, Pasaye E, et al. Brain activity correlated with food preferences: a functional study comparing advanced nonsmall cell lung cancer patients with and without anorexia. Nutrition. 2013;29:1013-1019.
- 219. Sahebjami H, Vassallo CL. Influence of starvation on enzyme-induced emphysema. Journal of applied physiology: respiratory, environmental and exercise physiology. 1980;48:284-288.
- Sahebjami H, MacGee J. Effects of starvation on lung mechanics and biochemistry in young and old rats. J Appl Physiol (1985). 1985;58:778-784.
- 221. Sahebjami H, Domino M. Effects of starvation and refeeding on elastase-induced emphysema. J Appl Physiol (1985). 1989;66:2611-2616.
- Bishai JM, Mitzner W. Effect of severe calorie restriction on the lung in two strains of mice. American journal of physiology Lung cellular and molecular physiology. 2008;295:L356-362.
- Coxson HO, Chan IH, Mayo JR, Hlynsky J, Nakano Y, Birmingham CL. Early emphysema in patients with anorexia nervosa. Am J Respir Crit Care Med. 2004;170:748-752.
- 224. Thurlbeck WM. Diaphragm and body weight in emphysema. Thorax. 1978;33:483-487.
- Golgeli A, Ozesmi C, Suer C. Effect of acute starvation on rat diaphragm function. Jpn J Physiol. 1994;44:743-747.
- 226. Efthimiou J, Fleming J, Gomes C, Spiro SG. The effect of supplementary oral nutrition in poorly nourished patients with chronic obstructive pulmonary disease. Am Rev Respir Dis. 1988;137:1075-1082.
- 227. Weekes CE, Emery PW, Elia M. Dietary counselling and food fortification in stable COPD: a randomised trial. Thorax. 2009;64:326-331.
- 228. Engelen MP, Rutten EP, De Castro CL, Wouters EF,

Schols AM, Deutz NE. Casein protein results in higher prandial and exercise induced whole body protein anabolism than whey protein in chronic obstructive pulmonary disease. Metabolism: clinical and experimental. 2012;61:1289-1300.

- 229. Engelen MP, De Castro CL, Rutten EP, Wouters EF, Schols AM, Deutz NE. Enhanced anabolic response to milk protein sip feeding in elderly subjects with COPD is associated with a reduced splanchnic extraction of multiple amino acids. Clin Nutr. 2012;31:616-624.
- 230. van de Bool C, Mattijssen-Verdonschot C, van Melick PP, et al. Quality of dietary intake in relation to body composition in patients with chronic obstructive pulmonary disease eligible for pulmonary rehabilitation. European journal of clinical nutrition. 2014;68:159-165.
- Rutten EP, Lenaerts K, Buurman WA, Wouters EF. Disturbed intestinal integrity in patients with COPD: effects of activities of daily living. Chest. 2014;145:245-252.
- 232. Engelen MP, Rutten EP, De Castro CL, Wouters EF, Schols AM, Deutz NE. Altered interorgan response to feeding in patients with chronic obstructive pulmonary disease. The American journal of clinical nutrition. 2005;82:366-372.
- 233. Pouw EM, Schols AM, Deutz NE, Wouters EF. Plasma and muscle amino acid levels in relation to resting energy expenditure and inflammation in stable chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 1998;158:797-801.
- Kojima M, Hosoda H, Date Y, Nakazato M, Matsuo H, Kangawa K. Ghrelin is a growth-hormonereleasing acylated peptide from stomach. Nature. 1999;402:656-660.
- 235. Sun Y, Wang P, Zheng H, Smith RG. Ghrelin stimulation of growth hormone release and appetite is mediated through the growth hormone secretagogue receptor. Proceedings of the National Academy of Sciences of the United States of America. 2004;101:4679-4684.
- 236. Miki K, Maekura R, Nagaya N, et al. Effects of ghrelin treatment on exercise capacity in underweight COPD patients: a substudy of a multicenter, randomized, double-blind, placebo-controlled trial of ghrelin treatment. BMC Pulm Med. 2013;13:37.
- 237. Garcia JM, Boccia RV, Graham CD, et al. Anamorelin for patients with cancer cachexia: an integrated analysis of two phase 2, randomised, placebocontrolled, double-blind trials. Lancet Oncol. 2014.

- Currow DC, Abernethy AP. Anamorelin hydrochloride in the treatment of cancer anorexiacachexia syndrome. Future Oncol. 2014;10:789-802.
- 239. Ziauddeen H, Farooqi IS, Fletcher PC. Obesity and the brain: how convincing is the addiction model? Nature reviews Neuroscience. 2012;13:279-286.
- Stice E, Yokum S, Burger KS, Epstein LH, Small DM. Youth at risk for obesity show greater activation of striatal and somatosensory regions to food. J Neurosci. 2011;31:4360-4366.
- 241. Roefs A, Stapert D, Isabella LA, Wolters G, Wojciechowski F, Jansen A. Early associations with food in anorexia nervosa patients and obese people assessed in the affective priming paradigm. Eat Behav. 2005;6:151-163.
- 242. Zipfel S, Wild B, Gross G, et al. Focal psychodynamic therapy, cognitive behaviour therapy, and optimised treatment as usual in outpatients with anorexia nervosa (ANTOP study): randomised controlled trial. Lancet. 2014;383:127-137.
- 243. Deckersbach T, Das SK, Urban LE, et al. Pilot randomized trial demonstrating reversal of obesity-related abnormalities in reward system responsivity to food cues with a behavioral intervention. Nutr Diabetes. 2014;4:e129.



5

Computed tomography derived muscle remodelling after bronchoscopic lung volume reduction in advanced emphysema

Karin J.C. Sanders, Karin Klooster, Lowie E.G.W. Vanfleteren, Dirk-Jan Slebos, Annemie M.W.J. Schols.

Thorax. 2019;74(2):206-07.

ABSTRACT

Muscle wasting frequently occurs in severe emphysema. Improving respiratory mechanics by bronchoscopic lung volume reduction using endobronchial valves (EBV) might prevent further loss or even increase muscle mass.

CT-derived skeletal muscle mass gain was observed in 39/49 patients 6 months after EBV. Multiple linear regression showed that gain in muscle (β =2.4 [95%CI 0.2-4.6]; p=0.036) and intramuscular fat (β =3.1 [95%CI 0.2-5.9]; p=0.035) associated with improved 6MWD independent of the change in residual volume.

Skeletal muscle remodelling associates with improved exercise capacity after EBV, independent of hyperinflation reduction.

INTRODUCTION

The emphysematous COPD phenotype is prone to develop cachexia (i.e., unintended weight loss and muscle wasting) [1], adversely affecting morbidity and mortality [1,2]. Patients with severe emphysema have limited treatment options and the efficacy of pulmonary rehabilitation on muscle function and exercise performance is hampered by impaired respiratory mechanics. Recently, lung volume reduction has regained new interest by the arrival of non-invasive bronchoscopic lung volume reduction. Bronchoscopic placement of endobronchial valves (EBV) has been shown to significantly improve symptoms, exercise capacity and quality of life [3].

The impairment of exercise tolerance in severe emphysema results from ventilatory limitation, deconditioning and loss of muscle mass and oxidative metabolism [4,5]. While improved lung function after EBV treatment was accompanied by increased exercise capacity and physical activity [3,6], the effect of EBV on muscle loss is unknown.

We hypothesize that hyperinflation reduction results in restoration of energy balance which increases body weight and maintains or increases muscle mass contributing to improved exercise capacity.

RESULTS

CT scans from 49 enrolled patients were eligible (supplemental figure 1). Baseline characteristics are displayed in supplemental table 1.

Changes in skeletal muscle and adipose tissue are depicted in table 1. After EBV treatment body weight, skeletal muscle, intramuscular fat and subcutaneous fat cross-sectional area significantly increased, while mean skeletal muscle Hounsfield units remained unchanged. Despite a similar improvement in lung function parameters, muscle mass was maintained or increased in the majority of patients (80%), whereas 20% experienced muscle loss.

Table 1. Change in skeletal muscle, adipose tissue and clinical outcome.

	Absolute change	p value
Weight, 95% CI (kg)	1.3 (0.5 to 2.1)	0.001
Skeletal muscle cross-sectional area, median (range) (cm ²)	2.4 (-14.4 to 16.6)	0.002
Hounsfield unit skeletal muscle, 95% CI	-0.1 (-1.3 to 0.6)	0.882
Intramuscular fat cross-sectional area, median (range) (cm ²)	1.3 (-7.0 to 23.8)	0.017
Subcutaneous fat cross-sectional area, median (range) (cm ²)	5.3 (-68.8 to 38.3)	0.014
Residual volume, 95% CI (%)	-17.0 (-20.5 to -13.5)	<0.001

Abbreviations: 6MWD, 6 minute walk distance.

Patients significantly improved in residual volume and their 6MWD after EBV treatment as previously reported for the total cohort [3]. Using multiple linear regression analysis, adjusting for gender and baseline 6MWD, skeletal muscle mass (β =2.4 [95% CI 0.2-4.6]; p=0.036) and intramuscular fat (β =3.1 [95% CI 0.2-5.9]; p=0.035) but not the change in residual volume, associated with 6MWD improvement (table 2).

	В	95% CI	p value
Delta skeletal muscle cross-sectional area, cm ²	2.4	0.16 to 4.63	0.036
Delta intramuscular fat cross-sectional area, cm ²	3.1	0.23 to 5.94	0.035
6MWD baseline	-0.3	-0.49 to -0.17	< 0.001
Delta residual volume, %	-0.7	-1.93 to 0.48	0.232

Table 2. Multivariate analysis for	 predictors of change in 6MWD, 	, corrected for gender (N=45)
------------------------------------	---	-------------------------------

Abbreviations: 6MWD, 6 minute walk distance.

DISCUSSION

This is to our knowledge the first study to present changes in skeletal muscle and adipose tissue cross-sectional area and distribution after EBV in patients with advanced emphysema derived from chest CT analysis. Skeletal muscle remodelling associated with increased exercise capacity, independent of the degree of reduced hyperinflation.

Mineo et al. [8] previously investigated body composition in emphysematous patients by dualenergy x-ray absorptiometry, prior to and 1, 3 and 5 year following lung volume reduction surgery. A sustained increase of whole body fat mass and fat-free mass was reported in patients treated with lung volume reduction surgery, while a decline in fat and fat-free mass was observed after 3 and 5 years in patients who received usual care. Dual-energy x-ray absorptiometry analyses however do not allow direct quantification of muscle mass or intramuscular changes.

A remarkable observation in this study was the observed increase in intramuscular fat associated with improved exercise performance independent of muscle cross-sectional area. The observed muscle remodelling may enhance exercise performance by improving muscle strength but also by alleviating fatigue. In line with the athlete's paradox, the observed increase in intramuscular adipose tissue could reflect improved muscle mitochondrial metabolism and insulin sensitivity via an effect of peroxisome proliferator-activated receptor gamma coactivator-1 on intramuscular lipid programming [9]. Supportive for this hypothesis are two studies by Mineo et al. showing that resting metabolic rate after lung volume reduction surgery was associated with a conversion from prevalent lipid to carbohydrate metabolism [8] and demonstrating a reduction in insulin resistance [10], but obviously this requires further in depth investigation.

Although the strength of our study comes from the well-defined randomized patient cohort, there are some limitations. Gender distribution in the study population was unequal with a predominance of females. We adjusted for gender in statistical analyses but the sample size did not allow investigating putative gender differences in the response. We do not expect profound differences as the previous studies by Mineo et al. [8,10] were performed in males. Furthermore, while the standard site to measure muscle tissue as proxy for whole body muscle mass is the third lumbar vertebra, this is typically outside the field of view of clinically acquired CT imaging in patients with COPD. Therefore, in this study we analysed CT scans at the level of the first lumbar vertebra. This hampered analysis of visceral fat mass, because CT scans at this level frequently include the lower lung lobes. Furthermore, a minimal clinically important difference is not yet been determined. We furthermore recognize that CT slice derived intramuscular fat cannot distinguish between intramyocellular and extramyocellular lipids. This requires additional biopsy derived muscle cross-sectional area as proxy for muscle fat deposits, which might be due to the sample size.

In conclusion, bronchoscopic lung volume reduction treatment using one-way endobronchial valves induces muscle remodelling in severe emphysematous patients which associates with improvements in exercise performance.

REFERENCES

- Vanfleteren LE, Spruit MA, Groenen M, et al. Clusters of comorbidities based on validated objective measurements and systemic inflammation in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med. 2013;187:728-735.
- Burgel PR, Paillasseur JL, Peene B, et al. Two distinct chronic obstructive pulmonary disease (COPD) phenotypes are associated with high risk of mortality. PloS one. 2012;7:e51048.
- Klooster K, ten Hacken NH, Hartman JE, Kerstjens HA, van Rikxoort EM, Slebos DJ. Endobronchial Valves for Emphysema without Interlobar Collateral Ventilation. N Engl J Med. 2015;373:2325-2335.
- Remels AH, Schrauwen P, Broekhuizen R, et al. Peroxisome proliferator-activated receptor expression is reduced in skeletal muscle in COPD. Eur Respir J. 2007;30:245-252.
- Pitta F, Troosters T, Probst VS, Spruit MA, Decramer M, Gosselink R. Physical activity and hospitalization for exacerbation of COPD. Chest. 2006;129:536-544.
- Hartman JE, Klooster K, Slebos DJ, Ten Hacken NH. Improvement of physical activity after endobronchial valve treatment in emphysema patients. Respir Med. 2016;117:116-121.
- Goodpaster BH, Kelley DE, Thaete FL, He J, Ross R. Skeletal muscle attenuation determined by computed tomography is associated with skeletal muscle lipid content. J Appl Physiol (1985). 2000;89:104-110.
- Mineo D, Ambrogi V, Lauriola V, Pompeo E, Mineo TC. Recovery of body composition improves longterm outcomes after lung volume reduction surgery for emphysema. Eur Respir J. 2010;36:408-416.
- Koves TR, Sparks LM, Kovalik JP, et al. PPARgamma coactivator-1alpha contributes to exerciseinduced regulation of intramuscular lipid droplet programming in mice and humans. J Lipid Res. 2013;54:522-534.
- Mineo D, Ambrogi V, Frasca L, Cufari ME, Pompeo E, Mineo TC. Effects of lung volume reduction surgery for emphysema on glycolipidic hormones. Chest. 2008;134:30-37.

SUPPLEMENTALS

Study population and study design

A post-hoc analysis of a randomized controlled crossover trial investigating endobronchial valve (EBV) treatment conducted at the University Medical Centre Groningen in the Netherlands from June 2011 until November 2014 (STELVIO trial NTR2876) was performed. The full and detailed methodology of this study has been published previously [3]. In short, patients with advanced emphysema and a confirmed absence of collateral ventilation by the Chartis measurement were included. Only EBV treated patients who fully completed the study were eligible for analysis in this study. The initial trial randomized patients to active treatment (EBV group), or control (control group), with at 6 months after randomisation crossover to active treatment for the control group. CT images were obtained at baseline and 6 months follow-up after EBV treatment. Spirometry and bodyplethysmography were performed according to the ATS/ERS guidelines [11]. Furthermore, exercise capacity was measured by a 6-minute walk distance (6MWD) test according to the ATS guidelines [12]. Height and body weight were assessed. All tests were performed at baseline and 6 months follow-up after EBV treatment. The study was approved by the ethics committee of the University Medical Centre Groningen.

Image analysis

In this post-hoc analysis of the STELVIO trial skeletal muscle and adipose tissue were analysed on CT scan by assessment of the cross-sectional area at the first lumbar level (L1). Skeletal muscle cross-sectional area, intramuscular fat cross-sectional area and subcutaneous fat crosssectional area were analysed with Slice-O-Matic software v5.0 (Tomovision, Montreal, Canada). One image was selected for each patient. During anatomical land marking, the first image at L1 with both vertebral transverse processes clearly visible, was used in the analysis. The skeletal muscle measurements included the psoas, erector spinae, quadratus lumborum, transversus abdominis, external and internal obligue, and rectus abdominis muscles. Cross-sectional area of these structures were quantified by one assessor on the basis of pre-established thresholds of Hounsfield units (skeletal muscle -29 to 150, subcutaneous fat -190 to -30, and intramuscular fat -190 to -30). Boundaries were corrected manually as necessary. Changes in cross-sectional area between CT scans were expressed in squared centimetres. Our group found a mean coefficient of variation between observers of 1.3% for skeletal muscle area [data unpublished], which is in line with a variation of 0-2% in other studies [13-15]. Therefore, changes of larger than -1.3% were considered as 'loss of skeletal muscle', while changes smaller than -1.3% were considered 'maintenance or gain of skeletal muscle'. Additionally, the mean Hounsfield units of the muscle cross-sectional area were assessed, as a measure for muscle fat deposits. Low values reflect increased intramuscular fat.

Statistical analyses

Patients were included if the CT scan contained L1. Descriptive statistics of demographic and clinical variables were obtained. Means (95% confidence interval) were provided for continuous normally distributed variables, median (range) for continuous not-normally distributed variables and percentages were shown for categorical variables. Baseline and 6 months follow up measurements were compared with a paired-samples t test or Wilcoxon signed-rank test. A multiple linear regression model was constructed to assess the contribution of body composition changes to changes of 6MWD (6 months follow-up compared to baseline). The model was constructed including covariates that were found significant in univariate analyses, i.e. change in skeletal muscle cross-sectional area, change in intramuscular fat cross-sectional area, 6MWD at baseline, change in residual volume and gender. Data were tested for multicollinearity and possible influential outliers. All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, Version 24.0, IBM, Armonk, NY). Results with two-sided p values (<0.05) were considered statistically significant.



Abbreviations: EBV, endobronchial valve; CT, computed tomography.

Male/Female, N	16/33
Age, mean (range),years	59 (42 to 76)
Pack years	37 (31 to 43)
Length, m	1.69 (1.67 to 1.72)
Weight, kg	70.1 (66.5 to 73.7)
BMI, kg/m ²	24.4 (23.4 to 25.5)
FEV1	
Litres	0.8 (0.8 to 0.9)
% of predicted value	30.3 (28.0 to 32.6)
FVC	
Litres	2.6 (2.4 to 2.9)
% of predicted value	78.1 (72.9 to 83.3)
FEV1/FVC	32.6 (30.6 to 34.6)
RV	
Litres	4.5 (4.2 to 4.8)
% of predicted value	215.2 (205.3 to 225.0)
TLC	
Litres	7.5 (7.1 to 7.9)
% of predicted value	130.8 (127.3 to 134.2)
RV/TLC	60.3 (57.7 to 62.8)
6MWD, meter	357.6 (332.8 to 382.3)
Skeletal muscle cross-sectional area, median (range)(cm ²)	88.0 (55.6 to 162.4)

Supplemental table 1. Patient characteristics (N=49).

Data are represented as mean (95% confidence interval), unless stated otherwise. Abbreviations: BMI, body mass index; FEV1, forced expiratory volume in 1 second; FVC, forced vital capacity; RV, residual volume; TLC, total lung capacity; 6MWD, 6 minute walk distance.

REFERENCES

- Klooster K, ten Hacken NH, Hartman JE, Kerstjens HA, van Rikxoort EM, Slebos DJ. Endobronchial Valves for Emphysema without Interlobar Collateral Ventilation. N Engl J Med. 2015;373:2325-2335.
- Miller MR, Hankinson J, Brusasco V, et al. Standardisation of spirometry. Eur Respir J. 2005;26:319-338.
- Brooks D, Solway S, Gibbons WJ. ATS statement on six-minute walk test. Am J Respir Crit Care Med. 2003;167:1287.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008;9:629-635.
- Mourtzakis M, Prado CM, Lieffers JR, Reiman T, McCargar LJ, Baracos VE. A practical and precise approach to quantification of body composition in cancer patients using computed tomography images acquired during routine care. Appl Physiol Nutr Metab. 2008;33:997-1006.
- Irving BA, Weltman JY, Brock DW, Davis CK, Gaesser GA, Weltman A. NIH ImageJ and Slice-O-Matic computed tomography imaging software to quantify soft tissue. Obesity (Silver Spring). 2007;15:370-376.

Muscle remodelling after bronchoscopic lung volume reduction



6

Effect of bronchoscopic lung volume reduction in advanced emphysema on energy balance regulation

Karin J.C. Sanders, K. Klooster, Lowie E.G.W. Vanfleteren, Guy Plasqui, Anne-Marie C. Dingemans, Dirk-Jan Slebos, Annemie M.W.J. Schols.

Accepted.

ABSTRACT

Background

Hypermetabolism and muscle wasting frequently occur in patients with severe emphysema. Improving respiratory mechanics by bronchoscopic lung volume reduction (BLVR) might contribute to muscle maintenance by decreasing energy requirements and alleviating eatingrelated dyspnoea. The goal was to assess the impact of BLVR on energy balance regulation.

Design

Twenty emphysematous subjects participated in a controlled clinical experiment before and 6 months after BLVR. Energy requirements were assessed: basal metabolic rate (BMR) by ventilated hood, total daily energy expenditure (TDEE) by doubly labelled water, whole body fat-free mass (FFM) by deuterium dilution and physical activity by accelerometry. Oxygen saturation, breathing rate and heart rate were monitored before, during and after a standardized meal via pulse oxymetry and dyspnoea was rated.

Results

Sixteen patients completed follow-up and among those 10 patients exceeded the minimal clinically important difference of residual volume reduction. Residual volume was reduced with median (range) 1285 ml (-2430, -540). Before BLVR, 90% of patients was FFM depleted despite a normal BMI (24.3 \pm 4.3kg/m²). BMR was elevated by 130%. TDEE/BMR was 1.4 \pm 0.2 despite a very low median (range) daily step count of 2188 (739, 7110). Following BLVR, the components of energy metabolism did not change significantly after intervention compared to before intervention but BLVR treatment decreased meal-related dyspnoea (4.1 vs 1.7, p=0.019).

Conclusions

Impaired respiratory mechanics in hyperinflated emphysematous patients did not explain hypermetabolism.

INTRODUCTION

Only very recently, a new chronic obstructive pulmonary disease (COPD) phenotype titled 'multi-organ loss of tissue' has been proposed. This phenotype includes those with accelerated emphysema progression and enhanced tissue loss in other extra-pulmonary compartments, including muscle and adipose tissue. Disturbed tissue maintenance is associated with worse clinical outcomes [1], and might be the result of changes in whole body energy expenditure.

Whole body energy expenditure can be distinguished in basal metabolic rate (BMR), dietinduced thermogenesis and physical activity induced energy expenditure. Basal metabolic rate is primarily determined by fat-free mass (FFM) and comprises the largest part of total daily energy expenditure (TDEE) [2]. Diet induced thermogenesis is +/- 10% of TDEE [3] and physical activity induced energy expenditure largely depends on physical activity level [4]. Whole body energy expenditure can only be measured over a prolonged period in daily life using doubly labelled water [5]. This stable isotope methodology is very expensive and requires analytical technology that is available in a limited number of centres worldwide.

In COPD, an increased BMR relative to predicted values has repeatedly been demonstrated [6], which is more aggravated in weight losing patients [6] and in those with emphysema [7]. Although hypermetabolic at rest, COPD patients do not exhibit increased diet-induced thermogenesis [8]. Besides the proposed triggers for hypermetabolism including activation of brown adipose tissue, inflammation and increased whole body protein turnover, impaired lung mechanics might also result in hypermetabolism [9]. Emphysema is hallmarked by a reduction in lung elastic recoil and progressive hyperinflation, resulting in elevated airway resistance and contributing to impaired lung mechanics [10]. This results in an increased workload of breathing (ml oxygen cost per litre ventilation) [11]. The increased breathing workload has shown to be more pronounced in patients with low body weight and correlated with the degree of hyperinflation [12].

Pharmacological interventions may alleviate dyspnoea, reduce exercise limitation and improve quality of life in COPD by decreasing airway resistance and reduce hyperinflation. However, response is limited in patients with predominant emphysema [13]. In selected severe emphysematous patients bronchoscopic lung volume reduction (BLVR) is an additional treatment option that results in marked benefits in terms of pulmonary function, dyspnoea, exercise capacity and also physical activity [14, 15]. Furthermore, in a recent post-hoc analysis of the STELVIO trial [15] we illustrated a significant increase in body weight, skeletal muscle and fat tissue, suggesting a positive effect on energy balance regulation [16].

BLVR is an unique model to test the influence of lung mechanics on energy balance regulation, as it diminishes thoracic hyperinflation, reduces breathing frequency, and reduces mechanical

constraints on lung volume expansion, thereby improving ventilatory mechanics [17]. Efficacy of this treatment highly depends on advanced patient selection to identify responders to the treatment, and thereby creating a homogeneous study population.

We hypothesize that a decline in breathing workload following BLVR would decrease energy expenditure, which might positively influence components and determinants of energy balance. Secondly, BLVR may also improve dietary intake by alleviating eating associated dyspnoea and meal-related oxygen desaturation [18].

METHODS

Participants

Twenty patients with advanced emphysema, an identified target lobe with confirmed absence of collateral ventilation by the Chartis measurement who underwent BLVR treatment using one-way endobronchial valves were included in this study.

Patients were recruited from Maastricht University Medical Centre (MUMC+) and University Medical Centre Groningen (UMCG) in the Netherlands from September 2016 until April 2017. The Ethics Committee of Maastricht University Medical Centre approved the study protocol, and all participants provided written informed consent. Procedures were conducted according to the principles of the Declaration of Helsinki. The trial was registered at ClinicalTrial.gov (NCT02500004).

Study design

The study design is shown in figure 1.



Figure 1. Study design in days. Black arrows indicate urine sampling.



Prior to BLVR treatment, patients underwent a 2-week assessment period.

At day 1, patients were visited at home and received a dose of doubly labelled water. They were also instructed to collect urine samples for assessment of TDEE and to wear an accelerometer for registration of physical activity. Furthermore, patients were asked to record their dietary intake in order to assess if they were in a state of stable energy balance. On day 15, fasted state urine and blood samples were collected, weight, height and BMR were assessed and a meal test was performed (vide infra). This 2-week assessment period was repeated 6 months after BLVR treatment.

Body composition

Body height was determined to the nearest 0.5 cm while subjects were standing barefoot. Weight was assessed with a beam scale to the nearest 0.1 kg while subjects were standing barefoot and in light clothing. FFM was calculated from total body water assessment using the deuterium dilution technique assuming a hydration fraction of FFM of 73%.

Resting metabolic rate

BMR was measured by indirect calorimetry using a ventilated hood (EZCAL; Maastricht Instruments, Maastricht, the Netherlands and COSMED QUARK; TulipMed B.V., the Netherlands). Patients received their maintenance inhalation according to their normal habits. The time interval between medication use and start of indirect calorimetry was documented. During the second 2-week assessment period 6 months following BLVR treatment, the same time interval between medication use and start of indirect calorimetry was employed. Patients were in a fasting state for at least 10 hours and had a period of 30 minutes bed rest prior to the measurement during which subjects were lying on bed in supine position. After stabilization, BMR was recorded during a period of 30 minutes. BMR was calculated from oxygen consumption (VO_2) and carbon dioxide (VCO_2) production using the abbreviated Weir formula [19]. BMR was also predicted using the equation from Slinde et al., which was especially designed for COPD patients [20].

Total daily energy expenditure

TDEE was determined by the doubly labelled water technique over two 2-week periods (before and after BLVR treatment) according to the Maastricht protocol [21]. In the evening, prior to dosing, a urine sample was collected for determination of background isotope enrichment. Each patient received a weighted oral dose of water labelled with deuterium and oxygen-18. The given dose was calculated based on the subjects' total body water, which was estimated based on body mass index, age and gender. Subjects received a dose of 2.5 g/L total body water containing 250 ppm deuterium and 2200 ppm oxygen-18. After overnight equilibration, a second urine sample was collected from the second morning voiding. Additional urine samples were collected in the evening of days 1, 7 and 14 and in the morning of days 8 and 15. TDEE was calculated by the linear regression from the difference between elimination constants of deuterium and oxygen-18.

Physical activity

Actigraph GTX3 accelerometers (Actigraph, Pensacola, FL, USA) were used to assess the physical activity level. This activity monitor has been validated against activity related energy expenditure measured by doubly labelled water in patients with different stages of COPD [22]. The tri-axial accelerometers were attached to the lower back with an elastic belt and worn for 7 consecutive days. Subjects were instructed to wear the accelerometer during the time they were not asleep, except when showering or bathing. Only days with \geq 8 hours of wear time were accepted as valid days.

Energy expenditure for activities was calculated by (0.9*TDEE)-BMR, assuming a diet-induced thermogenesis of 10% of TDEE.

Dietary intake

Food intake was recorded by a food diary for 2 week days and 1 weekend day to estimate baseline energy balance.

Meal test

On the measurement day at the hospital, subjects received a standardized breakfast with wheat bread, butter, eggs, and milk. This meal contained a total of 502 kcal derived from protein (24%), carbohydrate (28%) and fat (48%). Oxygen saturation, breathing rate and heart rate were monitored before, during and after the breakfast via pulse oximetry. Before and immediately after the meal, dyspnoea was rated using the Borg dyspnoea scale.

Systemic Inflammatory status

High sensitive C-reactive protein (hsCRP) was assessed from frozen stored plasma collected from a venepuncture after overnight fasting.

Statistics

Descriptive statistics of demographic and clinical variables were obtained. Means (±SD) were provided for continuous normally distributed variables, medians (interquartile range) for continuous not-normally distributed variables and percentages were shown for categorical variables. Baseline and 6 months follow up measurements were compared with a paired-samples t test or Wilcoxon signed-rank test. All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, Version 24.0, IBM, Armonk, NY). Results with two-sided p values (<0.05) were considered statistically significant.

RESULTS

Patient characteristics

Twenty patients (7 men, 13 women) with severe emphysema (N=10 MUMC+ and N=10 UMCG) were enrolled in this study and 16 patients completed the follow-up. Reasons for drop out were patients' decision due to deterioration in health (N=2), patients' decision due to lack of efficacy of BLVR treatment (N=1) and diagnosis of bladder cancer (N=1). In four of the 16 patients who completed follow-up, endobronchial valves were removed due to granulation tissue around endobronchial valves (N=2), torsio bronchus (N=1) and recurrent pneumothorax (N=1) (figure 2).





Baseline characteristics are depicted in table 1. The study population represented a COPD population with normal BMI (24.1±4.4 kg/m²) and low FFM (FFMI males 15.1 kg/m² [14.7-16.2], females 13.5 kg/m² [12.1-18.1]). The prevalence of depletion of FFM, defined as FFMI≤17 kg/m² for males or ≤15 kg/m² for females [23], was 90%.
	, ,
General	
Male/Female, N	7/13
Age, years	63±7
Pack years	42±25
Lung function	
FEV1 (% predicted)	23.3±6.6
FVC (% predicted)	76.2±15.9
FEV1/FVC	28.5±6.3
RV (% predicted)	238.3±38.2
TLC (% predicted)	136.6±17.9
RV/TLC	64.7±9.4

Table 1	Baseline	characteristics ((N=20)	١
TUDIC I	Daschine	characteristics	11-20/	. ا

Body composition	
Weight (kg)	70.3±16.3
BMI (kg/m²)	24.1±4.4
FFM, kg	38.2 (32.1-57.4)
FFMI, kg/m ²	
Male	15.1 (14.7-16.2)
Female	13.5 (12.1-18.1)

Data are represented as mean \pm SD or median (minimum-maximum).

Definition of abbreviations: FEV1, forced expiratory volume in 1 sec; FVC, forced vital capacity; RV, residual volume; TLC, total lung capacity; BMI, body mass index; FFM, fat-free mass; FFMI, fat-free mass index.

Baseline assessment

At baseline, the mean BMR was 1537±259 kcal/day, which corresponded to 130% of predicted, indicating pronounced hypermetabolism. The average TDEE over two weeks was 2133±294kcal/ day. The average daily TDEE of week 1 was not statistically significantly different from the average daily TDEE of week 2. Energy expenditure for activities was median (range) 275 kcal/ day (138-827) (11% of TDEE).

Among those who completed follow-up, from all subjects but two (due to an accelerometer device defect), 6.5±1.1 valid accelerometry days were available with a mean of 13±1 hours of wear time per day. Median (range) steps per day was 2188 (739-7110). Patients spent a significant part of the day in sedentary state (79.7 % of the wear time [56.5-89.6]) (table 2).

Table 2. Clinical variables and components of energy balance at baseline and 6 months after BLVRtreatment (N=10).

	Baseline	After BLVR	p value
Lung function and symptom burden			
FEV1, % of predicted value	27.5±6.9	34.9±8.3	0.003
FVC, % of predicted value	74.4±15.0	95.1±17.1	<0.001
FEV1/FVC	30.5±7.2	29.5±5.6	0.591
RV, % of predicted value	236.2±37.6	181.3±27.5	<0.001
TLC, % of predicted value	135.0±20.0	125.1±14.7	0.007
RV/TLC	65.8±6.2	51.3±6.1	<0.001
COPD Assessment Test, points	18.6±3.4	14.3±5.8	0.022
6MWD, meter	378±98	427±84	0.030

	Baseline	After BLVR	p value
FFMI, kg/m ²			
Male	15.5 (14.7-16.2)	16.6 (15.5-18.6)	0.068
Female	13.2 (12.4-17.6)	13.6 (12.1-17.3)	0.600
Body composition			
Weight, kg	71.4±17.4	73.0±18.7	0.096
BMI, kg/m ²	24.3±4.3	24.8±5.0	0.127
FFM, kg	40.4 (32.2-57.4)	41.1 (29.4-60.2)	0.074
Energy expenditure			
VCO ₂ (ml/min)	179.5±26.1	180.5±28.6	0.854
VO ₂ (ml/min)	224.7±40.0	224.0±3.1	0.903
RQ	0.81±0.07	0.81±0.05	0.916
BMR measured (kcal/day)	1537±259	1549±231	0.778
BMR predicted (kcal/day)	1213±155	1245±189	0.103
BMR measured /BMR predicted ratio	1.3±0.2	1.2±0.1	0.655
TDEE (kcal/24h)	2133±294	2192±480	0.576
TDEE/BMR ratio	1.4±0.2	1.4±0.3	0.934
Energy expenditure for activities (kcal/day)	275 (138-827)	397 (18-1262)	0.694
Energy expenditure for activities/TDEE ratio	0.2±0.1	0.2±0.1	0.995
Physical activity level			
Mean steps/day	2188 (739-7110)	2429 (990-6983)	0.161
Time spent in sedentary PA, % of wear time	79.7 (56.5-89.6)	79.4 (52.7-84.2)	0.123
Time spent in lifestyle PA, % of wear time	17.6 (10.1-34.2)	18.6 (14.1-37.9)	0.123
Time spent in MVPA, % of wear time	0.0 (0.0-1.1)	0.2 (0.0-1.4)	0.028
Inflammation			
hsCRP, mg/L	3.0±2.7	2.5±1.8	0.463

Table 2 Continues.

Data are represented as mean ± SD or median (minimum-maximum).

Definition of abbreviations: FEV1, forced expiratory volume in 1 second; FVC, forced vital capacity; RV, residual volume; TLC, total lung capacity; BMI, body mass index; FFM, fat-free mass; FFMI, fat-free mass index; 6MWD, 6-minute walk distance; MVPA, moderate-to-vigorous physical activity; PA, physical activity; BMR, basal metabolic rate; TDEE, total daily energy expenditure; hsCRP, high sensitive C-reactive protein. COPD Assessment Test, missing N=2

Systemic inflammation measured by hsCRP was 3.0±2.7 mg/L. BMR or TDEE was not associated with hsCRP or residual volume (RV) (% of predicted) [data not shown].

Reported dietary intake comprised 2065 \pm 507 kcal/24h, which equalled measured TDEE. Patients experienced more dyspnoea after eating (4.1 \pm 1.8 after meal vs 2.1 \pm 2.1 before meal, p=0.013). No significant change was shown in oxygen saturation, respiration rate and heart beat rate during the course of the meal (figure 3).

В



		Baseline	After BLVR	p value
Borg dyspnoea	Before meal	2.1 ± 2.1	1.3 ± 2.1	0.359
score	After meal	4.1 ± 1.8	1.7 ± 2.4	0.019
Oregon caturation	Before meal	93.2 ± 2.4	92.2 ± 3.1	0.193
(%)	During meal	92.5 ± 3.6	91.9 ± 3.3	0.272
	After meal	92.0 ± 3.1	92.4 ± 3.7	0.406
Pageiration rate	Before meal	16.1 ± 3.7	16.5 ± 2.6	0.647
(respirations/min)	During meal	16.7 ± 3.7	16.6 ± 3.4	0.850
	After meal	16.0 ± 4.5	17.1 ± 3.2	0.465
Heart rate	Before meal	74.6 ± 8.7	70.1 ± 10.2	0.155
(beats/min)	During meal	82.2 ± 8.4	77.4 ± 10.4	0.167
	After meal	83.1 ± 9.8	77.3 ± 11.3	0.134

Figure 3. (a) Borg dyspnoea score, (b) Borg dyspnoea score, oxygen saturation, respiration rate and heart rate before, during and after completion of a standardized meal, before and after BLVR treatment (N=10). Data are presented as mean values (± standard deviation).

Response after BLVR

Not all patients benefited from the BLVR treatment, in terms of hyperinflation reduction. We therefore took a closer look to the 10 patients who responded beyond the MCID for RV reduction of >430 ml [24]. At 6 months follow-up, patients significantly improved in RV and forced expiratory volume in 1 second, with respectively 1285 ml (-2430, -540) and 190 ml (10-390).

BMR did not significantly change over time (1537±259 kcal/day vs 1549±231 kcal/day, p=0.778) and patients remained hypermetabolic (BMR was 130% of predicted). No changes in TDEE were observed (2133±294 kcal/day vs 2192±480 kcal/day, p=0.576), in accordance to an unaltered physical activity expressed by mean number of daily steps. Although 6MWD increased significantly, the mean step count and activity-induced energy expenditure, did not change over time. hsCRP also remained unchanged (table 2).

A significant effect of BLVR treatment on meal-related dyspnoea was observed. Compared to baseline meal-related dyspnoea after the meal was significantly lower after BLVR treatment (1.7±2.4 vs 4.1±1.8, p=0.019). No changes were found in oxygen saturation, respiration rate or heart rate during the meal (figure 3).

DISCUSSION

This is the first study presenting a comprehensive analysis of energy balance in a homogeneous group of patients with severe emphysema and investigating the effect of BLVR. In contrast to our hypothesis, a median reduction of hyperinflation with 25% did not decrease BMR or TDEE adjusted for physical activity level. Eating-related dyspnoea however was diminished.

In line with the 'multi-organ loss of tissue' phenotype [1], we observed a very high prevalence of FFM depletion indicative for disturbed muscle maintenance. Nearly all patients were FFM depleted, but this was disproportionate to the FM as the majority of patients fell within a normal BMI range. Before BLVR, BMR was very high, up to 130% of predicted and energy expenditure for physical activities was very low (11%). This implies that in this patient group and at this stage of the disease, fat mass regulation is primarily determined by the balance between energy intake and whole body energy requirements and less or not yet by fat catabolism (i.e. increased lipolysis or brown adipose tissue activation). The normal BMI in this population hides FFM depletion, emphasizing the importance of body composition assessment for estimation of metabolic risk as proposed by the European Respiratory Society task force Nutritional Assessment and Therapy in COPD [25].

No studies to date have investigated the effect of lung volume reduction on TDEE, but a few studies previously reported the effect of lung volume reduction surgery on BMR. Mineo et al. showed a reduction of BMR with 5% [26], while Takayama et al. observed no change in BMR [27]. The degree of hyperinflation reduction was comparable to our cohort. Nevertheless, one needs to consider that, although our patients improved importantly after intervention, they still remain severely hyperinflated with a mean residual volume of 181% of predicted.

A contributor to BMR is whole body protein turnover, which explained approximately 20% of the between subject variation of BMR, in healthy young individuals [28]. Also in COPD, increased rates of whole body protein turnover have been reported [29, 30], which associated with BMR [31]. Increased muscle turnover signalling was accompanied with elevated myogenic signalling [32], which was most prominent in patients with FFM depletion. Therefore, persistence of high BMR after BLVR might be the result of energy cost of protein anabolism, supported by increased muscle mass observed previously in chest CT scans [16].

In the absence of catabolic drivers, fat mass is primarily regulated by the balance between energy intake and energy metabolism. In line with others [33, 34], our patients experienced an eating induced increase in dyspnoea. Vermeeren et al. reported the effects of different meals on dyspnoea sensation and found a significantly greater increase in dyspnoea after ingestion of a fat-rich meal than after a carbohydrate-rich meal [33]. Here, we show for the first time that dyspnoea after the same, standardized meal was significantly less following BLVR. In line with two other studies, these effects could not be explained by changes in meal related oxygen saturation [18,33].

Systemic inflammation has been proposed as putative trigger for hypermetabolism in particular during acute exacerbations [35, 36]. Indeed elevated CRP levels have previously been associated with higher BMR in clinically stable COPD [37, 38]. In this study, CRP levels were slightly elevated but did not change after BLVR.

The strength of this prospective well-controlled clinical proof of concept study comes from the well-defined patient cohort and from the use of gold standard methods to assess body composition, BMR and TDEE. We recognize that the study power was based on detection of changes in energy metabolism in relation to changes in lung function but not for changes in body composition. The technique of pulse oximetry has the advantage of providing a continuous and non-invasive measurement of oxygen saturation. However, this technique is limited by a poorer accuracy of 1-3% when compared to arterial blood sampling [39].

To conclude, the present work showed that impaired respiratory mechanics in hyperinflated emphysematous patients did not explain hypermetabolism.

REFERENCES

- Celli BR, Locantore N, Tal-Singer R, et al. Emphysema and extrapulmonary tissue loss in COPD: a multiorgan loss of tissue phenotype. Eur Respir J 2018; 51.
- Wang Z, Heshka S, Gallagher D, et al. Resting energy expenditure-fat-free mass relationship: new insights provided by body composition modeling. Am J Physiol Endocrinol Metab 2000; 279: E539-45.
- Westerterp KR. Diet induced thermogenesis. Nutr Metab (Lond) 2004; 1: 5.
- Pannemans DL, Westerterp KR. Energy expenditure, physical activity and basal metabolic rate of elderly subjects. Br J Nutr 1995; 73: 571-81.
- Schoeller DA, Ravussin E, Schutz Y, et al. Energy expenditure by doubly labeled water: validation in humans and proposed calculation. Am J Physiol 1986; 250: R823-30.
- Schols AM, Soeters PB, Mostert R, et al. Energy balance in chronic obstructive pulmonary disease. Am Rev Respir Dis 1991; 143: 1248-52.
- Cohen RI, Marzouk K, Berkoski P, et al. Body composition and resting energy expenditure in clinically stable, non-weight-losing patients with severe emphysema. Chest 2003; 124: 1365-72.
- Hugli O, Frascarolo P, Schutz Y, et al. Diet-induced thermogenesis in chronic obstructive pulmonary disease. Am Rev Respir Dis 1993; 148: 1479-83.
- Sanders KJ, Kneppers AE, Van De Bool C, et al. Cachexia in chronic obstructive pulmonary disease: new insights and therapeutic perspective. J Cachexia Sarcopenia Muscle 2016; 7: 5-22.
- Criner GJ, Cordova F, Sternberg AL, et al. The National Emphysema Treatment Trial (NETT): Part I: Lessons learned about emphysema. Am J Respir Crit Care Med 2011; 184: 763-70.
- Cherniack RM. The oxygen consumption and efficiency of the respiratory muscles in health and emphysema. J Clin Invest 1959; 38: 494-9.
- Donahoe M, Rogers RM, Wilson DO, et al. Oxygen consumption of the respiratory muscles in normal and in malnourished patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1989; 140: 385-91.
- O'donnell DE, Fluge T, Gerken F, et al. Effects of tiotropium on lung hyperinflation, dyspnoea and exercise tolerance in COPD. Eur Respir J 2004; 23: 832-40.
- 14. Hartman JE, Klooster K, Slebos DJ, et al.

Improvement of physical activity after endobronchial valve treatment in emphysema patients. Respir Med 2016; 117: 116-21.

- Klooster K, Ten Hacken NH, Hartman JE, et al. Endobronchial Valves for Emphysema without Interlobar Collateral Ventilation. N Engl J Med 2015; 373: 2325-35.
- Sanders KJ, Klooster K, Vanfleteren L, et al. CTderived muscle remodelling after bronchoscopic lung volume reduction in advanced emphysema. Thorax 2018.
- O'donnell DE, Webb KA, Bertley JC, et al. Mechanisms of relief of exertional breathlessness following unilateral bullectomy and lung volume reduction surgery in emphysema. Chest 1996; 110: 18-27.
- Schols AM, Mostert R, Cobben N, et al. Transcutaneous oxygen saturation and carbon dioxide tension during meals in patients with chronic obstructive pulmonary disease. Chest 1991; 100: 1287-92.
- Weir JB. New methods for calculating metabolic rate with special reference to protein metabolism. 1949. Nutrition 1990; 6: 213-21.
- Slinde F, Ellegard L, Gronberg AM, et al. Total energy expenditure in underweight patients with severe chronic obstructive pulmonary disease living at home. Clin Nutr 2003; 22: 159-65.
- Westerterp KR, Wouters L, Van Marken Lichtenbelt WD. The Maastricht protocol for the measurement of body composition and energy expenditure with labeled water. Obes Res 1995; 3 Suppl 1: 49-57.
- Rabinovich RA, Louvaris Z, Raste Y, et al. Validity of physical activity monitors during daily life in patients with COPD. Eur Respir J 2013; 42: 1205-15.
- 23. Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. Am J Respir Crit Care Med 2006; 173: 79-83.
- Hartman JE, Ten Hacken NH, Klooster K, et al. The minimal important difference for residual volume in patients with severe emphysema. Eur Respir J 2012; 40: 1137-41.
- Schols AM, Ferreira IM, Franssen FM, et al. Nutritional assessment and therapy in COPD: a European Respiratory Society statement. Eur Respir J 2014; 44: 1504-20.
- 26. Mineo TC, Pompeo E, Mineo D, et al. Resting

energy expenditure and metabolic changes after lung volume reduction surgery for emphysema. Ann Thorac Surg 2006; 82: 1205-11.

- Takayama T, Shindoh C, Kurokawa Y, et al. Effects of lung volume reduction surgery for emphysema on oxygen cost of breathing. Chest 2003; 123: 1847-52.
- Welle S, Nair KS. Relationship of resting metabolic rate to body composition and protein turnover. Am J Physiol 1990; 258: E990-8.
- Engelen MP, Deutz NE, Wouters EF, et al. Enhanced levels of whole-body protein turnover in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med 2000; 162: 1488-92.
- Morrison WL, Gibson JN, Scrimgeour C, et al. Muscle wasting in emphysema. Clin Sci (Lond) 1988; 75: 415-20.
- Kao CC, Hsu JW, Bandi V, et al. Resting energy expenditure and protein turnover are increased in patients with severe chronic obstructive pulmonary disease. Metabolism 2011; 60: 1449-55.
- Kneppers AEM, Langen RCJ, Gosker HR, et al. Increased Myogenic and Protein Turnover Signaling in Skeletal Muscle of Chronic Obstructive Pulmonary Disease Patients With Sarcopenia. J Am Med Dir Assoc 2017; 18: 637 e1-37 e11.
- Vermeeren MA, Wouters EF, Nelissen LH, et al. Acute effects of different nutritional supplements on symptoms and functional capacity in patients with chronic obstructive pulmonary disease. Am J Clin Nutr 2001; 73: 295-301.
- Wolkove N, Fu LY, Purohit A, et al. Meal-induced oxygen desaturation and dyspnea in chronic obstructive pulmonary disease. Can Respir J 1998; 5: 361-5.
- Hopkinson NS, Tennant RC, Dayer MJ, et al. A prospective study of decline in fat free mass and skeletal muscle strength in chronic obstructive pulmonary disease. Respir Res 2007; 8: 25.
- Rutten EP, Spruit MA, Mcdonald ML, et al. Continuous fat-free mass decline in COPD: fact or fiction? Eur Respir J 2015; 46: 1496-8.
- Broekhuizen R, Wouters EF, Creutzberg EC, et al. Raised CRP levels mark metabolic and functional impairment in advanced COPD. Thorax 2006; 61: 17-22.
- Schols AM, Buurman WA, Staal Van Den Brekel AJ, et al. Evidence for a relation between metabolic derangements and increased levels of

inflammatory mediators in a subgroup of patients with chronic obstructive pulmonary disease. Thorax 1996; 51: 819-24.

 Nickerson BG, Sarkisian C, Tremper K. Bias and precision of pulse oximeters and arterial oximeters. Chest 1988; 93: 515-7.

Energy balance following lung volume reduction



7

Brown adipose tissue activation in relation to hypermetabolism in emphysematous chronic obstructive pulmonary disease patients

Karin J.C. Sanders, Roel Wierts, Wouter D. van Marken Lichtenbelt, Judith de Vos-Geelen, Guy Plasqui, Marco C.J.M. Kelders, Vera B. Schrauwen-Hinderling, Jan Bucerius, Anne-Marie C. Dingemans, Felix M. Mottaghy, Annemie M.W.J. Schols.

Submitted.

ABSTRACT

Introduction

Brown adipose tissue (BAT) has been primarily researched as a potential target for mitigating obesity. However, the physiological significance of BAT in relation to cachexia remains unknown. The objective of this study was to investigate the putative contribution of BAT on different components of energy metabolism in patients with emphysema.

Methods

Twenty emphysematous patients (mean ± SD age 62 ± 6, median [range] BMI 22.4 [15.1-32.5] kg/m², 85% low fat-free mass index [FFMI]) were studied. Basal metabolic rate (BMR) was assessed by ventilated hood, total daily energy expenditure (TDEE) by doubly labelled water and physical activity by tri-axial accelerometry. BMR was adjusted for fat-free mass (FFM) as assessed by deuterium dilution. Analysis of BAT and white adipose tissue was conducted in a subset of ten patients and six age, gender and BMI matched healthy controls. BAT glucose uptake was assessed by means of cold-stimulated integrated [18F]FDG positron-emission tomography and magnetic resonance imaging. White adipose tissue was collected from subcutaneous abdominal biopsies to analyse metabolic and inflammatory gene expression levels. Lung function was assessed by spirometry and bodyplethysmography and systemic inflammation by high sensitivity C-reactive protein.

Results

Mean TDEE was 2209 \pm 394 kcal/day and mean BMR was 1449 \pm 214 kcal/day corresponding to 120% of predicted. Fat-free mass adjusted BMR did not correlate with lung function or systemic inflammation.

Upon cooling, energy expenditure increased, resulting in a non-shivering thermogenesis of (median [range] 20.1% [3.3-41.3] in patients and controls. Mean BAT glucose uptake was comparable between COPD and controls (1.5 [0.1-6.2] versus 1.1 [0.7-3.9]). In addition, no correlation was found between BMR adjusted for fat-free mass and BAT activity and between cold-induced non-shivering energy expenditure and BAT activity. Gene expression levels of the brown adipocyte or beige markers were also comparable between the groups.

Conclusions

Although COPD patients were hypermetabolic at rest, no correlation was found between energy expenditure and BAT activity. Furthermore, both BAT activity and gene expression levels of the brown adipocyte or beige markers were comparable between COPD and controls.

INTRODUCTION

Chronic obstructive pulmonary disease (COPD) is an airway and lung disease with persistent airflow obstruction, which is the result of small airway remodelling and loss of elastic recoil [1]. Disturbances in whole body and cellular energy metabolism are common in COPD and contribute to weight loss and muscle wasting in particular in the emphysematous phenotype [2-4].

Total daily energy expenditure (TDEE) consists of basal metabolic rate (BMR), diet induced thermogenesis and activity induced energy expenditure (AEE). AEE has been shown to be elevated due to decreased mechanical efficiency [5, 6] but may also be (relatively) decreased in patients with very low physical activity levels [3]. Diet induced thermogenesis was shown to be normal in previous research [7, 8]. Increased BMR relative to predictive values has been consistently reported in COPD [2] and found more aggravated among weight losing patients [2, 3] and among those with emphysema [9, 10]. Hypermetabolic patients do not necessarily exhibit an elevated TDEE [11]. Systemic inflammation has been proposed as cause of elevated BMR in cachexia induced by cancer [12] or COPD [13]. Another contributing factor to hypermetabolism in COPD might be the workload of breathing [14] due to lung hyperinflation. Nevertheless, we recently showed that decreasing hyperinflation with 25% by endoscopic lung volume reduction in patients with emphysema did not decrease BMR (own results, accepted for publication).

Activation of brown adipose tissue (BAT) or browning of white adipose tissue (WAT) has been proposed as putative trigger for hypermetabolism in cachexia but to date no evidence from human research is available [15, 16]. BAT tissue is characterized by abundance of mitochondria and by heat production via the process of proton leakage over the mitochondrial membrane mediated by uncoupling protein 1 [17]. Compared to BAT-negative healthy young lean subjects, those with BAT had a higher cold-induced energy expenditure, with an average difference of 370 kcal/day [18]. In healthy young lean men, cold induction increased energy expenditure by on average 17% [19]. This increase in energy expenditure upon cold in the absence of shivering (i.e. non-shivering thermogenesis) implies that BAT thermogenesis may be a significant component of BMR and thereby plays a role in regulation of body weight as suggested by an inverse association between cold-induced BAT activity and body mass index in adults [20-22].

BAT has been primarily researched as a potential target for mitigating obesity [23]. Yet few research attempts have been made to explore the role of BAT in relation to cachexia as observed in patients with COPD and no clinical studies are available. There is some indirect evidence pointing towards a possible role for BAT as driver of hypermetabolism in the emphysematous COPD subtype. Patients with advanced emphysema have high plasma levels of norepinephrine at rest [24], which is a potential browning factor [25]. Furthermore, whole body glucose

production [26] and skeletal muscle glycolysis [27, 28] is increased in COPD patients, which result in increased plasma lactate levels, even at rest [29]. Lactate in turn is able to control expression of uncoupling protein 1 (UCP1), which also promotes browning. We therefore hypothesized that BAT or browning of WAT contributes to energy expenditure in emphysematous COPD patients. Since limited studies are available regarding TDEE and the different components in mixed COPD phenotypes, we first measured energy expenditure in emphysematous COPD patients and related its components to airflow obstruction and hyperinflation. We then conducted a controlled clinical experiment in a subgroup to assess presence of BAT by cold induced BAT activity and BAT and beige gene expression markers in WAT.

METHODS

Participants

Whole body energy metabolism and its components were assessed in twenty Caucasian COPD patients with emphysema. The contribution of BAT and WAT was studied in a subgroup of ten patients and a control group, matched for ageing, male gender and adiposity, since these factors are known to negatively correlate with BAT activity [22]. Furthermore, BAT volumes may differ between ethnicities [30]. In addition, since cigarette smoke is acknowledged as the most important risk factor for COPD and control subjects may not be diagnosed with COPD yet, non-smoking controls were included.

Patients were recruited from Maastricht University Medical Centre (MUMC+) and the healthy controls were recruited via advertisements in local newspapers. The Ethics Committee of MUMC+ approved the study protocol, and all participants provided written informed consent. Procedures were conducted according to the principles of the Declaration of Helsinki. The study was registered at ClinicalTrial.gov (NCT02500004).

Exclusion criteria for the COPD as well as the healthy controls were diabetes mellitus, severe clotting disorder, active malignancy, claustrophobia, long-term oxygen therapy, history of radiotherapeutic radiation in the neck and/or upper chest, cervical or thoracic sympathectomy and the presence of MRI contra-indications, such as pacemaker, cochlear implant, vascular clips etc. In addition, subjects using medication that influences the sympathetic nerve system, including β -blockers, α -blockers, central antihypertensive drugs, certain anti-depression drugs (MAO inhibitors, tricyclic anti-depressives), reserpine, cocaine, calcium channel blockers, and certain tranquillizers (fenothiazines) were also not eligible. Furthermore, patients with pancreatic cancer were included, with similar exclusion criteria as mentioned above.

The original intention was to compare the COPD patients not only to healthy controls, but also to matched lung cancer and pancreatic cancer patients. However, the BAT protocol was too strenuous, regarding eligibility criteria as well as methodology used, for most of the cachectic cancer patients. We were able to measure only two pancreatic cancer patients and no lung cancer patients before study closure.

Study design

The overall study design is shown in figure 1. All participants completed the 2-week study period focussing on whole body energy metabolism. A subset underwent the protocol focussing on the contribution of BAT and subcutaneous adipose tissue (SAT) to energy metabolism. This protocol included an individualized cooling protocol [31], [18F]FDG positron-emission tomography (PET) and magnetic resonance imaging (MRI) scanning in order to measure BAT activity and a subcutaneous fat biopsy obtained after an overnight fast.



Figure 1. Schematic illustration of the study protocol.

Basal metabolic rate

BMR was measured by indirect calorimetry using a ventilated hood (EZCAL; Maastricht Instruments, Maastricht, the Netherlands). Patients were in a fasting state for at least ten hours and were requested to abstain from heavy exercise, inhalation medication, and smoking for 24 hours. Prior to the measurement the subjects had a period of 30 minutes bed rest during which they were lying on bed in supine position. After stabilization, BMR was recorded during a period of 30 minutes. BMR was calculated from oxygen consumption (VO₂) and carbon dioxide (VCO₂) production using the abbreviated Weir formula [32]. BMR was also predicted using the equation from Slinde et al. [4].

Doubly labelled water

TDEE was determined by the doubly labelled water technique over 2 weeks according to the Maastricht protocol [33]. In the evening, prior to dosing, an urine sample was collected for determination of background isotope enrichment. Each patient received a weighted oral dose of water labelled with ~ 7% deuterium and 10% oxygen-18. The given dose was calculated based on the subjects' total body water, which was estimated based on body mass index, age and gender. Subjects received a dose of 2.4 g per litre total body water resulting in an initial

excess enrichment of approximately 150 ppm for deuterium and 200 ppm for oxygen-18. The basic principle of DLW is that after equilibration with total body water, deuterium and oxygen-18 are eliminated from the body at a different rate. Deuterium is eliminated from the body as water whereas oxygen-18 is eliminated as water and carbon dioxide. The difference in elimination rates is hence a measure of CO_2 -production. After ingestion of doubly labelled water, urine samples were collected at standardized time points (in the evening of days 1, 7, 14 and in the morning of days 8 and 15). CO_2 -production was calculated by the linear regression from the difference between elimination constants of deuterium and oxygen-18. TDEE was calculated from CO_2 -production assuming an RQ of 0.85, corresponding to the RQ of an average Western diet.

Energy expenditure for activities was calculated by (0.9*TDEE)-BMR, assuming a diet-induced thermogenesis of 10% of TDEE.

Body composition

Body height was determined to the nearest 0.5 cm while subjects were standing barefoot. Weight was assessed with a beam scale to the nearest 0.1 kg while subjects were standing barefoot and in light clothing. FFM was calculated from total body water assessment using the deuterium dilution technique assuming a hydration fraction of FFM of 73%.

Physical activity

Actigraph GTX3 accelerometers (Actigraph, Pensacola, FL, USA) were used to assess physical activity. This activity monitor has been validated against activity related energy expenditure measured by doubly labelled water in patients with different stages of COPD [34]. The triaxial accelerometers were attached to the lower back with an elastic belt and worn for seven consecutive days. Subjects were instructed to wear the accelerometer during the time they were not asleep, except when showering or bathing. Only days with ≥ 8 hours of wear time were accepted as valid days.

Dietary intake

Energy intake was recorded by a food diary for two week days and one weekend day to judge if subjects were in a stable energy balance.

Individualized cooling and PET/MRI scanning

For individualized cooling, subjects were wrapped in a water-perfused suit. During the cooling procedure, skin temperature, blood pressure, heart rate and oxygen saturation were monitored. Skin temperatures were measured continually at 14 positions by means of iButtons (type DS1291H, Dallas Maxim Semiconductor Corp., USA) [35] and on three additional positions (supraclavicular, underarm and fingertip). Blood pressure and heart rate were measured via an automated blood pressure monitor (Omron Healthcare Inc, Field Court Lake Forest, USA)

and oxygen saturation was measured via pulse oximetry (Nellcor Puritan Bennett NPB-40, Pleasanton, CA, USA).

The first 45 minutes, subjects were kept at thermoneutral conditions with water temperature of the water-perfused suit at 34°C. Following this thermoneutral condition, a gradual stepwise decrease in temperature followed (water temperature was lowered with 4°C every 15 minutes) until shivering occurred. After the onset of shivering, subjects were warmed up for 5 minutes after which the temperature was set slightly above shivering level for 30 minutes. At thermoneutral condition and during cold exposure, indirect calorimetry was performed to measure energy expenditure. Subsequently, 150 MBq of 18F-FDG was injected intravenously while mild cold stimulation remained for 45 minutes after injection. At 60 minutes a hybrid PET/ magnetic resonance imaging (MRI) scan (Siemens Healthcare GmbH, Erlangen, Germany, software version VB20P) was acquired.

PET/MRI analysis

PET/MRI scans were analysed using PMOD software (version 3.7, PMOD Technologies, Zurich, Switzerland). For assessment of BAT activity, volumes of interest (VOIs) were drawn which encompassed supraclavicular adipose tissue. A sphere with radius 6.2 mm was deposited on the location where the average value was maximal in the selected VOI. Additionally, VOIs were placed in the liver, several skeletal muscle groups (biceps and triceps brachii, deltoid, erector spinae), subcutaneous adipose tissue and visceral adipose tissue. For each VOI and sphere, the measured activity concentration was corrected for radioactive decay, total administered activity and body weight, resulting in the mean standardized uptake value, SUVmean.

Blood analysis

Venous blood was obtained after an overnight fast for analysis of C-reactive protein (CRP), glucose, insulin, thyroid stimulating hormone, free fatty acids, cholesterol, triglycerides, highand low-density lipoprotein.

Fat biopsy

A subcutaneous fat biopsy was obtained para-umbilically during fasting conditions through needle biopsy. This biopsy was snap-frozen in liquid nitrogen and stored at-80°C until analysing gene expression by qPCR.

Statistics

Descriptive statistics of demographic and clinical variables were obtained. Means (± SD) are provided for continuous normally distributed variables, median (range) for continuous notnormally distributed variables and percentages are shown for categorical variables. Between group differences were tested with an independent t-test, Mann-Whitney U test or Chi square test. All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, Version 24.0, IBM, Armonk, NY). Results with two-sided p values (<0.05) were considered statistically significant.

RESULTS

Baseline characteristics of participants

Ten male and ten female COPD patients with emphysema were enrolled. Baseline characteristics are depicted in table 1.

	COPD	Controls
General		
Male/Female, N	10/10	4/2
Age, years	62 ± 6	60 ± 6
Pack years	36 ± 17	4 ± 8
Smoking status, N Never Current Former	0 4 16	4 0 2
Body composition		
Weight, kg	67 ± 15	70.8 ± 3.8
BMI, kg/m²	22.4 (15.1-32.5)	23.1 (21.2-30.8)
FFM, kg	44.7 (31.7-57.4)	50.5 (44.6-63.8)
FFMI, kg/m² Male Female	15.7 (14.2-18.2) 13.5 (12.7-17.6)	16.9 (15.3-19.7) 18.0 (16.2-19.8)
Lung function		
FEV1, % predicted	36.0 ± 15.4	not applicable
FVC, % predicted	82.2 ± 18.3	not applicable
FEV1/FVC	39.2 ± 15.4	not applicable
RV, % predicted	209.7 ± 61.2	not applicable
TLC, % predicted	128.7 ± 23.7	not applicable
RV/TLC	1.6 ± 0.3	not applicable
DLCO, % predicted	38.0 ± 9.4	not applicable

Table 1. Baseline characteristics of COPD patients (N=20) and healthy controls (N=6).

Data are represented as mean ± SD or median [range].

Definition of abbreviations: BMI, body mass index; FFM, fat-free mass; FFMI, fat-free mass index; FEV1, forced expiratory volume in 1 sec; FVC, forced vital capacity; RV, residual volume; TLC, total lung capacity; DLCO, diffusion capacity of lung for carbon monoxide.

The COPD population was characterized by moderate to severe airflow obstruction and low diffusion capacity (FEV1: $36 \pm 15\%$ of predicted and DLCO: $38 \pm 9\%$ predicted). Median BMI (22.4 [15.1-32.5]) was within the normal range but median (range). FFM index was below normal (FFMI males 15.7 kg/m² [14.2-18.2], females 13.5 kg/m² [12.7-17.6]) and 85\% was depleted according to current definition (FFMI ≤17 kg/m² for males or ≤15 kg/m² for females [36]).

A subset of ten COPD patients (five males and five females) and six controls (BMI 23.1 [21.2-30.8] kg/m², FFMI males 16.9 kg/m² [15.33-19.7] and females 18.0 kg/m² [16.20-19.8]) underwent individualized cooling and PET/MRI scanning in order to measure BAT activity. PET/MRI imaging was impossible in one COPD patient, because she experienced a panic attack immediately before initiation of the scan. Compared to controls, COPD patients exhibited statistically higher insulin (27.2 [12.0-86.6] versus 13.0 [12.0-19.0] pmol/L), while fasting glucose was comparable (5.8 \pm 1.0 versus 4.9 \pm 0.5 mmol/L). Groups were comparable regarding thyroid function and lipid profile. CRP levels on average were low but tended to be higher in the COPD patients (2.5 [2.0-21.0] versus 0.0 [0.0-3.0] mg/L) (supplemental table 1).

Participans are hypermetabolic at rest

Mean TDEE was 2209 \pm 394 kcal/day in the COPD patients and not significantly different from dietary intake (2235 \pm 600 kcal/day). AEE amounted to 538 \pm 258 kcal/day (23% of TDEE) in COPD and 812 \pm 255 (32% of TDEE) in healhy controls, indicating a low physical activity level in the COPD group. Indeed, median (range) steps per day were 3097 [468-8792] and 8519 [6517-13518] for respectively COPD patients and healthy controls. Patients compared to controls spend a larger part of the day in sedentary state (70.2% of the wear time [56.5-90.6] vs 51.3% of wear time [41.7-66.8]) (figure 2). TDEE was highly correlated to activity-induced energy expenditure (r=0.80, P<0.001).





In COPD, the mean BMR of 1449 ± 214 kcal/day corresponded to 120% of predicted. 17 patients were characterized as hypermetabolic based on measured BMR >110% of predicted. No significant correlations could be found between adjusted BMR and FEV1 % predicted (r=-0.22, P=0.366) and between adjusted BMR and RV % predicted (r=0.04, P=0.888). Furthermore, adjusted BMR was not correlated to CRP (r=-0.18, p=0.621).

Hypermetabolism is not associated with BAT activation

Upon cooling in both COPD patients and controls, mean skin temperature decreased from 31.9 \pm 2.3 to 27.6 \pm 3.3 °C and supraclavicular skin temperature decreased from 35.3 \pm 0.9 to 30.4 \pm 5.2 °C. The cooling protocol increased energy expenditure (4.5 [3.1-5.0] versus 5.1 [4.1-7.0] kJ/min), resulting in an increment of average non-shivering thermogenesis of 20.1% (3.3-41.3). Blood pressure and heart rate increased upon cold exposure, while oxygen saturation remained unchanged. This increment in blood pressure and heart rate was not different between groups, indicating a comparable cold stress reaction.

Mean SUV of the region of BAT was comparable between COPD and controls (1.5 g/mL [0.1-6.2] versus 1.1 g/mL [0.7-3.9]). Furthermore, cold-induced [18F]FDG uptake in skeletal muscle, subcutaneous adipose tissue and visceral adipose tissue were neither different (figure 3). In addition, no correlations were found between FFM adjusted BMR and BAT activity (r=-0.06, p=0.881) and between non-shivering thermogenesis and SUV uptake in skeletal muscles (r=0.011, p=0.970). Additionally, no significant relation was found between non-shivering thermogenesis and BAT activity (r=-0.044, p=0.877).



Figure 3. (a) area of brown adipose tissue on MRI and PET. (b) mean uptake of [18F]FDG in brown adipose tissue and other tissues during cold exposure. SUV, standardized uptake value.

Gene expression levels of BAT or beige markers in white fat are not upregulated

In subcutaneous adipose tissue UCP1 mRNA, which is considered as an important BAT and beige marker [37], was undetectable in most subjects. Furthermore, we did not detect differences in mRNA expression of the brown adipocyte marker Cidea or beige markers TMEM26 and CD137 between COPD patients and controls (figure 4).

Hypoxia marker GLUT-1 mRNA expression level was higher in COPD compared to controls and HIF1 alpha tended to be more expressed in COPD than in control subjects. Gene expression levels of markers of inflammation, macrophages, glycolysis, lipolysis and adipokines were not differentially expressed (supplemental table 1).



Figure 4. Adipose tissue gene expression levels of BAT and beige markers. AU, Arbitrary Units.

Role of BAT in cachectic cancer patients: a case report

A 61-year old male (BMI 23.9 kg/m²) and a 70-year old female (BMI 23.8 kg/m²) diagnosed with pancreatic cancer were both enrolled in one arm of the current study that was closed due to recruitment difficulties (see methods section for explanation). The male underwent a pancreaticoduodenectomy, and because of a relapse he started with chemotherapy which comprised oxaliplatin, folinezuur, irinotecan and fluorouracil. The female was diagnosed with pancreatic cancer with pulmonary and lymph node metastasis, and she was treated with gemcitabine. Both were cachectic at time of enrollement with a weight loss of more than 5% of body weight in the previous 6 months and both were hypermetabolic at rest (mean BMR was 120% of predicted). Despite a cold induced increased RMR from 4.5 to 5.3 kJ/min, no BAT activity was found. In line with the results from COPD patients and controls, no upregulation of BAT and beige markers were found in subcutaneous fat.

DISCUSSION

Activation of BAT or browning of WAT has been proposed as putative trigger for hypermetabolism and cachexia in wasting disorders including COPD. This is the first study that investigated the contribution of BAT on hypermetabolism in patients with emphysema in comparison to appropriately matched healthy controls. In contrast to our hypothesis, emphysematous patients did not exhibit enhanced BAT activity or altered gene expression of BAT markers in WAT compared to age-, gender- and BMI matched healthy non-smoking controls.

Consistent with previous publications [2, 9], we found that BMR was elevated among those with COPD. Although, upon cooling energy expenditure increased with an average non-shivering thermogenesis of 20.1 [3.3-41.3] %, no relation between BMR and BAT activity was found. This is in line with some [23, 38, 39], but not all [40, 41] previous studies using healthy young subjects. Absence of a relation between BMR and BAT activity and between non-shivering thermogenesis and BAT activity could be due to the low overall BAT activity in the current study population (also certainly related in part to the age). The SUV value of BAT was 1.5 [0.1-6.2], which was considerably lower than reported by others [18, 42, 43] who found SUV values in

the range of 2-18 among 20-30 year old healthy study subjects. The relatively low metabolic rate of BAT depots may not be entirely unexpected, as BAT is inversely correlated with age, and BAT is barely detected in healthy subjects above 60 years old [44]. Nevertheless, one autopsy study [45] and one retrospective imaging study [46] reported the presence of BAT activity among elderly with cancer, suggesting that independent of age, disease might be a trigger for BAT activity. However, in those studies [45, 46] no data are presented considering the stage of cancer. Furthermore, regarding the available imaging study, the results might be biased by the retrospective design, which did not allow to control environmental temperatures. In male patients, 2.8% of scans demonstrated BAT activity compared with 7.2% of scans in female patients [46].

Systemic inflammation has been proposed as another trigger for hypermetabolism and also as BAT activator in cancer cachexia [47, 48]. In the studied patients with clinically stable COPD, systemic inflammation assessed by CRP was low and not associated with BMR. Additionally, no altered gene expression of inflammation or macrophage markers was observed compared to healthy controls. In this study also two pancreatic cancer patients were enrolled. In line with the results in COPD patients, no BAT activity was found.

Interestingly, despite that no BAT activity was detected, the non-shivering thermogenesis was relatively high in the whole study cohort. Non-shivering thermogenesis reported in BAT negative healthy young subjects was on average 4% compared to 30% in BAT positive subjects [18]. Others reported average non-shivering thermogenesis of 12% in BAT positive subjects [41, 49, 50], which is remarkably lower compared to our results. No correlation between non-shivering thermogenesis and SUV of skeletal muscles was found, arguing that the elevated energy expenditure upon cold could not be attributed to involuntary muscle contractions. Furthermore, some studies indicate a role of insulin resistance that point towards either impaired glucose uptake by BAT [50] or reduced BAT activity [51]. However, the current cohort was not insulin resistant. Another explanation for the relatively high non-shivering thermogenesis might be that there is an effect of diet-induced energy expenditure. Three hours before cold-induced energy expenditure was assessed, participants had a light breakfast. In general, after three hours most of the thermic effect of a meal dissolved [52]. Although individual differences might exist, this cannot fully explain the high non-shivering thermogenesis.

This study is not without limitations. The major limitation in this study was the smaller number of enrolled subjects than intended for multiple group comparison (10 subjects per subgroup, in total 50 subjects). Accrual of COPD patients was hampered by the exclusion of subjects using medication that influences the sympathetic nerve system, including ß-blockers and calcium channel blockers. Partly due to shared risk factors such as smoking, cardiovascular diseases commonly coincide COPD [53]. Therefore, ß-blockers and calcium channel blockers are widely

used by COPD patients. Another factor contributing to low accrual was the use of MRI instead of computed tomography, since a significant part of the COPD patients who were approached to participate suffered from claustrophobia. This might not be totally unexpected because the patients already suffered from dyspnoea, which might be reinforced by MRI scanning. In addition, it appeared impossible to include lung cancer patients and only two pancreatic cancer patients were enrolled since the extensive study methodologies turned out to be too severe in this critically ill population. [54-56].

Nevertheless, the consistency of our findings in the hypermetabolic COPD patients and confirmed in the two cancer cases do not support a role for BAT activation as trigger of disease induced hypermetabolism and cachexia.

REFERENCES

- Vogelmeier CF, Criner GJ, Martinez FJ, et al. Global Strategy for the Diagnosis, Management, and Prevention of Chronic Obstructive Lung Disease 2017 Report. GOLD Executive Summary. Am J Respir Crit Care Med 2017; 195: 557-82.
- Schols AM, Soeters PB, Mostert R, et al. Energy balance in chronic obstructive pulmonary disease. Am Rev Respir Dis 1991; 143: 1248-52.
- Hugli O, Schutz Y, Fitting JW. The daily energy expenditure in stable chronic obstructive pulmonary disease. Am J Respir Crit Care Med 1996; 153: 294-300.
- Slinde F, Ellegard L, Gronberg AM, et al. Total energy expenditure in underweight patients with severe chronic obstructive pulmonary disease living at home. Clin Nutr 2003; 22: 159-65.
- Baarends EM, Schols AM, Akkermans MA, et al. Decreased mechanical efficiency in clinically stable patients with COPD. Thorax 1997; 52: 981-6.
- Baarends EM, Schols AM, Pannemans DL, et al. Total free living energy expenditure in patients with severe chronic obstructive pulmonary disease. Am J Respir Crit Care Med 1997; 155: 549-54.
- Hugli O, Frascarolo P, Schutz Y, et al. Diet-induced thermogenesis in chronic obstructive pulmonary disease. Am Rev Respir Dis 1993; 148: 1479-83.
- Dore MF, Laaban JP, Orvoen-Frija E, et al. Role of the thermic effect of food in malnutrition of patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med 1997; 155: 1535-40.
- Cohen RI, Marzouk K, Berkoski P, et al. Body composition and resting energy expenditure in clinically stable, non-weight-losing patients with severe emphysema. Chest 2003; 124: 1365-72.
- Green JH, Muers MF. The thermic effect of food in underweight patients with emphysematous chronic obstructive pulmonary disease. Eur Respir J 1991; 4: 813-9.
- Baarends EM, Schols AM, Westerterp KR, et al. Total daily energy expenditure relative to resting energy expenditure in clinically stable patients with COPD. Thorax 1997; 52: 780-5.
- Straub RH, Cutolo M, Buttgereit F, et al. Energy regulation and neuroendocrine-immune control in chronic inflammatory diseases. J Intern Med 2010; 267: 543-60.
- Broekhuizen R, Wouters EF, Creutzberg EC, et al. Raised CRP levels mark metabolic and functional impairment in advanced COPD. Thorax 2006; 61: 17-22.

- Donahoe M, Rogers RM, Wilson DO, et al. Oxygen consumption of the respiratory muscles in normal and in malnourished patients with chronic obstructive pulmonary disease. Am Rev Respir Dis 1989; 140: 385-91.
- Petruzzelli M, Schweiger M, Schreiber R, et al. A switch from white to brown fat increases energy expenditure in cancer-associated cachexia. Cell Metab 2014; 20: 433-47.
- De Vos-Geelen J, Fearon KC, Schols AM. The energy balance in cancer cachexia revisited. Curr Opin Clin Nutr Metab Care 2014; 17: 509-14.
- Heaton GM, Wagenvoord RJ, Kemp A, Jr., et al. Brown-adipose-tissue mitochondria: photoaffinity labelling of the regulatory site of energy dissipation. Eur J Biochem 1978; 82: 515-21.
- Yoneshiro T, Aita S, Matsushita M, et al. Brown adipose tissue, whole-body energy expenditure, and thermogenesis in healthy adult men. Obesity (Silver Spring) 2011; 19: 13-6.
- Claessens-Van Ooijen AM, Westerterp KR, Wouters L, et al. Heat production and body temperature during cooling and rewarming in overweight and lean men. Obesity (Silver Spring) 2006; 14: 1914-20.
- Cypess AM, Lehman S, Williams G, et al. Identification and importance of brown adipose tissue in adult humans. N Engl J Med 2009; 360: 1509-17.
- Wang Q, Zhang M, Xu M, et al. Brown adipose tissue activation is inversely related to central obesity and metabolic parameters in adult human. PLoS One 2015; 10: e0123795.
- Lee P, Greenfield JR, Ho KK, et al. A critical appraisal of the prevalence and metabolic significance of brown adipose tissue in adult humans. Am J Physiol Endocrinol Metab 2010; 299: E601-6.
- Van Marken Lichtenbelt WD, Vanhommerig JW, Smulders NM, et al. Cold-activated brown adipose tissue in healthy men. N Engl J Med 2009; 360: 1500-8.
- Hofford JM, Milakofsky L, Vogel WH, et al. The nutritional status in advanced emphysema associated with chronic bronchitis. A study of amino acid and catecholamine levels. Am Rev Respir Dis 1990; 141: 902-8.
- Atgie C, D'allaire F, Bukowiecki LJ. Role of beta1- and beta3-adrenoceptors in the regulation of lipolysis and thermogenesis in rat brown adipocytes. Am J Physiol 1997; 273: C1136-42.
- 26. Franssen FM, Sauerwein HP, Ackermans MT, et al.

Increased postabsorptive and exercise-induced whole-body glucose production in patients with chronic obstructive pulmonary disease. Metabolism 2011; 60: 957-64.

- Green HJ, Burnett ME, D'arsigny CL, et al. Altered metabolic and transporter characteristics of vastus lateralis in chronic obstructive pulmonary disease. J Appl Physiol (1985) 2008; 105: 879-86.
- Green HJ, Bombardier E, Burnett M, et al. Organization of metabolic pathways in vastus lateralis of patients with chronic obstructive pulmonary disease. Am J Physiol Regul Integr Comp Physiol 2008; 295: R935-41.
- Engelen MP, Schols AM, Does JD, et al. Exerciseinduced lactate increase in relation to muscle substrates in patients with chronic obstructive pulmonary disease. Am J Respir Crit Care Med 2000; 162: 1697-704.
- 30. Bakker LE, Boon MR, Van Der Linden RA, et al. Brown adipose tissue volume in healthy lean south Asian adults compared with white Caucasians: a prospective, case-controlled observational study. Lancet Diabetes Endocrinol 2014; 2: 210-7.
- Van Der Lans AA, Wierts R, Vosselman MJ, et al. Coldactivated brown adipose tissue in human adults: methodological issues. Am J Physiol Regul Integr Comp Physiol 2014; 307: R103-13.
- Weir JB. New methods for calculating metabolic rate with special reference to protein metabolism. 1949. Nutrition 1990; 6: 213-21.
- Westerterp KR, Wouters L, Van Marken Lichtenbelt WD. The Maastricht protocol for the measurement of body composition and energy expenditure with labeled water. Obes Res 1995; 3 Suppl 1: 49-57.
- Rabinovich RA, Louvaris Z, Raste Y, et al. Validity of physical activity monitors during daily life in patients with COPD. Eur Respir J 2013; 42: 1205-15.
- Van Marken Lichtenbelt WD, Daanen HA, Wouters L, et al. Evaluation of wireless determination of skin temperature using iButtons. Physiol Behav 2006; 88: 489-97.
- 36. Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. Am J Respir Crit Care Med 2006; 173: 79-83.
- Klingenberg M, Huang SG. Structure and function of the uncoupling protein from brown adipose tissue.

Biochim Biophys Acta 1999; 1415: 271-96.

- Ouellet V, Labbe SM, Blondin DP, et al. Brown adipose tissue oxidative metabolism contributes to energy expenditure during acute cold exposure in humans. J Clin Invest 2012; 122: 545-52.
- Muzik O, Mangner TJ, Leonard WR, et al. 150 PET measurement of blood flow and oxygen consumption in cold-activated human brown fat. J Nucl Med 2013; 54: 523-31.
- Yoneshiro T, Aita S, Matsushita M, et al. Recruited brown adipose tissue as an antiobesity agent in humans. J Clin Invest 2013; 123: 3404-8.
- Van Der Lans AA, Hoeks J, Brans B, et al. Cold acclimation recruits human brown fat and increases nonshivering thermogenesis. J Clin Invest 2013; 123: 3395-403.
- Vosselman MJ, Hoeks J, Brans B, et al. Low brown adipose tissue activity in endurance-trained compared with lean sedentary men. Int J Obes (Lond) 2015; 39: 1696-702.
- Van Der Lans AA, Vosselman MJ, Hanssen MJ, et al. Supraclavicular skin temperature and BAT activity in lean healthy adults. J Physiol Sci 2016; 66: 77-83.
- Yoneshiro T, Aita S, Matsushita M, et al. Age-related decrease in cold-activated brown adipose tissue and accumulation of body fat in healthy humans. Obesity (Silver Spring) 2011; 19: 1755-60.
- 45. Shellock FG, Riedinger MS, Fishbein MC. Brown adipose tissue in cancer patients: possible cause of cancer-induced cachexia. J Cancer Res Clin Oncol 1986; 111: 82-5.
- Au-Yong IT, Thorn N, Ganatra R, et al. Brown adipose tissue and seasonal variation in humans. Diabetes 2009; 58: 2583-7.
- Rutten EP, Spruit MA, Mcdonald ML, et al. Continuous fat-free mass decline in COPD: fact or fiction? Eur Respir J 2015; 46: 1496-8.
- Hopkinson NS, Tennant RC, Dayer MJ, et al. A prospective study of decline in fat free mass and skeletal muscle strength in chronic obstructive pulmonary disease. Respir Res 2007; 8: 25.
- 49. Broeders EP, Vijgen GH, Havekes B, et al. Thyroid Hormone Activates Brown Adipose Tissue and Increases Non-Shivering Thermogenesis--A Cohort Study in a Group of Thyroid Carcinoma Patients. PLoS One 2016; 11: e0145049.
- Hanssen MJ, Wierts R, Hoeks J, et al. Glucose uptake in human brown adipose tissue is impaired upon fasting-induced insulin resistance. Diabetologia 2015; 58: 586-95.

- Lean ME, Murgatroyd PR, Rothnie I, et al. Metabolic and thyroidal responses to mild cold are abnormal in obese diabetic women. Clin Endocrinol (Oxf) 1988; 28: 665-73.
- 52. Reed GW, Hill JO. Measuring the thermic effect of food. Am J Clin Nutr 1996; 63: 164-9.
- Mullerova H, Agusti A, Erqou S, et al. Cardiovascular comorbidity in COPD: systematic literature review. Chest 2013; 144: 1163-78.
- Vosselman MJ, Van Der Lans AA, Brans B, et al. Systemic beta-adrenergic stimulation of thermogenesis is not accompanied by brown adipose tissue activity in humans. Diabetes 2012; 61: 3106-13.
- Cypess AM, Chen YC, Sze C, et al. Cold but not sympathomimetics activates human brown adipose tissue in vivo. Proc Natl Acad Sci U S A 2012; 109: 10001-5.
- Cypess AM, Weiner LS, Roberts-Toler C, et al. Activation of human brown adipose tissue by a beta3-adrenergic receptor agonist. Cell Metab 2015; 21: 33-8.

SUPPLEMENTALS

	COPD	Controls
Blood analysis		
C-reactive protein, mg/L	2.5 (2.0 to 21.0)	0.0 (0.0 to 3.0)
Fasting glucose, mmol/L	5.8 ± 1.0	4.9 ± 0.5
Insulin, pmol/L	27.2 (12.0 to 86.6)	13.0 (12.0 to 19.0)
Thyroid stimulating hormone, mU/L	1.8 (1.0 to 7.2)	1.8 (1.5 to 2.8)
Free T4, pmol/L	14.6 (1.7 to 20.5)	13.6 (12.3 to 18.1)
Free fatty acids, mmol/L	0.7 (0.4 to 1.8)	0.6 (0.3 to 0.9)
Cholesterol, mmol/L	4.8 ± 0.7	4.5 ± 0.7
High-density lipoprotein, mmol/L	1.6 ± 0.3	1.7 ± 0.2
Low-density lipoprotein, mmol/L	2.8 ± 0.8	2.4 ± 0.7
Triglycerides, mmol/L	1.0 ± 0.3	0.9 ± 0.3
qPRC subcutaneous fat biopsies		
BAT and beige markers, AU		
Cidea	-6.14 (-6.85 to -5.47)	-6.00 (-6.25 to -5.77)
TMEM26	-8.99 (-9.47 to -8.82)	-8.84 (-12.29 to -8.76)
CD137	-8.60 (-9.44 to -8.40)	-9.14 (-9.75 to -8.31)
Hypoxia markers, AU		
	-7.63 (-7.86 l0 -7.45)	-7.51(-7.86[0-7.45))
VEGE-a	-7.62 (-7.66 to -7.53)	-7.61 (-7.89 to -7.15)
Inflammation and macrophages markers All	1.02 (1.00 to 1.00)	7.01 (7.05 to 7.15)
III-8		
MCP-1	-7.79 (-8.61 to -7.25)	-8.54 (-9.59 to -7.48)
CD68	-7.35 (-7.75 to -6.98)	-7.62 (-7.75 to -7.22)
CD163	-6.75 (-7.26 to -6.54)	-6.87 (-7.09 to -6.29)
CD206	-6.74 (-7.26 to -6.48)	-7.00 (-7.25 to -6.39)
CD11b	-7.11 (-7.59 to -6.81)	-7.42 (-7.65 to -6.88)
	-7.98 (-8.80 to -7.81)	-8.34 (-8.80 to -7.59)
Glycolysis and lipolysis markers, AU		
HKII	-6.22 (-6.41 to -6.03)	-6.26 (-6.51 to -6.11)
CGI58	-6.38 (-6.52 to -6.18)	-6.32 (-6.50 to -6.16)
ATGL	-5.90 (-6.12 to -5.81)	-5.95 (-6.22 to -5.67)
HSL	-6.13 (-6.20 to -5.85)	-6.10 (-6.24 to -5.91)
Perilipin1	-6.05 (-6.12 to -5.76)	-6.04 (-6.21 to -5.77)

Supplemental table 1. Blood analysis and qPRC analysis of subcutaneous fat biopsies of COPD patients (N=10) and healthy controls (N=6).

Supplemental table 1 Continues.

••		
	COPD	Controls
qPRC subcutaneous fat biopsies		
Adipokine markers, AU		
PAI-1	-8.25 (-8.45 to -7.91)	-8.31 (-8.59 to -7.89)
Adrenomedullin	-7.29 (-7.52 to -6.88)	-7.43 (-7.80 to -7.13)
Visfatin	-6.41 (-6.87 to -6.15)	-6.71 (-6.86 to -6.06)
Osteoprotegrin	-7.92 (-8.03 to -7.54)	-8.05 (-8.14 to -7.89)
Chemerin	-5.84 (-5.99 to -5.67)	-5.89 (-6.14 to -5.70)
Adiponectin	-5.57 (-5.80 to -5.33)	-5.49 (-5.81 to -5.28)
Leptin	-5.82 (-6.14 to -5.55)	-6.16 (-6.26 to -5.70)
Irisin	-8.48 (-8.54 to -8.18)	-8.30 (-8.70 to -8.17)

qPCR data are visualised by log 10 normalized mRNA expression in arbitrary units.

AU, arbitrary units.

Data are represented as mean \pm SD or median (range).

Brown adipose tissue in COPD



8

Early weight loss during chemoradiotherapy has detrimental impact on outcome in non-small cell lung cancer

Karin J.C. Sanders, Lizza E. Hendriks, Esther G.C. Troost, Gerben P. Bootsma, Ruud M.A. Houben, Annemie M.W.J. Schols, Anne-Marie C. Dingemans.

Journal of Thoracic Oncology. 2016;11(6):873-9.

ABSTRACT

The aim of this study was to assess the effect of early weight loss before the onset of radiation esophagitis on overall survival (OS) in non-small cell lung cancer (NSCLC) patients treated with concurrent chemoradiotherapy.

Characteristics (e.g., patient weight, radiation esophagitis score, gender, World Health Organisation performance status, chemotherapy dose, nodal status, and gross tumour volume) of 151 patients who received concurrent chemoradiotherapy (in 2006-2013) were retrospectively correlated with OS. Early weight loss was defined as weight loss of more than 5% between the start and third week of radiotherapy in patients whose weight was stable before treatment initiation.

In 17% of the patients early weight loss was observed. Median OS (95% confidence interval [CI]) was significantly shorter in the early weight loss group (OS = 13.0 months, 95% CI: 2.0-24.0) versus in the non-early weight loss group (OS = 23.0 months, 95% CI: 14.7-31.3) (hazard ratio [HR] = 1.8, 95% CI: 1.12-2.96, p = 0.017). On multivariate analysis gender (HR = 2.1, 95% CI 1.33-3.29, p = 0.001), World Health Organization performance status (HR = 1.9, 95% CI 1.20-2.97, p = 0.006), nodal status (HR = 2.9, 95% CI 1.38-6.01, p = 0.005) and early weight loss (HR = 1.9, 95% CI 1.10-3.19, p = 0.022) were associated with OS.

Early weight loss in patients with non-small cell lung cancer was found to be associated with worse prognosis. These data warrant further investigation into the efficacy of tailored intervention to prevent early weight loss.

INTRODUCTION

Attention towards management of involuntary weight loss in patients with non-small cell lung cancer (NSCLC) is increasing [1]. Although unintended weight loss was initially considered to be an indicator of terminal progression of the disease, there is now convincing evidence that it is an independent determinant of survival [2]. Besides their implications for survival, weight loss and muscle wasting have a detrimental impact on response to chemotherapy and quality of life [3]. Not only tumour-related factors but also treatment-related factors can contribute to weight loss and muscle wasting [4].

Because of improved survival rates in comparison with sequential chemoradiotherapy, concurrent chemoradiotherapy (CCRT) has become the standard treatment for patients with unresectable locally advanced NSCLC [5]. However, compared with sequential chemoradiotherapy, CCRT has more serious side effects including oesophageal toxicity [6]. Grade 3 to 4 dysphagia due to acute radiation esophagitis, which impairs spontaneous dietary intake resulting in weight loss, will develop in approximately 20% of patients [3, 7]. Symptoms of radiation esophagitis gradually start at the end of week 4 of CCRT and commonly dissipate 1 to 3 weeks after completion of therapy [7]. Nevertheless, our group has previously shown that weight loss may already occur in the first 3 weeks of CCRT, before onset of clinically relevant radiation esophagitis and irrespective of decreased dietary intake [8]. Furthermore, this early weight loss was accompanied by a rapid decline of muscle strength in the lower limbs. Taken together, the presence of weight loss and muscle weakness is indicative of early metabolic alterations that may warrant tailored nutritional or multimodal intervention.

The objective of the current study was to evaluate the impact of early weight loss (>5% between the start and third week of radiotherapy [RT]) on overall survival (OS) in patients with stable weight before initiation of treatment.

METHODS

Patient selection

Medical records of all consecutive patients with advanced stage NSCLC (stage IIIA, IIIB and oligometastatic stage IV) who were undergoing a radical course of RT in a single institute but receiving concurrent chemotherapy in one of two neighbouring Dutch teaching hospitals between January 2006 and December 2013 were retrieved and retrospectively analysed. For staging purposes all patients received a diagnostic 18F-fluorodeoxyglucose positron emission tomography (FDG-PET) scan combined with a computed tomography (CT) scan and magnetic resonance imaging or CT of the brain. Patients were excluded from this analysis if they had a neuroendocrine carcinoma, another malignancy within 5 years of diagnosis, pre-existing swallowing difficulties, weight loss of more than 5% preceding lung cancer treatment, weight loss before diagnosis, unknown and/or no weight recorded at the start of RT or week 3 of RT, or

an uncompleted course of CCRT. This retrospective study was approved by the medical ethical committee of the Maastricht University Medical Centre+ and the Institutional Review Board of the Department of Radiation Oncology (MAASTRO clinic, the Netherlands).

Treatment

Patients were treated with CCRT. Chemotherapy was administered according to national and international guidelines (chemotherapy treatment scheme displayed in table 1). Generally, patients received one or more cycles of induction therapy followed by CCRT. The concurrent chemotherapy regimen in most cases consisted of platinum-based chemotherapy combined with etoposide. Other regimens used included cisplatin-vinorelbine, cisplatin-vinorelbine-cetuximab [9] and carboplatin-vinorelbine.

Two different RT schedules were used. The first consisted of individualized hyperfractionated accelerated external beam RT [10]. The schedule consisted of a maximum total dose of 69 Gy, first delivered in fractions of 1.5 Gy twice-daily up to 45 Gy, followed by an individualized dose of 8 to 24 Gy delivered in once-daily 2.0 Gy fractions, limited by the dose to surrounding organs at risk (e.g., mean lung dose and maximum dose to the spinal cord). The second was the delivery of RT within the PET-boost trial, in which the radiation dose to the entire tumour volume or on FDG-avid subvolumes is escalated until normal tissue dose constraints have been reached (see van Elmpt et al. [11] for details). RT planning was performed on the XiO system (Elekta Instrument AB, Stockholm, Sweden) until July 2012, using a convolution–superposition algorithm with inhomogeneity corrections and according to International Commission on Radiation Units & Measurements 50 guidelines. As of July 2012, radiation treatment was planned using RapidArc (Eclipse version 11.0, Varian Medical Systems, Palo Alto, CA), with a type B dose calculation algorithm (AcurosXB-10.0 [Varian Medical Systems]).

Data collection

All medical records were retrospectively reviewed. Data collected included patient demographics (age and gender); World Health Organization performance status (WHO PS); medical history; Charlson comorbidity score; clinical tumour stage and clinical nodal stage (cN); tumour, node, and metastasis (TNM) disease stage (according to TNM6 until 2009 or TNM7 from 2010 onward); weight loss before initiation of CCRT; premorbid weight, height and body mass index; weight weekly assessed from start of CCRT until 4 weeks after completion of CCRT; chemotherapy used (type and dose reductions); hospitalization and episodes of febrile neutropenia.

From the treatment planning system, the gross tumour volume (GTV) encompassing the primary tumour and the metastastic hilar/mediastinal lymph nodes (confirmed by histopathology after endobronchial/endoesophageal ultrasonography, enlarged or presence of malignant features on CT scan, and/or FDG-PET positivity) was retrieved. Additionally, RT parameters

were extracted, including total RT dose and mean oesophagus dose.

Patients were monitored weekly by a dietician. During and after completion of treatment, patients were followed up weekly to evaluate weight and onset/resolution of acute radiation esophagitis. Radiation esophagitis was evaluated according to the Common Toxicity Criteria version 3.0 grading scale. Local follow-up policy to evaluate tumour response included a CT scan 3 months after completion of RT, followed by three monthly chest X-ray and annual CT scan.

The primary objective of this study was to assess the impact of early weight loss on OS. Early weight loss was defined as weight loss of more than 5% between the start and third week of RT. A cut-off limit of 5% was chosen according to the international consensus of cancer cachexia [12]. The secondary objective included the determination of the contribution of gender, age, WHO PS, tumour- and nodal status, histological diagnosis, and early weight loss to OS. In addition, the relation between aforementioned parameters and progression-free survival (PFS) was evaluated.

Statistical analysis

All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, Version 22.0, IBM, Armonk, NY). Descriptive statistics of demographic and clinical variables were obtained. Means (± SD) were provided for continuous variables and percentages were shown for categorical variables. Continuous variables were compared using Student's t-test for independent samples, categorical variables were compared using chi-square test, and nonparametric variables were tested using Mann-Whitney U test. OS was calculated from day of pathologically proven diagnosis of NSCLC until death. PFS was defined as day of pathologically proven diagnosis of NSCLC until first documented disease progression assessed on a CT scan or death. For survival analysis, patients lost to follow-up were censored at the time of their last visit or observation. The probabilities of both OS and PFS were calculated using the Kaplan-Meier method. Survival curves of patients with and without early weight loss were tested for significance using the log-rank test. To assess the contribution of early weight loss to OS, multivariate Cox regression analysis was performed with several clinical parameters, including gender, age, WHO PS, tumour and nodal status, histological diagnosis, and early weight loss as independent variables. Proportional hazard assumption was tested using visual inspection of log-minus-log survival plots. Results with two-sided exact p values (<0.05) were considered statistically significant.

RESULTS

Patient and treatment characteristics

A total of 287 patients with NSCLC were identified during the study period. One hundred thirtysix patients were excluded because of histological diagnosis (5), another malignancy within 5 years of the diagnosis of NSCLC (15), pre-existing swallowing difficulties (11), weight loss of more than 5% before treatment (58), missing weight records before treatment (9), start of RT (9), or week 3 of RT (15), and no completion of CCRT (14). As a result, 151 patients were eligible for analysis (figure 1).



Figure 1. Flow chart study population.

Baseline patient and treatment characteristics are described in table 1. The median age was 63 years and 42% of the participants were female. Seventy-five patients with stage IIIA disease, 70 patients with stage IIIB disease, and six patients with oligometastatic stage IV disease were included. Twenty-six patients (17%) experienced weight loss of more than 5% during the first three weeks of RT. Mean (\pm SD) weight loss in the non-early weight loss group and early weight loss group were -0.5 \pm 2.3 kg and -5.4 \pm 1.6 kg, respectively. The course of mean weight loss during therapy is displayed in supplemental data.

Table 1. Comparison of chemotherapy r	egimen between patient:	s without early weight lo	ss and with
early weight loss.			

	≤5% early weight loss (N=125)	>5% early weight loss (N=26)
Number induction chemotherapy cycles $0/1/2/3$	11.2 / 72.0 / 11.2 / 2.4	3.8 / 76.9 / 7.7 / 11.5
Platinum-based induction chemotherapy Cisplatin / carboplatin	52.8 / 33.6	61.5 / 34.6
Type induction chemotherapy Etoposide / gemcitabine / paclitaxel / vinorelbine	35.2 / 40.8 / 1.6 / 0.8	42.3 / 42.3 / 7.7 / 3.8
No. of concurrent chemotherapy cycles 2 / 3	86.4 / 12.0	92.3 / 3.8
Platinum-based concurrent chemotherapy Cisplatin / carboplatin	81.6 / 16.8	84.6 / 11.5

Table 1. Continued.		
Type concurrent chemotherapy Etoposide / vinorelbine / vinorelbine- cetuximab	68.0 / 24.0 / 6.4	61.5 / 26.9 / 3.8

Data are presented as %.

Patients stratified by early weight loss were comparable with regard to age at diagnosis, WHO PS, and number and severity of comorbidities. There was a trend towards higher nodal staging and GTV in the patients with early weight loss; however this was not statistically significant and the GTV had a large and overlapping range. Furthermore, treatment regimens were comparable with respect to RT (table 2).

 Table 2. Comparison of clinical and treatment characteristics between patients without early weight loss and with early weight loss.

	≤5% early weight loss (N=125)	>5% early weight loss (N=26)	P value
Age, years	63 ± 9.5	61 ± 9.9	0.276
Male/female, %	56.0/44.0	69.2/30.8	0.276
Weight change, kg	-0.5 ± 2.3	-5.4 ± 1.6	0.000
Weight change prior to treatment initiation, kg	-0.58 ± 1.2	-0.92 ± 1.5	0.287
Weight change during induction chemotherapy, kg	-0.27 ± 2.6	0.76 ± 3.1	0.077
WHO performance status 0 / 1, 2, %	37.6 / 62.4	50.0 / 50.0	0.274
Charlson comorbidity index	5 ± 1.5	4 ± 1.8	0.208
Disease stage, % IIIA / IIIB / IV	52.8 / 43.2 / 4.0	34.6 / 61.5 / 3.9	0.213
Clinical tumour stage, % Tx, T1, T2 T3, T4	50.8 49.2	34.6 65.4	0.195
Clinical nodal status, % N0, N1 N2, N3	17.5 82.5	3.8 96.2	0.129
Histology, % Squamous Adenocarcinoma Large cell carcinoma NSCLC NOS	24.6 42.1 12.5 20.8	42.3 23.1 15.4 19.2	0.092
Total dose radiotherapy (Gy)*	65.5 (60.1- 69.0)	69.0 (65.0- 69.0)	0.132
Gross tumour volume*	79.9 (51.2- 158.2)	181.7 (43.5- 280.8)	0.248
Mean oesophagus dose (Gy)*	20.0 (14.0- 26.3)	18.2 (16.5- 29.2)	0.477

Data are presented as mean \pm SD or %, unless indicated otherwise. *Median (interquartile range). Abbreviations: NSCLC NOS, non-small cell lung cancer not otherwise specified.
Adverse events

No statistically significant differences were found between the groups with respect to adverse events (table 3). Percentage of patients receiving chemotherapy dose reductions was not statistical significantly different between the early weight loss and non-early weight loss group. Additionally, no significant differences were found for hospitalization rates, neutropenic fever episodes, and incidence of esophagitis.

	≤5% early weight loss (N=125)	>5% early weight loss (N=26)	P value
Chemotherapy dose reduction	11.2	19.2	0.339
Hospitalization	55.2	73.1	0.122
Neutropenic fever	18.4	34.6	0.112
Esophagitis week 3 radiotherapy grade ≥2	32.5	32.0	1.000

Table 3. Comparison of adverse events between patients without early weight loss and with early weight loss.

Data are presented as %.

Survival

Median follow-up was 40.0 months (range 28.4 to 51.7), and at the time of analysis 86 patients (56.6%) had died. Median OS (95% confidence interval [CI]) was significantly shorter in the early weight loss group compared with the non-early weight loss group (OS = 13.0 months, 95% CI: 2.0-24.0 versus OS = 23.0 months, 95% CI: 14.7-31.3) (hazard ratio [HR] = 1.8, 95% CI: 1.12-2.96, p = 0.017; [figure 2a]). Although not statistically significant, a trend was observed regarding the influence of early weight loss group, with a median PFS of 8.0 months (95% CI: 5.5-10.5) versus 15.0 months (95% CI: 9.5-20.5) (HR = 1.5, 95% CI: 0.95-2.51, p = 0.08 [figure 2b]).



Figure 2. Survival analyses. (a) Kaplan Meier OS curve; (b) Kaplan Meier PFS curve of patients experiencing >5% early weight loss (solid line) compared to patients with ≤5% early weight loss (dotted line).

On multivariate analysis gender (HR = 2.1, 95% CI: 1.33-3.29, p = 0.001), WHO PS (HR = 1.9, 95% CI: 1.20-2.97; p = 0.006), nodal status (HR = 2.9, 95% CI: 1.38-6.01, p = 0.005) and early weight loss (HR = 1.9, 95% CI: 1.10-3.19, p = 0.022) were associated with OS (table 4). Gender (HR = 2.4, 95% CI: 1.48-3.79, p < 0.001), WHO PS (HR = 1.9, 95% CI: 1.18-2.98, p = 0.008), nodal status (HR = 2.4, 95% CI: 1.16-5.14, p = 0.019), histology (HR = 2.4, 95% CI: 1.33-4.17, p = 0.003) and early weight loss (HR = 1.8, 95% CI: 1.01-3.05, p = 0.046) were associated with PFS (data not shown).

Variable HR (95% CI) P value Gender female vs male 2.092 (1.329-3.293) 0.001 Age at diagnosis <65 vs ≥65 years 1.353 (0.889-2.061) 0.159 WHO performance status 0 vs 1-2 1.888 (1.201-2.967) 0.006 Clinical tumour status cTx, cT1, cT2 vs cT3, cT4 1.304 (0.839-2.027) 0.238 Clinical nodal status cN0, cN1 vs cN2, cN3 2.883 (1.382-6.015) 0.005 Histology non-squamous vs squamous 1.635 (0.950-2.814) 0.076 Weight stable vs early weight loss 1.868 (1.095-3.187) 0.022

DISCUSSION

Despite weekly advice from a dietician, in our retrospective analysis 17% of patients who were weight stable before curative intent CCRT experienced early weight loss and this adversely affected overall survival (HR = 1.9, 95% CI: 1.10-3.19, p = 0.022). Furthermore, early weight loss could not be attributed to a higher incidence of radiation esophagitis, restricted chemotherapy dose, or differences in GTVs. Therefore, early weight loss may provide predictive information additional to known prognostic factors, such as TNM staging, WHO PS and comorbidities. Attention towards early weight loss despite stable nutritional intake is therefore mandatory.

Weight loss is a frequent problem in lung cancer, as up to 54% of patients with NSCLC report weight loss before diagnosis, including weight loss of 10% or more in up to 30% of patients with lung cancer [2, 13]. Previously, patients with stage IIIB NSCLC with and without weight loss during treatment were compared, and it was reported that the latter group had a statistically significantly superior OS and PFS [14]. However, in that study weight loss was defined as a reduction body mass index of more than 0.5 kg/m² at the end of CCRT. Furthermore, among weight-losing patients, the need for reduction of chemotherapy dose or abandonment of chemotherapy as well as the incidence of grade 3 or 4 adverse events were significantly higher. At present, there are no published data on the effects of weight loss occurring before the onset of radiation esophagitis on OS and PFS. Therefore, it is unclear whether the weight loss by itself or the adverse events and/or restricted chemotherapy/RT dose was cause of the worse prognosis. It is of note that weight loss is associated with delivery of fewer chemotherapy

145

cycles and increased treatment delay [15]. It is therefore unknown from the previous literature whether weight loss is a causal factor of decreased survival or merely a symptom that occurs during disease progression. This is the first study showing the impact of early weight loss on outcome in previously weight stable patients, independent of treatment characteristics, treatment side effects and GTV.

Our observations regarding early weight loss are in line with body weight changes during CCRT in other malignancies. Weight loss has been observed from initiation of CCRT until several weeks thereafter in patients treated for head and neck cancer [16]. Recently however, Ghadjar et al. [17] found no association between weight loss during treatment and survival outcomes in head and neck cancer patients. Nonetheless, weight loss in this tumour type is often attributed to tumour- and treatment-related swallowing difficulties. Notably, however, the authors only recorded weight at week 1 and week 5 of RT and no dysphagia data were presented. Furthermore, a trend towards more advanced disease and higher WHO PS in the weight-losing group may obscure its impact on survival.

Although it did not reach statistical significance in univariate analysis, we observed a trend concerning the impact of early weight loss on PFS. This is in contrast with Topkan et al. [14] reporting a clear statistically significant difference in median PFS of 4.6 months in patients with stage IIIB NSCLC. However we recognize that (both in the study of Topkan et al. and in the current study), follow-up was not standardised and the time-line of follow-up was – owing to the retrospective analysis of the data - not uniformly defined. This limits us to draw hard conclusions about the impact of early weight loss on PFS.

From RT studies it is known that a large GTV, representing the tumour load, is associated with worse survival rates [18,19]. In our study, GTVs in the early weight loss group were larger even though not statistically significant, but no effect of the GTV on OS or PFS was found. Additionally, cN2-cN3 involvement tended to be somewhat higher among the weight loss group, again not statistically significant. This may indicate that in part the lower survival rates seen in the early weight loss group may be attributed to more advanced disease. Nevertheless, early weight loss adjusted for disease stage was found to have a statistically significant detrimental impact on OS in multivariate analysis.

Although the strength of our study is the well-defined patient cohort, there are also some limitations. The day of pathological proven diagnosis of NSCLC was used as starting point of the current study. However, we recognize that the time-line of diagnosis until start of the treatment may be different between subjects. Weight measurement was not standardized, with respect to the use of different scales and different time points of measurement during the day. We tried to overcome these differences by define early weight loss as weight loss of

more than 5%, which is considered to be beyond the range explained by the aforementioned factors. Nevertheless, patients were closely monitored by a dietician, who was able to start nutritional intervention, and therefore attention was paid to weight change. The nature of weight loss is unclear because there are no data available about body composition changes and putative catabolic stimuli like systemic inflammation. It is well established that enhanced levels of inflammatory mediators associate with weight loss [4, 20].

In conclusion, the present study provides evidence for the negative impact of early weight loss on outcome in patients with NSCLC who are receiving curative intent CCRT. Lower survival rates were demonstrated for patients experiencing weight loss of more than 5% during the first 3 weeks of treatment. Given the anabolic potential of dietary protein, we suggest aggressive supplementation of proteins initiated from start of CCRT, in combination with a tailored exercise regime to combat loss of muscle mass and strength. The feasibility and effectiveness of early intervention to prevent or postpone CCRT-induced weight loss needs to be explored in a prospective setting.

REFERENCES

- Ramnath N, Dilling TJ, Harris LJ, et al. Treatment of stage III non-small cell lung cancer: Diagnosis and management of lung cancer, 3rd ed: American College of Chest Physicians evidence-based clinical practice guidelines. Chest. 2013;143:e314S-340S.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. The Lancet Oncology. 2008;9:629-635.
- 3. Werner-Wasik M. Treatment-related esophagitis. Seminars in oncology. 2005;32:S60-66.
- Antoun S, Birdsell L, Sawyer MB, Venner P, Escudier B, Baracos VE. Association of skeletal muscle wasting with treatment with sorafenib in patients with advanced renal cell carcinoma: results from a placebo-controlled study. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2010;28:1054-1060.
- Auperin A, Le Pechoux C, Rolland E, et al. Metaanalysis of concomitant versus sequential radiochemotherapy in locally advanced nonsmall-cell lung cancer. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2010;28:2181-2190.
- Govaert SL, Troost EG, Schuurbiers OC, et al. Treatment outcome and toxicity of intensitymodulated (chemo) radiotherapy in stage III nonsmall cell lung cancer patients. Radiation oncology. 2012;7:150.
- Bar-Ad V, Leiby B, Witek M, et al. Treatment-related acute esophagitis for patients with locoregionally advanced non-small cell lung cancer treated with involved-field radiotherapy and concurrent chemotherapy. American journal of clinical oncology. 2014;37:433-437.
- Op den Kamp CM, De Ruysscher DK, van den Heuvel M, et al. Early body weight loss during concurrent chemo-radiotherapy for non-small cell lung cancer. Journal of cachexia, sarcopenia and muscle. 2014;5:127-137.
- Dingemans AM, Bootsma G, van Baardwijk A, et al. A phase I study of concurrent individualized, isotoxic accelerated radiotherapy and cisplatinvinorelbine-cetuximab in patients with stage III non-small-cell lung cancer. Journal of thoracic oncology : official publication of the International Association for the Study of Lung Cancer. 2014;9:710-716.

- van Baardwijk A, Wanders S, Boersma L, et al. Mature results of an individualized radiation dose prescription study based on normal tissue constraints in stages I to III non-small-cell lung cancer. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2010;28:1380-1386.
- van Elmpt W, De Ruysscher D, van der Salm A, et al. The PET-boost randomised phase II doseescalation trial in non-small cell lung cancer. Radiotherapy and oncology : journal of the European Society for Therapeutic Radiology and Oncology. 2012;104:67-71.
- Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus. The Lancet Oncology. 2011;12:489-495.
- Staal-van den Brekel AJ, Schols AM, ten Velde GP, Buurman WA, Wouters EF. Analysis of the energy balance in lung cancer patients. Cancer research. 1994;54:6430-6433.
- Topkan E, Parlak C, Selek U. Impact of weight change during the course of concurrent chemoradiation therapy on outcomes in stage IIIB non-small cell lung cancer patients: retrospective analysis of 425 patients. International journal of radiation oncology, biology, physics. 2013;87:697-704.
- Antoun S, Baracos VE, Birdsell L, Escudier B, Sawyer MB. Low body mass index and sarcopenia associated with dose-limiting toxicity of sorafenib in patients with renal cell carcinoma. Annals of oncology : official journal of the European Society for Medical Oncology / ESMO. 2010;21:1594-1598.
- Silver HJ, Dietrich MS, Murphy BA. Changes in body mass, energy balance, physical function, and inflammatory state in patients with locally advanced head and neck cancer treated with concurrent chemoradiation after lowdose induction chemotherapy. Head & neck. 2007;29:893-900.
- Ghadjar P, Hayoz S, Zimmermann F, et al. Impact of weight loss on survival after chemoradiation for locally advanced head and neck cancer: secondary results of a randomized phase III trial (SAKK 10/94). Radiation oncology. 2015;10:21.
- Bradley JD, leumwananonthachai N, Purdy JA, et al. Gross tumor volume, critical prognostic factor in patients treated with three-dimensional conformal radiation therapy for non-small-cell lung carcinoma. International journal of radiation

oncology, biology, physics. 2002;52:49-57.

- Werner-Wasik M, Swann RS, Bradley J, et al. Increasing tumor volume is predictive of poor overall and progression-free survival: secondary analysis of the Radiation Therapy Oncology Group 93-11 phase I-II radiation dose-escalation study in patients with inoperable non-small-cell lung cancer. International journal of radiation oncology, biology, physics. 2008;70:385-390.
- 20. Op den Kamp CM, Langen RC, Snepvangers FJ, et al. Nuclear transcription factor kappa B activation and protein turnover adaptations in skeletal muscle of patients with progressive stages of lung cancer cachexia. The American journal of clinical nutrition. 2013;98:738-748.

8

SUPPLEMENTALS



Supplemental figure 1. The course of mean (± standard error of the mean) weight loss during therapy of patients experiencing >5% early weight loss (solid line) compared to patients with ≤5% early weight loss (dotted line). Abbreviations: CT, chemotherapy; RT, radiotherapy.

Early weight loss during CCRT in NSCLC



9

The prognostic value of early onset CT derived loss of muscle and adipose tissue during chemotherapy in metastatic non-small cell lung cancer

Juliette H.R.J. Degens*, Karin J.C. Sanders*, Evelyn E.C. de Jong, Harry J.M. Groen, Egbert F. Smit, Joachim G. Aerts, Annemie M.W.J. Schols, Anne-Marie C. Dingemans. * Authors contributed equally

Lung Cancer. 2019;133:130-35.

ABSTRACT

Objectives

To evaluate the relationship between early changes in muscle and adipose tissue during chemotherapy and overall survival in stage IV non-small cell lung cancer (NSCLC).

Methods

In this post-hoc analysis of the first line NVALT12 trial (NCT01171170) in stage IV NSCLC, skeletal muscle, radiation attenuation, subcutaneous adipose tissue (SAT) and visceral adipose tissue (VAT) were assessed at the third lumbar level on computed tomography scans obtained before initiation of chemotherapy and shortly after administration of the second cycle. The contribution of changes in different body compartments to overall survival was assessed.

Results

Computed tomography scans of 111 patients were included. Analysis of body composition changes between the baseline and the follow-up scan, revealed that overall skeletal muscle cross sectional area (CSA), radiation attenuation and SAT CSA decreased respectively by-1.2 \pm 2.9 cm²/m² (p<0.001),-0.7 \pm 3.3 HU (p = 0.026) and-1.9 \pm 8.7 cm²/m² (p = 0.026), while no significant changes in VAT tissue were observed. Longitudinally, median overall survival was significantly shorter among patients losing skeletal muscle compared to patients with preserved skeletal muscle (9.4 versus 14.2 months; hazard ratio [HR] = 1.9, 95% confidence interval [CI]: 1.23-2.79, p=0.003). Multivariate analyses showed that proportional loss of muscle mass was associated with poor overall survival (HR = 0.949, 95% CI: 0.915-0.985, p=0.006) independent from important clinical prognostic factors including gender, age, World Health Organisation performance status and Charlson comorbidity index.

Conclusion

Early loss of skeletal muscle during first line chemotherapy is a poor prognostic factor in stage IV NSCLC patients. Future studies have to reveal whether early supportive intervention guided by initial computed tomography muscle response to chemotherapy can influence the wasting process and related mortality risk.

INTRODUCTION

Despite recent developments in cancer diagnostics and treatment modalities, mortality rates maintain high in non-small cell lung cancer (NSCLC) [1]. One of the factors contributing to high mortality rates is progressive unintentional weight loss of body weight and muscle mass (i.e. cachexia) [2].

Computed tomography (CT) has emerged as a promising tool in assessment of cancer cachexia. CT scans are routinely acquired in cancer patients for disease diagnosis, staging and treatment follow-up and thereby readily available from medical records to extract body composition data. Skeletal muscle- and adipose tissue depletion, both characteristics of cachexia [3], are clinically important as predictors of cancer outcomes [4-6]. Cachexia is frequently observed in lung cancer [7] and low skeletal muscle mass at presentation has been linked to functional deterioration, chemotherapy intolerance, and poor survival rates [8-10].

Muscle tissue is plastic, undergoing constant remodelling in response to anabolic and catabolic signals related to aging [11], exercise [12], malnutrition [13] inflammation [14] and drug interventions [15]. Evidence from experimental research suggests that chemotherapy might contribute to muscle wasting, via activation of nuclear factor kappa B and upregulation of myostatin [16]. Thereby, even weight stable patients with normal muscle mass before therapy initiation might lose muscle mass during the course of treatment [17].

To date, a few longitudinal studies addressed changes in muscle mass during anti-tumour therapy in NSCLC. In 35 stage IV NSCLC patients treated with palliative chemotherapy, muscle mass loss was observed in half of the study population over the duration of chemotherapy (approximately 9 weeks). However, the authors found no significant survival effect in univariate analysis [18]. Another small study reported muscle depletion in 30 stage III and IV NSCLC patients, which was accompanied by physical functional decline. The prognostic effect of muscle depletion was not examined [19].

Next to skeletal muscle depletion, cachexia might also effect other tissues, including adipose tissue. Although adipose tissue mass has been emerging as a prognostic factor, results to date are confusing. Patients with low baseline subcutaneous adipose tissue in gastrointestinal- and lung cancer [20] or low visceral adipose tissue in renal cell carcinoma exhibited lower survival rates [21]. In contrast, high adipose infiltration in the skeletal muscle as a measure of muscle quality, reflected on CT analysis by low skeletal muscle radiodensity, was associated with shorter overall survival and shorter disease free survival in patients with various other cancer types [5,22-25].

The prognostic value of early onset changes in muscle- and adipose tissue compartments in therapy naive NSCLC patients during chemotherapy is lacking. Particularly, in this patient population known for poor survival rates, timely information on body compartment changes could be of clinical relevance for patient guidance and treatment decision making. Therefore, the primary goal of our study is to evaluate early body composition changes in relation to survival in therapy naive metastatic NSCLC patients undergoing chemotherapy.

METHODS

Study population and study design

This study was a post-hoc analysis of a multicentre randomized phase II trial investigating the effect on survival of nitroglycerin patches added to paclitaxel-carboplatin-bevacizumab in 223 therapy naive patients with stage IV non-squamous NSCLC (NCT01171170). The full and detailed methodology and results of this study have been published previously [26].

In short, patients were randomized between paclitaxel-carboplatin-bevacizumab with (experimental arm) and without nitroglycerin patches (control arm). Adding nitroglycerin to first-line carboplatin-paclitaxel-bevacizumab did not improve progression-free survival and overall survival (OS) in patients with stage IV non-squamous NSCLC. Exploratory endpoint of the study was to assess whether nitroglycerin was related with an early decrease in 18F-fluorodeoxyglucose (18F-FDG) uptake. According to the protocol, a second 18F-FDG positron emission tomography (PET) scan combined with CT scan was assessed for patients with a baseline PET/CT after the second cycle of chemotherapy [27], with median follow-up time of 23 (17-50) days between follow-up scan and initiation of therapy. Specific patient characteristics (age, gender, World Health Organisation performance status [WHO PS], smoking status, histology, body mass index [BMI] and Charlson comorbidity index) were selected from patient records [28]. Patients were selected from this study because the population is a well-defined randomized cohort in which all patients have received a standardized chemotherapeutic regime. Next to this, all patients were assessed by a physician with follow-up scans at predetermined time points.

Image analysis

In this post-hoc analysis body composition was analysed on low dose CT scan by assessment of the cross-sectional area (CSA) at the third lumbar level by two assessors. CSA of skeletal muscle, subcutaneous adipose tissue and visceral adipose tissue were analysed with Slice-O-Matic software v5.0 (Tomovision, Montreal, Canada). One image in each scan was selected. During anatomical land marking, the first image at the third lumbar level with both vertebral transverse processes clearly visible was used for analysis. CSA of these structures were quantified based on pre-established thresholds of Hounsfield units (skeletal muscle-29 to 150, subcutaneous adipose tissue-190 to-30, and visceral adipose tissue-150 to-50). Boundaries were corrected manually when necessary.

Baseline CSA was normalised for height in meters squared and reported in cm²/m². Changes in CSA between CT scans were expressed as a percentage. We found a mean coefficient of variation between observers of 1.3% for skeletal muscle area, subcutaneous- and visceral adipose tissue in a random sample of 15 patients, which is in line with a variation of 0-2% in other studies [7,29-31]. Therefore, a measurement error of 1.3% was adopted. Changes of equal or larger than-1.3% were considered as 'loss of tissue', while changes of smaller than-1.3% were considered for the total skeletal muscle radiation attenuation was assessed as the average Hounsfield units of the total skeletal muscle area within the range -29 to 150 (i.e. excluding intramuscular adipose tissue).

Study endpoints and statistical analyses

Patients were included if CT scans both at baseline and follow-up were available, were assessed within 3 weeks after the second chemotherapy and contained images of the third lumbar level. Descriptive statistics of demographic and clinical variables were obtained. Means (±SD) were provided for continuous normally distributed variables, median (range) for continuous not-normally distributed variables and percentages were shown for categorical variables. Comparisons within groups were performed with paired t-test and between groups with an independent t-test.

The primary endpoint of this study was OS. OS was defined as the interval from randomization to death from any cause. The probabilities of OS were calculated using the Kaplan-Meier method. Survival curves of patients with and without skeletal muscle loss were tested for significance using the log-rank test. To assess the contribution of different body compartments to OS, multivariate Cox regression analysis was performed with body composition changes, gender, age, Charlson comorbidity index, and BMI as independent variables. Variables were tested for interactions. Proportional hazard assumption was tested using visual inspection of log-minus-log survival plots. All analyses were performed using SPSS statistical software (SPSS Statistics for Windows, Version 24.0, IBM, Armonk, NY). Results with two-sided exact p values (<0.05) were considered statistically significant.

RESULTS

Patients and characteristics

CT scans from 111 of the 223 enrolled patients were eligible (figure 1). Baseline patient characteristics are shown in table 1.



Figure 1. Flow chart study population.

Table 1. Patient characteristics (N=111).

Experimental arm/control arm, N	63 / 48
Male/female, N	60 / 51
Age, mean (range)	61 (39- 79)
WHO performance status, N (%) 0 / 1 / 2	50 (45.0) / 46 (41.1) / 3 (2.7)
Comorbidity, N (%) COPD Cardiovascular disease Diabetes mellitus Gastrointestinal disease >1 comorbidity	16 (14.4) 41 (36.9) 12 (10.8) 15 (13.5) 50 (45%)
Smoker, N (%) Current / ex / never	48 (43.2) / 49 (44.1) / 14 (12.6)
Histology, N (%) Adenocarcinoma Large cell Other	96 (86.5) 6 (5.4) 7 (6.4)
Body weight, kg	74.1 ± 14.7
BMI, kg/m ²	25.0 ± 4.3

WHO performance status missing N=12, histology missing N=2.

Table 2 shows baseline measurements and body composition changes parameters of the whole group. In total 65 (58.5%) lost muscle mass during the course of chemotherapy. Analysis of body composition changes between baseline and follow-up scan revealed that muscle CSA, radiation attenuation and subcutaneous adipose tissue CSA decreased respectively by-1.2 \pm 2.9 cm²/m² (p<0.001),-0.7 \pm 3.3 HU (p = 0.026) and-1.9 \pm 8.7 cm²/m² (p = 0.026), while no significant changes in visceral adipose tissue were observed.

Table 2. Body composition at baseline and change during therapy.

	Baseline	Change, absolute	Change, percentage	p value
Skeletal muscle CSA, cm ² /m ²	44.4 ±6.9	-1.2 ± 2.9	-2.7 ± 6.6	<0.001
Radiation attenuation (housefield units)	31.5 ± 7.2	-0.7 ± 3.3	-1.9 ± 12.2	0.026
Subcutaneous adipose tissue CSA, cm^2/m^2	54.5 ± 31.8	-1.9 ± 8.7	-2.1 ± 20.2	0.026
Visceral adipose tissue CSA, cm ² /m ²	34.3 ± 22.5	0.6 ± 8.0	2.1 ± 29.3	0.438

Data are represented as mean ± SD. Abbreviations: CSA, cross-sectional area.

Survival

Median OS (95% confidence interval [CI]) was similar between subjects with normal versus low skeletal muscle index at baseline (12.2 months, 95% CI: 9.8-14.7 versus 10.3 months, 95% CI: 7.8-12.7, hazard ratio [HR] 1.1, 95% CI: 0.8-1.7, p = 0.496 [data not shown]).



Figure 2. Kaplan Meier overall survival curve for patients with loss of muscle mass compared to patients without loss of muscle mass.

Longitudinally, median OS was significantly shorter among patients losing skeletal muscle mass compared to patients with preserved skeletal muscle mass (9.4 months, 95% CI: 7.1-11.6 versus 14.2 months, 95% CI: 11.3-17.1, HR 1.9, 95% CI: 1.23-2.79, p = 0.003) (figure 2). Multivariate analyses showed that proportional loss of muscle mass (HR 0.949, 95% CI: 0.915-0.985, p = 0.006) was associated with poor OS independent from important clinical prognostic factors including gender, age, WHO PS and Charlson comorbidity index (table 3).

Variable	HR (95% CI)	p value
Delta muscle mass (%)	0.949 (0.915-0.985)	0.006
Age (years)	1.008 (0.978-1.039)	0.593
Gender	1.593 (0.915-2.772)	0.100
WHO performance status	1.084 (0.847-1.387)	0.520
Charlson comorbidity index	1.132 (0.960-1.335)	0.139

Table 3. Multivariate analysis for predictors of overall survival.

Adipose tissue compartments

On multivariate analysis, changes in skeletal muscle mass were associated with OS (table 3). We therefore stratified patients in two groups; 'maintenance of skeletal muscle mass' and 'loss of skeletal muscle mass'. At baseline, there were no differences between groups regarding weight and body composition. Furthermore, groups did not differ in tumour progression (data not shown). Muscle loss was accompanied by a generalized loss of tissue as both subcutaneous fat and visceral fat decreased. Furthermore, radiation attenuation decreased, which could reflect gain in intramuscular fat (table 4) [32].

DISCUSSION

In the current study, we identified body composition changes after two cycles of chemotherapy treatment initiation in stage IV NSCLC patients. Skeletal muscle CSA, muscle radiation attenuation and to lesser extent subcutaneous adipose tissue decreased throughout treatment indicating a rapid loss in muscle mass and quality. Sixty-nine percent of the patients lost muscle mass and exhibited worse survival rates compared to those with preserved muscle mass.

Next to the loss of muscle mass, a rapid decline in skeletal muscle radiation attenuation was observed during the course of treatment. Reduced radiation attenuation is believed to reflect fat infiltration (myosteatosis) in the muscle [33]. Myosteatosis in the muscle has indeed been observed in muscle biopsies from cancer patients and was more aggravated in those with weight loss [34]. The phenomenon of myosteatosis is shown in several patient groups including both obese patients [35] and patients with low BMI. This suggests that high muscle fat infiltration is not merely a result of disturbed energy balance but could be driven by disturbances in muscle oxidative metabolism.

Table 4. Characterization of the	he overall wasti	ng pattern in gr	oups stratified I	by muscle resp	onse.			
	Maintenance	e of muscle mas	ss (n=46)	Loss of	muscle mass (r	1=65)		
	Baseline	Change, %	Within group p value	Baseline	Change, %	Within group p value	Between group baseline p value	Between group change p value
Skeletal muscle, cm²/m²	43.8 <u>±</u> 6.9	2.9 ± 4.0	<0.001	44.9 ± 7.0	-6.6±5.1	<0.001	0.392	<0.001
Radiation attenuation (HU)	30.6 ± 7.6	-0.3 ± 11.9	0.600	32.1±6.9	-3.0 ± 12.4	0.015	0.243	0.233
Subcutaneous adipose tissue, cm²/m²	58.3 ± 35.2	7.9 ± 18.5	0.043	51.9 ± 29.1	-9.3 ± 18.3	<0.001	0.291	<0.001
Visceral adipose tissue, cm²/m²	36.4 ± 24.6	11.0 ± 23.4	0.004	32.8 ± 21.1	-4.1 ± 31.5	0.029	0.445	0.006
Weight, kg	74.7 ± 15.7	0.7 ± 2.4	0.068	73.7 ± 14.1	-1.7 ± 3.2	<0.001	0.811	<0.001
Data are represented as mean	ד ± SD. Abbrevi∂	ations: HU, hour	nsefield units.					

A recent study, comparing changes in skeletal muscle during treatment with chemotherapy or targeted therapy (EGFR or ALK tyrosine kinase inhibitor) showed significant differences in the loss of skeletal muscle in advanced NSCLC patient [36]. Skeletal muscle was decreased in both groups, but the loss of muscle mass was significantly lower in the targeted therapy group compared to chemotherapy.

In this study, all patients were treated with the same chemotherapeutic regimen (carboplatinum, paclitaxel, bevacizumab). Therefore, it is interesting to speculate about the influence of anti-cancer drugs and toxicity on skeletal muscle changes in advanced NSCLC. From experimental research it is known that chemotherapy can directly affect regulation of muscle maintenance, possibly mediated by nuclear factor kappa B activation [37]. Cisplatin is known to induce nuclear factor kappa B, which triggers muscle wasting. In addition, also the tumour itself is able to increase nuclear factor kappa B. Thereby, muscle wasting might be both tumour- and chemotherapy induced via common pathways mediated by nuclear factor kappa B. In this study, nineteen (17%) of the patients experienced an adverse event graded with a CTCAE of \geq 3. There was no significant difference in toxicity between the group "muscle loss" and "muscle maintenance". The underlying mechanism why some patients experience chemotherapy-induced muscle wasting, while others maintain muscle mass is unknown.

Muscle and fat loss during chemotherapy

The added value of the current study is that also in patients receiving standardized chemotherapy, clinically relevant inter-individual responses in body composition were observed (46 patients maintained muscle mass and had a stable body weight) which may partly be reversible by early supportive interventions. Importantly in this context and in contrast to some previous studies [4,6,9,10,25], low muscle mass and attenuation at baseline were not prognostic for overall survival.

The main strength of our study comes from the well-defined randomized patient cohort, receiving a standardized chemotherapeutic regime [26]. However, this study is not without limitations. All patients were treated with the same chemotherapeutic agents however in 63 of 111 patient nitroglycerin patches were added to the therapy. No difference in OS between the two groups was found [26]. The follow-up CT scans were executed according to the protocol on predefined times, CT scans were performed in different medical centres, whereby minor variation in scanning procedure between hospitals and quality of the images cannot be ruled out. Additionally, data on pre-treatment weight loss were unavailable.

To conclude, this study shows prognostic value of body composition changes after two cycles of chemotherapy treatment in stage IV NSCLC patients. Therefore, CT derived assessment of body composition may provide an additional tool for the treating physician to judge disease severity and related prognosis. Future studies have to reveal whether early supportive intervention, for example with intensive nutritional support, guided by initial CT muscle response to chemotherapy can influence the systemic consequences of NSCLC (treatment) and related elevated mortality risk.

REFERENCES

- Global Health Estimates 2016: Deaths by Cause, Age, Sex, by Country and by Region, 2000-2016. Geneva. World Health Organization. 2018.
- Martin L, Senesse P, Gioulbasanis I, et al. Diagnostic criteria for the classification of cancer-associated weight loss. J Clin Oncol. 2015;33:90-99.
- Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus. Lancet Oncol. 2011;12:489-495.
- Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle. 2017;8:759-767.
- Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapyinduced haematological toxicity in advanced nonsmall cell lung cancer. Lung Cancer. 2015;90:85-91.
- Cho KM, Park H, Oh DY, et al. Skeletal muscle depletion predicts survival of patients with advanced biliary tract cancer undergoing palliative chemotherapy. Oncotarget. 2017;8:79441-79452.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008;9:629-635.
- Suzuki Y, Okamoto T, Fujishita T, et al. Clinical implications of sarcopenia in patients undergoing complete resection for early non-small cell lung cancer. Lung Cancer. 2016;101:92-97.
- Martin L, Birdsell L, Macdonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol. 2013;31:1539-1547.
- Kim EY, Kim YS, Park I, Ahn HK, Cho EK, Jeong YM. Prognostic Significance of CT-Determined Sarcopenia in Patients with Small-Cell Lung Cancer. J Thorac Oncol. 2015;10:1795-1799.
- Frontera WR, Hughes VA, Fielding RA, Fiatarone MA, Evans WJ, Roubenoff R. Aging of skeletal muscle: a 12-yr longitudinal study. J Appl Physiol (1985). 2000;88:1321-1326.
- Hawley JA, Hargreaves M, Joyner MJ, Zierath JR. Integrative biology of exercise. Cell. 2014;159:738-749.
- Matsakas A, Patel K. Skeletal muscle fibre plasticity in response to selected environmental and physiological stimuli. Histol Histopathol.

2009;24:611-629.

- Remels AH, Gosker HR, van der Velden J, Langen RC, Schols AM. Systemic inflammation and skeletal muscle dysfunction in chronic obstructive pulmonary disease: state of the art and novel insights in regulation of muscle plasticity. Clin Chest Med. 2007;28:537-552, vi.
- Damrauer JS, Stadler ME, Acharyya S, Baldwin AS, Couch ME, Guttridge DC. Chemotherapy-induced muscle wasting: association with NF-kappaB and cancer cachexia. Eur J Transl Myol. 2018;28:7590.
- Costelli P, Muscaritoli M, Bonetto A, et al. Muscle myostatin signalling is enhanced in experimental cancer cachexia. Eur J Clin Invest. 2008;38:531-538.
- Eriksson S, Nilsson JH, Strandberg Holka P, Eberhard J, Keussen I, Sturesson C. The impact of neoadjuvant chemotherapy on skeletal muscle depletion and preoperative sarcopenia in patients with resectable colorectal liver metastases. HPB (Oxford). 2017;19:331-337.
- Stene GB, Helbostad JL, Amundsen T, et al. Changes in skeletal muscle mass during palliative chemotherapy in patients with advanced lung cancer. Acta Oncol. 2015;54:340-348.
- Naito T, Okayama T, Aoyama T, et al. Skeletal muscle depletion during chemotherapy has a large impact on physical function in elderly Japanese patients with advanced non-small-cell lung cancer. BMC Cancer. 2017;17:571.
- Ebadi M, Martin L, Ghosh S, et al. Subcutaneous adiposity is an independent predictor of mortality in cancer patients. British journal of cancer. 2017;117:148-155.
- Mizuno R, Miyajima A, Hibi T, et al. Impact of baseline visceral fat accumulation on prognosis in patients with metastatic renal cell carcinoma treated with systemic therapy. Medical oncology. 2017;34:47.
- Antoun S, Lanoy E, Iacovelli R, et al. Skeletal muscle density predicts prognosis in patients with metastatic renal cell carcinoma treated with targeted therapies. Cancer. 2013;119:3377-3384.
- Sabel MS, Lee J, Cai S, Englesbe MJ, Holcombe S, Wang S. Sarcopenia as a prognostic factor among patients with stage III melanoma. Annals of surgical oncology. 2011;18:3579-3585.
- Rollins KE, Tewari N, Ackner A, et al. The impact of sarcopenia and myosteatosis on outcomes of unresectable pancreatic cancer

or distal cholangiocarcinoma. Clinical nutrition. 2016;35:1103-1109.

- Kroenke CH, Prado CM, Meyerhardt JA, et al. Muscle radiodensity and mortality in patients with colorectal cancer. Cancer. 2018;124:3008-3015.
- Dingemans AM, Groen HJ, Herder GJ, et al. A randomized phase II study comparing paclitaxelcarboplatin-bevacizumab with or without nitroglycerin patches in patients with stage IV nonsquamous nonsmall-cell lung cancer: NVALT12 (NCT01171170)dagger. Ann Oncol. 2015;26:2286-2293.
- de Jong EE, van Elmpt W, Leijenaar RT, et al. [18F]FDG PET/CT-based response assessment of stage IV non-small cell lung cancer treated with paclitaxel-carboplatin-bevacizumab with or without nitroglycerin patches. Eur J Nucl Med Mol Imaging. 2017;44:8-16.
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987;40:373-383.
- Mourtzakis M, Prado CM, Lieffers JR, Reiman T, McCargar LJ, Baracos VE. A practical and precise approach to quantification of body composition in cancer patients using computed tomography images acquired during routine care. Appl Physiol Nutr Metab. 2008;33:997-1006.
- Irving BA, Weltman JY, Brock DW, Davis CK, Gaesser GA, Weltman A. NIH ImageJ and Slice-O-Matic computed tomography imaging software to quantify soft tissue. Obesity (Silver Spring). 2007;15:370-376.
- Gallagher D, Kuznia P, Heshka S, et al. Adipose tissue in muscle: a novel depot similar in size to visceral adipose tissue. Am J Clin Nutr. 2005;81:903-910.
- Schrauwen-Hinderling VB, Hesselink MK, Schrauwen P, Kooi ME. Intramyocellular lipid content in human skeletal muscle. Obesity (Silver Spring). 2006;14:357-367.
- Goodpaster BH, Kelley DE, Thaete FL, He J, Ross R. Skeletal muscle attenuation determined by computed tomography is associated with skeletal muscle lipid content. J Appl Physiol (1985). 2000;89:104-110.
- Stephens NA, Skipworth RJ, Macdonald AJ, Greig CA, Ross JA, Fearon KC. Intramyocellular lipid droplets increase with progression of cachexia in cancer patients. J Cachexia Sarcopenia Muscle. 2011;2:111-117.

- 35. van Loon LJ, Koopman R, Manders R, van der Weegen W, van Kranenburg GP, Keizer HA. Intramyocellular lipid content in type 2 diabetes patients compared with overweight sedentary men and highly trained endurance athletes. Am J Physiol Endocrinol Metab. 2004;287:E558-565.
- 36. Kakinuma K, Tsuruoka H, Morikawa K, et al. Differences in skeletal muscle loss caused by cytotoxic chemotherapy and molecular targeted therapy in patients with advanced non-small cell lung cancer. Thorac Cancer. 2018;9:99-104.
- Li H, Malhotra S, Kumar A. Nuclear factor-kappa B signaling in skeletal muscle atrophy. J Mol Med (Berl). 2008;86:1113-1126.

Muscle and fat loss during chemotherapy



10

Can radiomics help to predict skeletal muscle response to chemotherapy in stage IV non-small cell lung cancer?

Evelyn E.C. de Jong, Karin J.C. Sanders, Timo M. Deist, Wouter van Elmpt, Arthur Jochems, Janita E. van Timmeren, Ralph T.H. Leijenaar, Juliette H.R.J. Degens, Annemie M.W.J. Schols, Anne-Marie C. Dingemans, Philippe Lambin.

European Journal of Cancer. 2019;120:107-13.

ABSTRACT

Background

Muscle depletion negatively impacts treatment efficacy and survival rates in cancer. Prevention and timely treatment of muscle loss requires prediction of patients at risk. We aimed to investigate the potential of skeletal muscle radiomic features to predict future muscle loss.

Methods

A total of 116 patients with stage IV non-small cell lung cancer included in a randomised controlled trial (NCT01171170) studying the effect of nitroglycerin added to paclitaxelcarboplatin-bevacizumab were enrolled. In this post hoc analysis, muscle cross-sectional area and radiomic features were extracted from computed tomography images obtained before initiation of chemotherapy and shortly after administration of the second cycle. For internal cross-validation, the cohort was randomly split in a training set and validation set 100 times. We used least absolute shrinkage and selection operator method to select features that were most significantly associated with muscle loss and an area under the curve (AUC) for model performance.

Results

Sixty-nine patients (59%) exhibited loss of skeletal muscle. One hundred ninety-three features were used to construct a prediction model for muscle loss. The average AUC was 0.49 (95% confidence interval: 0.36-0.62). Differences in intensity and texture radiomic features over time were seen between patients with and without muscle loss.

Conclusions

The present study shows that skeletal muscle radiomics did not predict future muscle loss during chemotherapy in non-small cell lung cancer. Differences in radiomic features over time might reflect myosteatosis. Future imaging analysis combined with muscle tissue analysis in patients and in experimental models is needed to unravel the biological processes linked to the radiomic features.

INTRODUCTION

Cachexia is a frequently observed phenomenon of skeletal muscle and adipose tissue depletion among patients with non-small cell lung cancer (NSCLC) [1,2]. The progressive loss of muscle has a devastating impact on the quality of life [3] and survival rates in patients with NSCLC [2,4-6]. Although both muscle and fat become depleted, there is evidence that body fat is lost more rapidly than muscle [7,8]. Cancer may therefore shift lipid metabolism to a catabolic state, which in turn may affect skeletal muscle. In cancer, depletion of subcutaneous fat is driven by increased lipolysis [9,10]. Experimental research has shown that lipolysis generates fatty acids which are able to transport into myocytes and stimulate protein degradation [11]. Indeed, skeletal muscle of patients with cancer contained more intramyocellular fat than ageand gender-matched controls [12]. However, there are currently no (non-invasive) biomarkers available to predict muscle loss.

Radiomics is a method that quantitatively extracts features including shape, size, intensity and texture that are related to pathophysiology from standard-of-care medical images [13-17]. Until now, radiomics has mainly been applied to extract tumour features in oncologic patients to visualise tumour heterogeneity [17] and to assess prognosis [18-20]. Besides quantitative evaluation of Hounsfield units, radiomics also explores patterns. Given the quantitative and qualitative differences in skeletal muscles of cachectic patients, radiomics might be helpful to predict future muscle loss. Therefore, the primary goal of this exploratory study is to investigate whether baseline skeletal muscle radiomic features are different between patients who develop muscle loss and those who maintain their muscle mass after chemotherapy. We furthermore investigated longitudinally if muscle loss is associated with changes in muscle radiomic features.

METHODS

Patient cohort

Computed tomography (CT) scans derived from the multicentre randomised phase II trial (NVALT12 trial, NCT01171170) were investigated. In this trial the effect of nitroglycerin added to paclitaxel-carboplatin-bevacizumab on progression-free survival in chemotherapy naive stage IV non-squamous NSCLC was investigated. The methodology and results of this trial have been published previously [21].

Image analysis

CT scans made at baseline and after the second cycle of chemotherapy, as part of a secondary end point, were used [22]. To evaluate whether or not patients lost skeletal muscle, crosssectional measurements of skeletal muscle areas were made on transverse images at the third lumbar level using Slice-O-Matic software version 5.0 (Tomovision, Montreal, Canada). One slice at the third lumbar level in each scan was selected for each patient. During anatomical land marking, the first image at the third lumbar level with both vertebral transverse processes clearly visible was used for analysis. Skeletal muscle cross-sectional area was quantified on the basis of pre-established thresholds of Hounsfield units (-29 to 150). Boundaries were corrected manually when necessary. It is of note, that this delineation excludes intramuscular fat. Changes in muscle cross-sectional areas between CT scans were expressed as a percentage. A measurement error of 1.3% was adopted, based on previous reported literature [2,4]. Changes greater than or equal to -1.3% were considered as 'loss of skeletal muscle', while changes less than -1.3% were considered 'maintenance of skeletal muscle'. In addition, the mean Hounsfield units of the muscle CSA were assessed, as a measure for muscle fat deposits. Low values reflect increased muscle fat.

Then, to evaluate radiomic features, skeletal muscle cross-sectional area was delineated at the third lumbar level using the same thresholds of Hounsfield units as described previously. Now, it was extended one slice in the cranial direction and one slice in the caudal direction using Mirada software (Mirada Medical, Oxford, UK), to be able to calculate three-dimensional image features. Segmentation for radiomics was a semi-automatic process which was manually adjusted if needed. Image features were calculated on both baseline and follow-up scans, using an adapted version of Computational Environment for Radiotherapy Research extended with in-house developed radiomic image analysis software (Matlab 2014a; The Mathworks, Natick, MA). Before the extraction of features, a grey-level discretisation using a bin width of 25 Hounsfield units was applied. To minimise the possible effect of the variation in image parameters, all scans were resampled to a voxel size of 1 x 1 x 3 mm³ using a cubic interpolation as recommended in the study by Larue et al. [23]. Although collection of the CT scans was predefined in the clinical trial, the analysis of muscle mass and radiomics was not part of the pre-registered outcomes of the original trial. Because of this well-defined randomised patient cohort, with CT scans executed according to study protocol-predefined time points, this might be an appropriate data set to explore our hypothesis.

Radiomic feature selection and statistics

Intensity and texture features were analysed. Shape and size features were excluded because the volumes of interest has been segmented manually, which might influence the outcome of these feature categories. In addition, a three-dimensional wavelet transformation was applied to the CT scan to create filtered, next to the unfiltered intensity and texture features. Features without a range (i.e. features with an exact similar value in all patients), which were not able to discriminate patients, were excluded.

Spearman's correlation coefficient (ρ) was used to assess the correlation between all texture and intensity radiomic features. Of each feature pair with ρ >0.85, the feature that was strongest correlated to all other features was excluded. This process was repeated until no feature pair with all ρ >0.85 was remaining. To calculate to which extent the variation among the radiomic features on baseline scans is explained by muscle loss, a logistic Least Absolute Shrinkage and Selection Operator (LASSO) regression model adopting a 100-fold Monte Carlo cross-validation in Matlab 2017b was applied (The Mathworks, Natick, MA).

The cohort was randomly split into a training set (approximately two-third) and validation set (approximately one-third). Patients were randomised such that the ratio between patients with and without skeletal muscle loss was similar in each group. Features in the training and validation sets were standardised by subtracting the respective mean feature value in the training set and dividing by the feature standard deviation in the training set. The logistic LASSO model was used to reduce the number of features and estimate regression coefficients for the remaining features. The model-intrinsic parameter λ was estimated using an internal five-fold cross-validation on the training set. The out-of-sample area under the curve (AUC) of the Receiver operating characteristic curve (ROC) was computed on the validation set to assess the prognostic power. This process was repeated 100 times, each time with a different randomisation of the patients into training and validation sets, and the average AUC over the hundred different models was calculated. This analysis was carried out for the radiomic texture and intensity features of the baseline and for the absolute difference in feature values between the baseline scan and the follow-up scan (delta features). For all radiomic features that were selected at least once in the LASSO feature selection procedure the difference in feature value between patients with and without muscle loss was compared by plotting a heatmap. For the heatmap all radiomic features were normalised to have values between 0 and 1. A hierarchical clustering was applied on the same heatmap to identify clusters of patients with different radiomic texture and intensity values.

RESULTS

Patients and characteristics

In total, 223 patients were enrolled in the randomised controlled trial. One hundred three patients were excluded because of unavailability of one or both CT scans, two patients were excluded because L3 was not evaluable, one patient was excluded because of lacking overall survival (OS) data and one was excluded for insufficient quality of the scans. After exclusion, CT scans from 116 patients were eligible. The mean age was 61 years, and 64 patients (55%) were male; survival rates of the included group were comparable with the whole study group of 223 patients.

Muscle maintenance and muscle loss

Delineations of skeletal muscle cross-sectional area made using Slice-O-Matic for cross-sectional area and using Mirada for extraction of radiomic features are shown in figure 1. Analysis of skeletal muscle cross-sectional area at baseline and during follow-up revealed that in the whole

cohort, skeletal muscle decreased with mean (\pm standard deviation) 2.9 \pm 6.7%. Of those, 69 patients (59%) exhibited loss of skeletal muscle.



Figure 1. Skeletal muscle area on transverse computed tomography images at the third lumbar level, using (a) Slice-O-Matic for evaluation of cross-sectional area and (b) Mirada software for extraction of radiomic features.

Radiomic features

A total of 1298 radiomic features were extracted. For analysis of baseline radiomic features, those who had no range (n=9) were excluded. After removing the redundant features using the spearman correlation method, 193 radiomic features, 11 unfiltered and 182 filtered, could be used for the analysis.

Figure 2 shows a heatmap in which radiomic features that are selected at least once in the LASSO models are plotted against the patients ranked in descending order of the decrease in muscle mass. The upper half of the graph shows the patients with muscle loss, and the lower half of the graph the patients with a stable muscle mass. No difference in radiomic features is seen between those with maintenance of muscle and those with muscle loss. The average AUC over 100 repetitions for radiomic features of the baseline scan, with muscle loss as outcome, is 0.49 (95% CI: 0.36-0.62). In addition, no differences were observed in muscle mass changes between patients treated with and without nitroglycerin. A list with the selected radiomic features in alphabetic order can be found in the supplementals.

We also analysed the delta features (differences in feature values between the baseline scan and the follow-up scan) with muscle loss as outcome. The average AUC was 0.68 (95% CI: 0.51-0.84). In this model the grey-level co-occurrence matrix (GLCM) was most often selected (data not shown).



Figure 2. Heatmap of the normalised selected delta radiomics features in alphabetic order (list of features is reported in supplementals). Patients are ranked according to the descending difference in muscle mass; upper half are the patients with muscle depletion, and lower half patients with stable muscle mass.

DISCUSSION

To the best of our knowledge, this is the first exploratory study evaluating the potential of skeletal muscle radiomics to predict skeletal muscle loss. Some cross-sectional studies show that baseline low CT-derived muscle mass is an important prognostic factor for OS [4,24], which contradicts to other studies [25,26]. The lack of properly validated and population-specific cut-off values for CT-derived low muscle mass may explain this discrepancy. We therefore were interested to know if radiomic features of baseline skeletal muscle are prognostic for longitudinal muscle loss. In this study, baseline radiomic features had no discriminatory value with regard to longitudinal skeletal muscle changes. However, longitudinal differences in radiomic features were seen between those who lost muscle and those who maintained muscle mass.

The most distinct delta radiomic feature was a feature in the GLCM category. To calculate the GLCM, each pixel in an image is assigned a numerical value depending on the combination of grey-level intensity values in two neighbouring pixels. Second-order statistics calculate mathematical algorithms to derive textural homogeneity, contrast, variance etc. [27]. GLCM features depend on the grey-level pattern, which might implicate that patients with muscle loss develop more muscle fat deposits, leading to a more heterogeneous grey-level intensity pattern (as low grey levels reflect increased intramuscular fat). Indeed analysis of muscle biopsies demonstrated that compared with controls, patients with cancer exhibited increased numbers of lipid droplets in skeletal muscle. Moreover, the amount of lipid droplets increased with progression of weight loss [28]. Muscle fat depots assessed by muscle radiation attenuation on CT are associated with poor survival in cancer [4]. A phantom study showed concordance between radiation attenuation and muscle lipid content [29], therefore, reduced muscle radiation attenuation is believed to reflect fat infiltration. However, it is unclear if this radiation attenuation indicates intramyocellular or extramyocellular lipids. Therefore, the aetiology and prognostic significance of muscle lipids in cachexia progression is object for further research.

While the strength of our study comes from the well-defined randomised patient cohort, with CT scans executed according to study protocol-predefined time points, there are some limitations. Pre-treatment changes in muscle mass are unknown. Patients currently identified as having 'muscle maintenance' could have exhibited muscle loss before the first CT scan, which may influence radiomic features. The NVALT12 trial is a multicentre study. Consequently, the CT scans included in this analysis are performed in different hospitals on different CT scanners. Although all scans were low-dose CT scans, there might be differences in scan acquisition and the use of contrast. A phantom study showed that slice thickness variability influenced radiomic features, however this variability can be reduced by resampling to a standardised voxel size [23]. While it is known that different exposures do not influence the radiomic features values [23], the impact of use of contrast is unknown.

In conclusion, the present study shows that baseline skeletal muscle radiomics did not predict future muscle loss in patients with metastatic NSCLC. Nevertheless, longitudinal differences in intensity and texture muscle radiomic features are detected, which slightly differ between patients with muscle loss and patients with preserved muscle. Future research in experimental models and human radiomics combined with muscle tissue analysis is required to unravel the biological processes linked to the radiomic features.

REFERENCES

- Fearon K, Strasser F, Anker SD, et al. Definition and classification of cancer cachexia: an international consensus. Lancet Oncol. 2011;12:489-495.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008;9:629-635.
- Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle. 2017;8:759-767.
- Martin L, Birdsell L, Macdonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol. 2013;31:1539-1547.
- Stene GB, Helbostad JL, Amundsen T, et al. Changes in skeletal muscle mass during palliative chemotherapy in patients with advanced lung cancer. Acta Oncol. 2015;54:340-348.
- Degens J, Sanders K, De Jong E, Schols A, Dingemans A. P2.01-048 Early Changes in Body Composition in Metastatic Non-Small Cell Lung Cancer (NSCLC) Are Predictive for Poor Overall Survival. Journal of Thoracic Oncology. 2017;12:S2087-S2088.
- Fouladiun M, Korner U, Bosaeus I, Daneryd P, Hyltander A, Lundholm KG. Body composition and time course changes in regional distribution of fat and lean tissue in unselected cancer patients on palliative care--correlations with food intake, metabolism, exercise capacity, and hormones. Cancer. 2005;103:2189-2198.
- Agustsson T, Wikrantz P, Ryden M, Brismar T, Isaksson B. Adipose tissue volume is decreased in recently diagnosed cancer patients with cachexia. Nutrition. 2012;28:851-855.
- Ryden M, Agustsson T, Laurencikiene J, et al. Lipolysis--not inflammation, cell death, or lipogenesis--is involved in adipose tissue loss in cancer cachexia. Cancer. 2008;113:1695-1704.
- Agustsson T, Ryden M, Hoffstedt J, et al. Mechanism of increased lipolysis in cancer cachexia. Cancer Res. 2007;67:5531-5537.
- Zhou Q, Du J, Hu Z, Walsh K, Wang XH. Evidence for adipose-muscle cross talk: opposing regulation of muscle proteolysis by adiponectin and Fatty acids. Endocrinology. 2007;148:5696-5705.
- 12. Weber MA, Krakowski-Roosen H, Schroder L, et

al. Morphology, metabolism, microcirculation, and strength of skeletal muscles in cancer-related cachexia. Acta Oncol. 2009;48:116-124.

- Lambin P, van Stiphout RG, Starmans MH, et al. Predicting outcomes in radiation oncology-multifactorial decision support systems. Nat Rev Clin Oncol. 2013;10:27-40.
- Lambin P, Rios-Velazquez E, Leijenaar R, et al. Radiomics: extracting more information from medical images using advanced feature analysis. Eur J Cancer. 2012;48:441-446.
- Kumar V, Gu Y, Basu S, et al. Radiomics: the process and the challenges. Magn Reson Imaging. 2012;30:1234-1248.
- Grossmann P, Stringfield O, El-Hachem N, et al. Defining the biological basis of radiomic phenotypes in lung cancer. Elife. 2017;6.
- Aerts HJ, Velazquez ER, Leijenaar RT, et al. Decoding tumour phenotype by noninvasive imaging using a quantitative radiomics approach. Nat Commun. 2014;5:4006.
- Ferreira Junior JR, Koenigkam-Santos M, Cipriano FEG, Fabro AT, Azevedo-Marques PM. Radiomicsbased features for pattern recognition of lung cancer histopathology and metastases. Comput Methods Programs Biomed. 2018;159:23-30.
- Coroller TP, Agrawal V, Narayan V, et al. Radiomic phenotype features predict pathological response in non-small cell lung cancer. Radiother Oncol. 2016;119:480-486.
- Mattonen SA, Tetar S, Palma DA, Louie AV, Senan S, Ward AD. Imaging texture analysis for automated prediction of lung cancer recurrence after stereotactic radiotherapy. J Med Imaging (Bellingham). 2015;2:041010.
- de Jong EE, van Elmpt W, Leijenaar RT, et al. [18F]FDG PET/CT-based response assessment of stage IV non-small cell lung cancer treated with paclitaxel-carboplatin-bevacizumab with or without nitroglycerin patches. Eur J Nucl Med Mol Imaging. 2016.
- de Jong EE, van Elmpt W, Leijenaar RT, et al. [18F]FDG PET/CT-based response assessment of stage IV non-small cell lung cancer treated with paclitaxel-carboplatin-bevacizumab with or without nitroglycerin patches. Eur J Nucl Med Mol Imaging. 2017;44:8-16.
- Larue R.T.H.M vTJE, de Jong E.E.C., Feliciani G., Leijenaar R.T.H., Schreurs W.M.J., Sosef M.N., Raat F.H.P.J., van der Zande F.H.R., Das M., van Elmpt W.,

Lambin P/. Influence of gray level discretization on radiomic feature stability for different CT scanners, tube currents and slice thicknesses: a comprehensive phantom study. Acta Oncol. 2017;In press:11.

- 24. Suzuki Y, Okamoto T, Fujishita T, et al. Clinical implications of sarcopenia in patients undergoing complete resection for early non-small cell lung cancer. Lung Cancer. 2016;101:92-97.
- Rutten IJ, van Dijk DP, Kruitwagen RF, Beets-Tan RG, Olde Damink SW, van Gorp T. Loss of skeletal muscle during neoadjuvant chemotherapy is related to decreased survival in ovarian cancer patients. J Cachexia Sarcopenia Muscle. 2016;7:458-466.
- Sjoblom B, Gronberg BH, Wentzel-Larsen T, et al. Skeletal muscle radiodensity is prognostic for survival in patients with advanced non-small cell lung cancer. Clin Nutr. 2016;35:1386-1393.
- Haralick RM, Shanmugam K, Dinstein I. Textural Features for Image Classification. IEEE Transactions on Systems, Man, and Cybernetics. 1973;SMC-3:610-621.
- Stephens NA, Skipworth RJ, Macdonald AJ, Greig CA, Ross JA, Fearon KC. Intramyocellular lipid droplets increase with progression of cachexia in cancer patients. J Cachexia Sarcopenia Muscle. 2011;2:111-117.
- Goodpaster BH, Kelley DE, Thaete FL, He J, Ross R. Skeletal muscle attenuation determined by computed tomography is associated with skeletal muscle lipid content. J Appl Physiol (1985). 2000;89:104-110.

SUPPLEMENTALS

Fractal average, GLCM inverseVar, GLRLM RLN, GLSZM IN, GLSZM IV, GLSZM LILAE, GLSZM SAE, IH igr, IH mingrad, IH gcod, LocInt peakGlobal, LocInt peakLocal, NGLDM DV. NGTDM coarseness, Wavelet HHH Fractal lacunarity, Wavelet HHH GLCM clusShade. Wavelet HHH GLCM correl1, Wavelet HHH GLCM infoCorr2, Wavelet HHH GLDZM LISDE, Wavelet HHH GLSZM HILAE, Wavelet HHH GLSZM IN, Wavelet HHH GLSZM SAE, Wavelet HHH IH cov, Wavelet HHH IH gcod, Wavelet HHH NGTDM coarseness, Wavelet HHH Stats cov. Wavelet HHH Stats max. Wavelet HHH Stats mean. Wavelet HHH Stats median, Wavelet HHH Stats gcod, Wavelet HHL Fractal lacunarity, Wavelet HHL GLCM clusShade, Wavelet HHL GLCM correl1, Wavelet HHL GLCM infoCorr1, Wavelet HHL GLCM infoCorr2, Wavelet HHL GLDZM DZE, Wavelet HHL GLDZM LDE, Wavelet HHL GLRLM GLN, Wavelet HHL GLRLM RE, Wavelet HHL GLRLM SRLGE, Wavelet HHL GLSZM HILAE, Wavelet HHL GLSZM IN, Wavelet HHL GLSZM SZNN, Wavelet HHL IH igr, Wavelet HHL IH gcod, Wavelet HHL NGLDM DE, Wavelet HHL NGLDM DNN, Wavelet HHL NGLDM DV, Wavelet HHL NGLDM LGSDE, Wavelet HHL NGTDM coarseness, Wavelet HHL NGTDM contrast, Wavelet HHL NGTDM strength, Wavelet HHL Stats cov, Wavelet HHL Stats max, Wavelet HHL Stats median, Wavelet HHL Stats gcod, Wavelet HLH Fractal average, Wavelet HLH GLCM infoCorr2, Wavelet HLH GLCM invDiffnorm, Wavelet HLH GLCM maxCorr, Wavelet HLH GLSZM LISAE, Wavelet HLH IH cov, Wavelet HLH IH iqr, Wavelet HLH IH mediand, Wavelet HLH IH min, Wavelet HLH IH gcod, Wavelet HLH NGLDM LGSDE, Wavelet HLH NGTDM contrast, Wavelet HLH Stats cov, Wavelet HLH Stats max, Wavelet HLH Stats mean, Wavelet HLH Stats median, Wavelet HLH Stats gcod, Wavelet HLH Stats skewness, Wavelet HLL GLCM infoCorr1, Wavelet HLL GLCM maxCorr, Wavelet HLL GLDZM DZV, Wavelet HLL GLRLM GLN, Wavelet HLL GLRLM LRHGE, Wavelet HLL GLRLM RLN, Wavelet HLL GLSZM HILAE, Wavelet HLL GLSZM SZV, Wavelet HLL IH cov, Wavelet HLL IH igr, Wavelet HLL IH mediand, Wavelet HLL IH gcod, Wavelet HLL IH skewness, Wavelet HLL NGLDM DN, Wavelet HLL NGLDM DV, Wavelet HLL NGTDM coarseness, Wavelet HLL NGTDM strength, Wavelet HLL Stats cov, Wavelet HLL Stats qcod, Wavelet LHH Fractal average, Wavelet LHH Fractal lacunarity, Wavelet LHH GLCM correl1, Wavelet LHH GLCM infoCorr1, Wavelet LHH GLCM infoCorr2, Wavelet LHH GLCM invDiffnorm, Wavelet LHH GLCM maxCorr, Wavelet LHH GLDZM DZV, Wavelet LHH GLSZM IN, Wavelet LHH GLSZM SZNN, Wavelet LHH GLSZM SZV, Wavelet LHH IH cov, Wavelet LHH IH mediand, Wavelet_LHH_IH_qcod, Wavelet_LHH_IH_skewness, Wavelet LHH NGLDM LGSDE, Wavelet LHH NGTDM contrast, Wavelet LHH Stats gcod, Wavelet LHL Fractal average, Wavelet LHL GLCM correl1, Wavelet LHL GLCM infoCorr1, Wavelet LHL GLCM infoCorr2, Wavelet LHL GLCM maxCorr, Wavelet LHL GLDZM SDE, Wavelet LHL GLRLM GLN, Wavelet LHL GLRLM LRLGE, Wavelet LHL GLRLM RLN, Wavelet LHL GLRLM SRLGE, Wavelet LHL GLSZM HILAE, Wavelet LHL GLSZM IN, Wavelet LHL GLSZM ZE, Wavelet LHL IH cov, Wavelet LHL IH iqr, Wavelet LHL IH mediand,

Wavelet_LHL_NGLDM_DN, Wavelet_LHL_NGTDM_strength, Wavelet_LHL_Stats_cov, Wavelet_LHL_Stats_kurtosis, Wavelet_LHL_Stats_max, Wavelet_LHL_Stats_mean, Wavelet_ LLH_Fractal_sd, Wavelet_LLH_GLCM_correl1, Wavelet_LLH_GLCM_infoCorr2, Wavelet_LLH_ GLCM_maxCorr, Wavelet_LLH_GLSZM_IN, Wavelet_LLH_IH_cov, Wavelet_LLH_NGLDM_ SM2, Wavelet_LLH_Stats_kurtosis, Wavelet_LLH_Stats_max, Wavelet_LLH_Stats_median, Wavelet_LLH_Stats_skewness, Wavelet_LLL_GLCM_correl1, Wavelet_LLL_GLCM_infoCorr2, Wavelet_LLL_GLDZM_DZNN, Wavelet_LLL_GLRLM_RE, Wavelet_LLL_GLSZM_INN, Wavelet_ LLL_GLSZM_SZNN, Wavelet_LLL_GLSZM_SZV, Wavelet_LLL_IH_iqr, Wavelet_LLL_IH_maxgrad, Wavelet_LLL_NGLDM_DN, Wavelet_LLL_NGLDM_SM2, Wavelet_LLL_NGTDM_contrast, Wavelet_LLL_Stats_p10, and Wavelet_LLL_Stats_p90.


11

Summary, general discussion, and future perspectives

SUMMARY AND GENERAL DISCUSSION

Cachexia is a frequently observed phenomenon in chronic obstructive pulmonary disease (COPD) and lung cancer and encompasses loss of skeletal muscle and adipose tissue. Tissue loss and redistribution is associated with worse clinical outcomes and interferes with treatment efficacy. Prevention and timely treatment of cachexia requires adequate and early identification of patients at risk and detailed understanding of the pathophysiology involved. In this thesis, standard-of-care medical images for respiratory disease evaluation were used to characterize muscle and adipose tissue. In addition, skeletal muscle radiomic features derived from these images were evaluated for predicting future cachexia development in lung cancer. We furthermore investigated the putative contribution of brown adipose tissue and breathing mechanics to elevated energy requirements in COPD. In this final chapter, these findings are positioned in a broader scientific and clinical perspective considering the latest developments in the management of COPD and lung cancer.

(EARLY) IDENTIFICATION OF PATIENTS AT RISK

It is now widely recognized that loss of skeletal muscle mass adversely impacts clinical outcome and increases healthcare utilization in patients facing cancer or a chronic disease like COPD [1-4]. Low muscle mass is not restricted to those with lean stature and even obese patients can have substantial ongoing muscle depletion [5,6]. Furthermore, stable weight may mask changes in body composition, such as a shift from muscle towards fat mass. There is a growing interest to assess muscle mass with use of techniques that are easily applicable in the clinic or are already being used in regular clinical care for other diagnostic purposes. Thereby, the role of medical imaging is evolving from a diagnostic tool of lung pathology towards an instrument for more extensive disease phenotyping (chapter 2).

Cross-sectional measurement of skeletal muscle

Muscle mass constitutes the major part of fat-free mass, which can be measured in clinical practice using bioimpedance analysis and dual-energy x-ray absorptiometry (DEXA) [7,8]. Since, these modalities are not typically incorporated in routine clinical care, other strategies have been employed. Currently, computed tomography (CT) scans are increasingly utilized in clinical research to measure body composition [9]. This technique allows assessment of individual muscles separately and discriminates different tissue types. Also, assessment of the degree of fat infiltration into the muscle is possible, which is suggested to be an indirect measure of muscle quality [10].

CT derived evaluation of body composition is done by measuring muscle cross sectional area at the level of the third lumbar vertebra [10,11]. Thoracic scans in non-small cell lung cancer (NSCLC) patients usually extend to the third lumbar vertebra level, considering scan protocols include the adrenals to detect the presence of metastasis. However, this is not the case for other lung disease-related clinical purposes as most reach not further than the first lumbar vertebra. Therefore, in COPD a single-muscle approach using the pectoralis muscle has been proposed as alternative, based on associations between low pectoralis muscle with markers of disease severity, symptoms and exercise capacity [12-14]. However, pectoralis muscle was suggested as marker of whole body muscle mass based on moderate associations with clinical outcome measures. Moreover, pectoralis muscle has never been compared with other measurements including a gold standard such as whole body potassium or the gold standard for CT (third lumbar level). In line, we showed a poor correlation between skeletal muscle at the third lumbar level and the pectoralis muscle (chapter 3). Therefore, one might speculate that there is no true association between the two and that causality does not exist. However, the correlation between first and third lumbar level was better in cross-sectional and longitudinal comparison and would therefore, rather than the pectoralis muscle, be a better alternative for analysis of muscle mass from regular chest CT scans (chapter 3). In COPD and aging research, DEXA derived whole body fat-free mass or appendicular lean mass are commonly used to assess muscle mass [15-17]. Correlations reported between DEXA lean mass and CT muscle cross sectional area were moderate [18,19]. One study reported CT derived muscle mass and multiplied the measured area with 1.04 [20], which is the assumed constant density of fatfree skeletal muscle [21]. However, higher degrees of intermuscular fat will decrease density and thereby impacting the accuracy. Therefore, current single slice CT scans cannot substitute DEXA for assessment of cross-sectional whole body muscle mass or vice versa. Furthermore, the current skeletal muscle cut-offs that define cachexia or sarcopenia are based on DEXA and not yet standardized and tailored towards CT measurements [22]. Thereby, one cannot diagnose cachexia or sarcopenia based on a single time point CT scan.

Early changes of skeletal muscle

Longitudinally, there is evidence that repeating DEXA increases measurement errors. DEXA reliability studies have demonstrated good repeatability (less than 2% coefficient of variation) [23]. However, the percentage of change in lean tissue required to be accurately detected by DEXA has been reported in the range of 2.7–6.6% for individual extremities [23]. In this study serial scans were performed within 5 minutes and after repositioning of the subject, indicating that the measured differences can be attributed to machine errors. Changes of muscle mass or changes in hydration status might magnify these measurement errors. Additionally, few longitudinal comparison studies have reported moderate to poor agreement between DEXA and CT derived changes [24,25].

We showed that longitudinally, CT is able to detect subtle changes in body composition which correlated with clinical outcome (chapter 3, chapter 5, and chapter 9). From a clinical practice point of view, future studies to reveal if these changes are also detected by other clinical assessment methods including DEXA would be worth an effort.

Prediction of future muscle loss

Baseline low muscle mass does not necessarily result in longitudinal loss and till now, there are no biomarkers or techniques available to predict muscle loss. In oncology, tumour features are used to visualize tumour heterogeneity and to aid in prognosis [26,27]. This is carried out with the use of radiomics. The concept underlying the process of radiomics is that standard-of-care medical images contain qualitative and quantitative information regarding tumour shape, size, intensity and texture that are related to the underlying pathophysiology of a tissue [28,29]. In this thesis the use of skeletal muscle radiomics was explored for its potential to predict future cachexia development in lung cancer. Skeletal muscle radiomic features had no discriminatory value with regard to longitudinal loss of skeletal muscle. However, we did observe longitudinal differences in radiomic features between those who lost muscle and those who maintained muscle mass. Future imaging analysis combined with muscle tissue analysis may unravel the biological processes linked to the radiomic features (chapter 10). However, radiomic features are impacted by constant developments in radiology regarding scanner acquisition protocols, exposure control, dose modulation and reconstruction algorithms and parameters. Up to now, despite encouraging results these challenges prevented the use of radiomics in routine practice to aid in clinical decision making. Within this context it is therefore crucial to focus on stable parameters across different scanners and to validate these parameters in a multicentric context [30-33].

ADIPOSE TISSUE

In cachexia, skeletal muscle loss is often accompanied by depletion of adipose tissue, driven by increased lipolysis [34,35]. Lipolysis generates free fatty acids which are able to transport into myocytes and through a process of oxidation stimulate protein degradation [36,37]. Indeed, skeletal muscle of cancer patients contains more intramyocellular fat compared to age- and gender matched controls [38]. Furthermore, a rise in the number of lipid droplets has been demonstrated in relation to increasing weight loss and loss of adipose tissue in cancer [39]. Inhibition of lipolysis in a lung cancer mouse model not only prevented depletion of fat but also prevented loss of muscle mass [40]. Together these observations point towards a possible muscle-fat crosstalk in cachexia.

While lipolysis represents the loss of white adipose tissue, brown adipose tissue (BAT) or browning of white adipose tissue uses stored energy for thermogenesis, which is an energy costly process [41]. Thereby, the association between BAT and elevated energy expenditure in cachexia has been suspected for a long time [42]. It was shown that activation of BAT contributed to development of cachexia in mice with colorectal cancer. Furthermore, BAT activation was not present in non-cachectic mice with colorectal cancer [43]. Human research of BAT in relation to cachexia is scarce and conflicting. Autopsy reports of cachectic cancer patients revealed higher incidence of BAT compared to age-matched controls [42]. Unfortunately,

no data were presented regarding the stage of cancer. In contrast to the autopsy reports, a retrospective imaging study reported a similar prevalence of BAT between subjects with and without cancer [44]. However, these results might be biased by the retrospective design, which did not allow to control environmental temperatures as BAT is known to be activated upon cold in healthy young subjects [45,46]. As direct evidence linking BAT activity, energy metabolism and cachexia was not available, this thesis investigated the contribution of BAT on different components of energy metabolism in muscle depleted patients. In COPD, muscle wasting [47,48] and hypermetabolism [49] have particularly been demonstrated among those with emphysema. Therefore, this thesis studied the contributing role of BAT to energy metabolism in emphysematous COPD patients. In contrast to our hypothesis, hypermetabolic COPD patients with the emphysematous phenotype did not exhibit enhanced BAT activity or altered gene expression of BAT or beige markers in white adipose tissue compared to age-, gender- and body mass index matched healthy non-smoking controls (chapter 7). The original intention was to compare the COPD patients not only to healthy controls, but also to matched cachectic lung cancer and cachectic pancreatic cancer patients. However, the BAT protocol was too strenuous, regarding eligibility criteria as well as methodology, for most of the cachectic cancer patients. The current study protocol used cold to activate BAT. Study subjects were wrapped in a water-perfused suit for several hours while temperature decreased slightly from 36 to on average 20 degrees. Furthermore, medication influencing the sympathetic nerve system and thereby interfering with BAT activity, including ß-blockers and calcium channel blockers was prohibited. Since ß-blockers and calcium channel were widely used among our study population, many potential particiants were excluded. This resulted in the inclusion of only two pancreatic cancer patients before study closure. Both pancreatic patients lost > 5% body weight in the previous six months and were both hypermetabolic at rest. Surprisingly no BAT activity was found among those two cachectic cancer patients. Although based on only two cancer patients, these results indicate that BAT does not play a role in the pathophysiology of cancer associated cachexia.

SUPPORTIVE INTERVENTIONS IN COPD

In addition to the loss of lung tissue resulting in emphysema, a subgroup shows a concordant enhanced loss of muscle tissue [50]. Muscle depletion observed in emphysema is associated with hypermetabolism (chapter 4, chapter 6). It has been proposed that the working load of breathing increases energy metabolism. We indeed showed that patients with severe emphysema and hyperinflation were hypermetabolic at rest (130% of predicted) (chapter 6). Innovative endoscopic interventions including endobronchial valves to reduce hyperinflation are a growing new therapeutic field [51]. After hyperinflation reduction CT analysis showed that patients gained skeletal muscle and fat mass (chapter 5) implying an overall positive energy balance. Nevertheless, no impact of hyperinflation reduction by endobronchial valves on the different components of energy metabolism was observed after six months and patients remained hypermetabolic at rest (chapter 6). Improving body composition abnormalities and aiming to provide the best possible baseline condition before treatment with EBV, might result in greater clinical benefits from EBV. To improve body composition, improve exercise capacity and positively influencing the clinical course of the disease in patients with COPD, pulmonary rehabilitation is currently accepted as evidence-based intervention strategy [52]. Mean improvements following pulmonary rehabilitation are comparable after stratification for baseline airflow limitation [53] and static lung hyperinflation [54]. However, data on pulmonary rehabilitation outcomes in patients with severe emphysema are scarce.

The National Emphysema Treatment Trial performed in the United States evaluating the effect of lung volume reduction surgery in patients with severe emphysema included a mandatory preoperative pulmonary rehabilitation program. The rationale was to facilitate early postoperative mobilization and to provide optimized preoperative exercise capacity [55]. Although the study was not specifically designed to evaluate the effect of pulmonary rehabilitation, significant improvements in exercise capacity, dyspnoea and health related quality of life were observed [56]. This provides evidence that pulmonary rehabilitation is feasible among patients with severe emphysema and that it alleviates symptoms and improves functional capacity [57]. One study evaluated whether preoperative improvements due to pulmonary rehabilitation can augment the benefits from lung volume reduction surgery or reduce postoperative complication rates. A cohort of 22 COPD patients who received 3 weeks of pulmonary rehabilitation prior to lung volume reduction surgery significantly improved in 6 minute walk distance. However, no additional postoperative gain in 6 minute walk distance was observed [58]. Nevertheless, follow-up time postoperative was only 5 weeks and data regarding complication rates and quality of life were lacking. There are currently no data available whether pre- and postoperative pulmonary rehabilitation can augment the clinical outcome of patients receiving EBV.

SUPPORTIVE INTERVENTIONS IN NSCLC

Despite increased understanding of the pathophysiology, there are still no evidence-based guidelines on the management of cancer cachexia. Given the multifaceted pathophysiology, it is thought that it requires a multimodal treatment approach encompassing nutrition as well as physical exercise, to preserve muscle mass, strength and function [59,60]. The optimal time to initiate such treatment might be early in the disease trajectory. A significant part of non-resectable stage III NSCLC patients who were weight stable before concurrent chemoradiation experienced weight loss during the first three weeks after therapy initiation (chapter 8). Furthermore, nearly two-third of stage IV NSCLC patients lose skeletal muscle after only two cycles of first line chemotherapy (chapter 9). Together these results argue for implementing cachexia-directed therapy in the anti-tumour treatment.

The average caloric deficit in patient with advanced-stage NSCLC has been estimated to be approximately 200 kcal per day. Moreover, the mean protein intake of these patients was 0.9 g/kg/day [61,62], which was significantly lower than the recommended daily protein intake of 1.0-1.5 g/kg/day [63]. Although dietary counselling in patients with various tumour types including NSCLC has found to improve body weight [64-66], systematic reviews and meta-analyses of randomised controlled trials found limited evidence of benefit of dietary counselling alone on clinical outcomes in lung cancer in terms of quality of life and mortality [67-69].

Regarding physical activity, low activity levels have been reported among NSCLC patients [70] with an average daily step count of approximately 4500 steps [71,72], and it is estimated that only 20% of the patients followed the advice regarding daily exercise [73]. As a potent anabolic stimulus, exercise might antagonize muscle atrophy in cachexia. Although proven feasible [74] and despite positive effects following exercise intervention in lung cancer patients at an earlier stage of disease [75-77], studies evaluating the effects of exercise in advanced lung cancer to combat cachexia are scarce. Only two exercise intervention studies have included advanced lung cancer patients. A feasibility single-arm study reported a trend towards increased functional capacity and muscle strength following a 8-week aerobic and resistance training in advanced lung cancer [78]. Others found significant improvement in functional capacity and muscle strength following a 8-week aerobic and resistance training in advanced lung cancer [78]. Others found significant improvement in functional capacity and muscle strength following a 8-week aerobic and resistance training in advanced lung cancer [78]. Others found significant improvement in functional capacity and muscle strength following a similar exercise intervention [74]. Despite that no significant changes in quality of life were found [74,78], an improvement in lung cancer symptoms was reported [78]. The evidence suggests benefit of combined resistance and aerobic muscle training; however randomized clinical trials in patients with advanced cancer are warranted to establish such programmes to counteract cachexia [79,80].

In stage III and stage IV NSCLC patients evidence of a multimodal cachexia treatment is very scarce. One study examined the effects of treatment with an appetite stimulant medroxyprogesterone and an anti-inflammatory drug celecoxib with food supplementation, recommendation of regular exercise and psychological assistance in 15 stage III and IV cachectic NSCLC patients. After 6 weeks, caloric intake increased with 22% and 13 patients either had stable or gained weight [73]. A feasibility study evaluated the effects of a combined intervention with polyunsaturated fatty acid nutritional supplements, exercise and anti-inflammatory drug celecoxib in advanced NSCLC and pancreatic cancer patients. Most of the included patients exhibited stage IV NSCLC with good performance score and who lost on average 5% of body weight prior to the intervention. Although the study was not specifically designed to evaluate the effect of the intervention, weight maintenance compared to standard care was reported [81].

Currently, results of the phase III multimodal-exercise, nutrition and anti-inflammatory medication for cachexia trial (NCT02330926) are awaited to fully assess the effect of the

multimodal intervention [82]. This trial aims to prevent the development of cachexia and/or to attenuate cachexia progression in patients with stage III-IV NSCLC, stage III-IV pancreatic cancer or cholangiocarcinoma, who commence first or second line anticancer treatment. The intervention consists of a 6-week program with nutritional supplements and advice, home-based exercise program, anti-inflammatory drug ibuprofen and standard care. The primary endpoint is weight loss and secondary endpoints are change in muscle mass and change in physical activity.

Two studies integrated exercise intervention in anticancer therapy in patients with advanced lung cancer [74,78]. Although the authors concluded that the intervention during anticancer therapy was feasible, exercise adherence was low since on average 42% of the participants completed all exercise sessions. Patients were unable to complete mainly due to clinical deterioration, hospitalization or loss of motivation.

Up to now unimodal strategies aiming to counteract cachexia have failed to demonstrate consistent clinical benefits and results of multimodal interventions are awaited. There is a need for intervention studies recruiting patients early during their disease trajectory which are feasible for this vulnerable population and which target both metabolism, nutritional intake and muscle anabolism.

FUTURE PERSPECTIVES

When looking at the future, after a period without major innovations in the field of cachexia prevention and treatment, ongoing research at different domains have the potential to change the landscape.

Energy balance: from an intake perspective

A net catabolic state is the result from an imbalance in energy expenditure and energy intake. In order to compensate for increased energy requirements patients should be able to adapt their dietary intake. Dietary intake is controlled by the central nervous system, which integrates peripheral signals of satiety and adiposity. Brain imaging studies have revealed specific brain regions that are related to food reward-value [83,84]. Functional neuroimaging studies reported altered reactivity in the brain reward system in people with altered eating patterns, including anorexia nervosa and obesity [85-87].

In relation to cancer only one study has been performed identifying brain activity in anorectic and non-anorectic lung cancer patients. In contrast to non-anorectic patients, anorectic patients showed no brain activity differences in response to pleasant versus unpleasant food cues, implying an overall blunted response in the perceptual and motivational system. Unfortunately, the relation between blunted food reward and cachexia was not evaluated and requires further investigation [88].

Cachexia and the efficacy of anticancer treatment

There is some evidence from experimental research suggesting that cancer therapy aiming to achieve tumour control might stimulate catabolic processes in the muscle via activation of nuclear factor kappa B and upregulation of myostatin, and thereby contribute to onset or progression of cachexia [89]. Also the other way around, cachexia impacts the efficacy of anticancer treatment. For example, low muscle mass in NSCLC patients is associated with haematological toxicity [90], which is thought to be the result of a high absolute dose of chemotherapy with a reduced volume of distribution. Currently, chemotherapy dosing utilizes body surface area [91], which does not account for the distribution of muscle and adipose tissue. Some studies have tried to explore the use of body composition to guide chemotherapy dosing in patients with advanced cancer [92,93]. CT derived lean body mass might provide a basis for better dosing chemotherapy individually. However, prospective clinical trials are warrant.

Currently, the landscape of (locally) advanced NSCLC treatment is evolving rapidly. In the absence of treatable oncogenic alterations, platinum-based chemotherapy remains the cornerstone of treatment [94]. In patients with advanced NSCLC, immune checkpoint inhibitors, which unleash a patients' own T lymphocytes to kill tumours, have demonstrated to improve outcomes in stage IV patients [95-100]. More recently, immunotherapy has shown to be also promising in stage III patients following concurrent chemoradiotherapy [101], and trials regarding (neo-) adjuvant immunotherapy are ongoing (NCT03425643, NCT02998528).

Although premature, it has recently been suggested that the exposure-response relationship for immunotherapy is confounded by cachexia. A post-hoc analysis of immunotherapy treated NSCLC and melanoma patients reported that rapid clearance of pembrolizumab was strongly linked to decreased overall survival, and that those with rapid clearance exhibited more weight loss compared to those with normal-to-slow clearance [102]. Additionally, nivolumab clearance was greater in patients with lower albumin levels (mild hepatic impairment did not affect clearance) [103]. T cells rely on glucose for cytolytic activity and availability of adequate nutrients is crucial. Experimental research showed that high glycolysis within the tumour cells cause depletion of extracellular glucose which restricts T cells in its direct microenvironment [104,105]. This metabolic competition between tumour and immune cells may lead to T cell hyporesponsiveness, emphasizing the close relationship between cellular nutritional status and T lymphocyte function. Whole body catabolism such as seen in cachexia may also influence the T-cell microenvironment, however studies are lacking. Together these studies argue for a possible role for cachexia in efficacy of immunotherapy. To conclude, the underlying mechanisms of cachexia are still poorly defined hampering effective medical interventions including targeted drug therapies. With the results of this thesis we were able to add relevant insights into the pathophysiology, but there is still potential for improvement. New molecular and clinical studies are running which hopefully prove valuable in the management of cachexia in COPD and lung cancer.

REFERENCES

- Schols AM, Broekhuizen R, Weling-Scheepers CA, Wouters EF. Body composition and mortality in chronic obstructive pulmonary disease. Am J Clin Nutr. 2005;82:53-59.
- Shoup R, Dalsky G, Warner S, et al. Body composition and health-related quality of life in patients with obstructive airways disease. Eur Respir J. 1997;10:1576-1580.
- Shachar SS, Deal AM, Weinberg M, et al. Skeletal Muscle Measures as Predictors of Toxicity, Hospitalization, and Survival in Patients with Metastatic Breast Cancer Receiving Taxane-Based Chemotherapy. Clin Cancer Res. 2017;23:658-665.
- Malietzis G, Currie AC, Athanasiou T, et al. Influence of body composition profile on outcomes following colorectal cancer surgery. Br J Surg. 2016;103:572-580.
- Prado CM, Lieffers JR, McCargar LJ, et al. Prevalence and clinical implications of sarcopenic obesity in patients with solid tumours of the respiratory and gastrointestinal tracts: a population-based study. Lancet Oncol. 2008;9:629-635.
- Tan BH, Birdsell LA, Martin L, Baracos VE, Fearon KC. Sarcopenia in an overweight or obese patient is an adverse prognostic factor in pancreatic cancer. Clin Cancer Res. 2009;15:6973-6979.
- Jebb SA. Measurement of soft tissue composition by dual energy X-ray absorptiometry. Br J Nutr. 1997;77:151-163.
- Steiner MC, Barton RL, Singh SJ, Morgan MD. Bedside methods versus dual energy X-ray absorptiometry for body composition measurement in COPD. Eur Respir J. 2002;19:626-631.
- Mitsiopoulos N, Baumgartner RN, Heymsfield SB, Lyons W, Gallagher D, Ross R. Cadaver validation of skeletal muscle measurement by magnetic resonance imaging and computerized tomography. J Appl Physiol (1985). 1998;85:115-122.
- Shen W, Punyanitya M, Wang Z, et al. Total body skeletal muscle and adipose tissue volumes: estimation from a single abdominal cross-sectional image. J Appl Physiol (1985). 2004;97:2333-2338.
- Schweitzer L, Geisler C, Pourhassan M, et al. What is the best reference site for a single MRI slice to assess whole-body skeletal muscle and adipose tissue volumes in healthy adults? Am J Clin Nutr. 2015;102:58-65.
- 12. McDougall R, Martin D, Gillam L, Hallowell N,

Brookes A, Guillemin M. Therapeutic appropriation: a new concept in the ethics of clinical research. J Med Ethics. 2016;42:805-808.

- Bak SH, Kwon SO, Han SS, Kim WJ. Computed tomography-derived area and density of pectoralis muscle associated disease severity and longitudinal changes in chronic obstructive pulmonary disease: a case control study. Respir Res. 2019;20:226.
- Diaz AA, Martinez CH, Harmouche R, et al. Pectoralis muscle area and mortality in smokers without airflow obstruction. Respir Res. 2018;19:62.
- Schols AM, Soeters PB, Dingemans AM, Mostert R, Frantzen PJ, Wouters EF. Prevalence and characteristics of nutritional depletion in patients with stable COPD eligible for pulmonary rehabilitation. Am Rev Respir Dis. 1993;147:1151-1156.
- Vestbo J, Prescott E, Almdal T, et al. Body mass, fat-free body mass, and prognosis in patients with chronic obstructive pulmonary disease from a random population sample: findings from the Copenhagen City Heart Study. Am J Respir Crit Care Med. 2006;173:79-83.
- 17. Marzetti E, Calvani R, Tosato M, et al. Sarcopenia: an overview. Aging Clin Exp Res. 2017;29:11-17.
- Levine JA, Abboud L, Barry M, Reed JE, Sheedy PF, Jensen MD. Measuring leg muscle and fat mass in humans: comparison of CT and dual-energy X-ray absorptiometry. J Appl Physiol (1985). 2000;88:452-456.
- Bredella MA, Ghomi RH, Thomas BJ, et al. Comparison of DXA and CT in the assessment of body composition in premenopausal women with obesity and anorexia nervosa. Obesity (Silver Spring). 2010;18:2227-2233.
- Visser M, Fuerst T, Lang T, Salamone L, Harris TB. Validity of fan-beam dual-energy X-ray absorptiometry for measuring fat-free mass and leg muscle mass. Health, Aging, and Body Composition Study--Dual-Energy X-ray Absorptiometry and Body Composition Working Group. J Appl Physiol (1985). 1999;87:1513-1520.
- Chowdhury B, Sjostrom L, Alpsten M, Kostanty J, Kvist H, Lofgren R. A multicompartment body composition technique based on computerized tomography. Int J Obes Relat Metab Disord. 1994;18:219-234.
- Taguchi S, Nakagawa T, Fukuhara H. Inconsistencies in currently used definitions of sarcopenia in oncology. Ann Oncol. 2020;31:318-319.

- Hind K, Oldroyd B. In-vivo precision of the GE Lunar iDXA densitometer for the measurement of appendicular and trunk lean and fat mass. Eur J Clin Nutr. 2013;67:1331-1333.
- Delmonico MJ, Kostek MC, Johns J, Hurley BF, Conway JM. Can dual energy X-ray absorptiometry provide a valid assessment of changes in thigh muscle mass with strength training in older adults? Eur J Clin Nutr. 2008;62:1372-1378.
- Nelson ME, Fiatarone MA, Layne JE, et al. Analysis of body-composition techniques and models for detecting change in soft tissue with strength training. Am J Clin Nutr. 1996;63:678-686.
- Aerts HJ, Velazquez ER, Leijenaar RT, et al. Decoding tumour phenotype by noninvasive imaging using a quantitative radiomics approach. Nat Commun. 2014;5:4006.
- Mattonen SA, Tetar S, Palma DA, Louie AV, Senan S, Ward AD. Imaging texture analysis for automated prediction of lung cancer recurrence after stereotactic radiotherapy. J Med Imaging (Bellingham). 2015;2:041010.
- Lambin P, van Stiphout RG, Starmans MH, et al. Predicting outcomes in radiation oncology-multifactorial decision support systems. Nat Rev Clin Oncol. 2013;10:27-40.
- Lambin P, Rios-Velazquez E, Leijenaar R, et al. Radiomics: extracting more information from medical images using advanced feature analysis. Eur J Cancer. 2012;48:441-446.
- Kumar V, Gu Y, Basu S, et al. Radiomics: the process and the challenges. Magn Reson Imaging. 2012;30:1234-1248.
- Gatta R, Depeursinge A, Ratib O, Michielin O, Leimgruber A. Integrating radiomics into holomics for personalised oncology: from algorithms to bedside. Eur Radiol Exp. 2020;4:11.
- Rizzo S, Botta F, Raimondi S, et al. Radiomics: the facts and the challenges of image analysis. Eur Radiol Exp. 2018;2:36.
- Hatt M, Lucia F, Schick U, Visvikis D. Multicentric validation of radiomics findings: challenges and opportunities. EBioMedicine. 2019;47:20-21.
- Ryden M, Agustsson T, Laurencikiene J, et al. Lipolysis--not inflammation, cell death, or lipogenesis--is involved in adipose tissue loss in cancer cachexia. Cancer. 2008;113:1695-1704.
- Agustsson T, Ryden M, Hoffstedt J, et al. Mechanism of increased lipolysis in cancer cachexia. Cancer Res. 2007;67:5531-5537.

- Zhou Q, Du J, Hu Z, Walsh K, Wang XH. Evidence for adipose-muscle cross talk: opposing regulation of muscle proteolysis by adiponectin and Fatty acids. Endocrinology. 2007;148:5696-5705.
- Julienne CM, Dumas JF, Goupille C, et al. Cancer cachexia is associated with a decrease in skeletal muscle mitochondrial oxidative capacities without alteration of ATP production efficiency. J Cachexia Sarcopenia Muscle. 2012;3:265-275.
- Weber MA, Krakowski-Roosen H, Schroder L, et al. Morphology, metabolism, microcirculation, and strength of skeletal muscles in cancer-related cachexia. Acta Oncol. 2009;48:116-124.
- Stephens NA, Skipworth RJ, Macdonald AJ, Greig CA, Ross JA, Fearon KC. Intramyocellular lipid droplets increase with progression of cachexia in cancer patients. J Cachexia Sarcopenia Muscle. 2011;2:111-117.
- Das SK, Eder S, Schauer S, et al. Adipose triglyceride lipase contributes to cancer-associated cachexia. Science. 2011;333:233-238.
- Heaton GM, Wagenvoord RJ, Kemp A, Jr., Nicholls DG. Brown-adipose-tissue mitochondria: photoaffinity labelling of the regulatory site of energy dissipation. Eur J Biochem. 1978;82:515-521.
- Shellock FG, Riedinger MS, Fishbein MC. Brown adipose tissue in cancer patients: possible cause of cancer-induced cachexia. J Cancer Res Clin Oncol. 1986;111:82-85.
- Tsoli M, Moore M, Burg D, et al. Activation of thermogenesis in brown adipose tissue and dysregulated lipid metabolism associated with cancer cachexia in mice. Cancer Res. 2012;72:4372-4382.
- Lee P, Greenfield JR, Ho KK, Fulham MJ. A critical appraisal of the prevalence and metabolic significance of brown adipose tissue in adult humans. Am J Physiol Endocrinol Metab. 2010;299:E601-606.
- van Marken Lichtenbelt WD, Vanhommerig JW, Smulders NM, et al. Cold-activated brown adipose tissue in healthy men. N Engl J Med. 2009;360:1500-1508.
- Yoneshiro T, Aita S, Matsushita M, et al. Brown adipose tissue, whole-body energy expenditure, and thermogenesis in healthy adult men. Obesity (Silver Spring). 2011;19:13-16.
- 47. Engelen MP, Schols AM, Lamers RJ, Wouters EF. Different patterns of chronic tissue wasting among

patients with chronic obstructive pulmonary disease. Clin Nutr. 1999;18:275-280.

- Engelen MP, Schols AM, Baken WC, Wesseling GJ, Wouters EF. Nutritional depletion in relation to respiratory and peripheral skeletal muscle function in out-patients with COPD. Eur Respir J. 1994;7:1793-1797.
- Cohen RI, Marzouk K, Berkoski P, O'Donnell CP, Polotsky VY, Scharf SM. Body composition and resting energy expenditure in clinically stable, nonweight-losing patients with severe emphysema. Chest. 2003;124:1365-1372.
- Celli BR, Locantore N, Tal-Singer R, et al. Emphysema and extrapulmonary tissue loss in COPD: a multi-organ loss of tissue phenotype. Eur Respir J. 2018;51.
- Shah PL, Herth FJ, van Geffen WH, Deslee G, Slebos DJ. Lung volume reduction for emphysema. Lancet Respir Med. 2017;5:147-156.
- Spruit MA, Singh SJ, Garvey C, et al. An official American Thoracic Society/European Respiratory Society statement: key concepts and advances in pulmonary rehabilitation. Am J Respir Crit Care Med. 2013;188:e13-64.
- Augustin IML, Wouters EFM, Houben-Wilke S, et al. Comprehensive Lung Function Assessment Does not Allow to Infer Response to Pulmonary Rehabilitation in Patients with COPD. J Clin Med. 2018;8.
- Vanfleteren MJ, Koopman M, Spruit MA, et al. Effectiveness of Pulmonary Rehabilitation in Patients With Chronic Obstructive Pulmonary Disease With Different Degrees of Static Lung Hyperinflation. Arch Phys Med Rehabil. 2018;99:2279-2286 e2273.
- 55. Rationale and design of The National Emphysema Treatment Trial: a prospective randomized trial of lung volume reduction surgery. The National Emphysema Treatment Trial Research Group. Chest. 1999;116:1750-1761.
- Ries AL, Make BJ, Lee SM, et al. The effects of pulmonary rehabilitation in the national emphysema treatment trial. Chest. 2005;128:3799-3809.
- Ries AL, Make BJ, Reilly JJ. Pulmonary rehabilitation in emphysema. Proc Am Thorac Soc. 2008;5:524-529.
- Irie M, Nakanishi R, Hamada K, Kido M. Perioperative short-term pulmonary rehabilitation for patients undergoing lung volume reduction

surgery. COPD. 2011;8:444-449.

- Fearon KC. Cancer cachexia: developing multimodal therapy for a multidimensional problem. Eur J Cancer. 2008;44:1124-1132.
- Fearon K. Cachexia: Treat wasting illness on multiple fronts. Nature. 2016;529:156.
- Mohan A, Poulose R, Kulshreshtha I, et al. High prevalence of malnutrition and deranged relationship between energy demands and food intake in advanced non-small cell lung cancer. Eur J Cancer Care (Engl). 2017;26.
- Percival C, Hussain A, Zadora-Chrzastowska S, White G, Maddocks M, Wilcock A. Providing nutritional support to patients with thoracic cancer: findings of a dedicated rehabilitation service. Respir Med. 2013;107:753-761.
- Arends J, Bachmann P, Baracos V, et al. ESPEN guidelines on nutrition in cancer patients. Clin Nutr. 2017;36:11-48.
- 64. Baldwin C, Spiro A, McGough C, et al. Simple nutritional intervention in patients with advanced cancers of the gastrointestinal tract, non-small cell lung cancers or mesothelioma and weight loss receiving chemotherapy: a randomised controlled trial. J Hum Nutr Diet. 2011;24:431-440.
- 65. Evans WK, Nixon DW, Daly JM, et al. A randomized study of oral nutritional support versus ad lib nutritional intake during chemotherapy for advanced colorectal and non-small-cell lung cancer. J Clin Oncol. 1987;5:113-124.
- 66. Ovesen L, Allingstrup L, Hannibal J, Mortensen EL, Hansen OP. Effect of dietary counseling on food intake, body weight, response rate, survival, and quality of life in cancer patients undergoing chemotherapy: a prospective, randomized study. J Clin Oncol. 1993;11:2043-2049.
- Baldwin C, Spiro A, Ahern R, Emery PW. Oral nutritional interventions in malnourished patients with cancer: a systematic review and metaanalysis. J Natl Cancer Inst. 2012;104:371-385.
- Halfdanarson TR, Thordardottir E, West CP, Jatoi A. Does dietary counseling improve quality of life in cancer patients? A systematic review and metaanalysis. J Support Oncol. 2008;6:234-237.
- Kiss NK, Krishnasamy M, Isenring EA. The effect of nutrition intervention in lung cancer patients undergoing chemotherapy and/or radiotherapy: a systematic review. Nutr Cancer. 2014;66:47-56.
- Granger CL, McDonald CF, Irving L, et al. Low physical activity levels and functional decline

in individuals with lung cancer. Lung Cancer. 2014;83:292-299.

- Bade BC, Hyer JM, Bevill BT, et al. A Patient-Centered Activity Regimen Improves Participation in Physical Activity Interventions in Advanced-Stage Lung Cancer. Integr Cancer Ther. 2018;17:921-927.
- Bade BC, Brooks MC, Nietert SB, et al. Assessing the Correlation Between Physical Activity and Quality of Life in Advanced Lung Cancer. Integr Cancer Ther. 2018;17:73-79.
- Cerchietti LC, Navigante AH, Peluffo GD, et al. Effects of celecoxib, medroxyprogesterone, and dietary intervention on systemic syndromes in patients with advanced lung adenocarcinoma: a pilot study. J Pain Symptom Manage. 2004;27:85-95.
- 74. Quist M, Rorth M, Langer S, et al. Safety and feasibility of a combined exercise intervention for inoperable lung cancer patients undergoing chemotherapy: a pilot study. Lung Cancer. 2012;75:203-208.
- Spruit MA, Janssen PP, Willemsen SC, Hochstenbag MM, Wouters EF. Exercise capacity before and after an 8-week multidisciplinary inpatient rehabilitation program in lung cancer patients: a pilot study. Lung Cancer. 2006;52:257-260.
- Bobbio A, Chetta A, Ampollini L, et al. Preoperative pulmonary rehabilitation in patients undergoing lung resection for non-small cell lung cancer. Eur J Cardiothorac Surg. 2008;33:95-98.
- Cesario A, Ferri L, Galetta D, et al. Pre-operative pulmonary rehabilitation and surgery for lung cancer. Lung Cancer. 2007;57:118-119.
- Temel JS, Greer JA, Goldberg S, et al. A structured exercise program for patients with advanced nonsmall cell lung cancer. J Thorac Oncol. 2009;4:595-601.
- Grande AJ, Silva V, Riera R, et al. Exercise for cancer cachexia in adults. Cochrane Database Syst Rev. 2014:CD010804.
- Grande AJ, Silva V, Maddocks M. Exercise for cancer cachexia in adults: Executive summary of a Cochrane Collaboration systematic review. J Cachexia Sarcopenia Muscle. 2015;6:208-211.
- Solheim TS, Laird BJA, Balstad TR, et al. A randomized phase II feasibility trial of a multimodal intervention for the management of cachexia in lung and pancreatic cancer. J Cachexia Sarcopenia Muscle. 2017;8:778-788.
- 82. Solheim TS, Laird BJA, Balstad TR, et al. Cancer

cachexia: rationale for the MENAC (Multimodal-Exercise, Nutrition and Anti-inflammatory medication for Cachexia) trial. BMJ Support Palliat Care. 2018;8:258-265.

- Sescousse G, Caldu X, Segura B, Dreher JC. Processing of primary and secondary rewards: a quantitative meta-analysis and review of human functional neuroimaging studies. Neurosci Biobehav Rev. 2013;37:681-696.
- Murdaugh DL, Cox JE, Cook EW, 3rd, Weller RE. fMRI reactivity to high-calorie food pictures predicts short- and long-term outcome in a weightloss program. Neuroimage. 2012;59:2709-2721.
- Stice E, Yokum S, Burger KS, Epstein LH, Small DM. Youth at risk for obesity show greater activation of striatal and somatosensory regions to food. J Neurosci. 2011;31:4360-4366.
- Roefs A, Stapert D, Isabella LA, Wolters G, Wojciechowski F, Jansen A. Early associations with food in anorexia nervosa patients and obese people assessed in the affective priming paradigm. Eat Behav. 2005;6:151-163.
- Dimitropoulos A, Tkach J, Ho A, Kennedy J. Greater corticolimbic activation to high-calorie food cues after eating in obese vs. normal-weight adults. Appetite. 2012;58:303-312.
- Sanchez-Lara K, Arrieta O, Pasaye E, et al. Brain activity correlated with food preferences: a functional study comparing advanced nonsmall cell lung cancer patients with and without anorexia. Nutrition. 2013;29:1013-1019.
- Costelli P, Muscaritoli M, Bonetto A, et al. Muscle myostatin signalling is enhanced in experimental cancer cachexia. Eur J Clin Invest. 2008;38:531-538.
- Sjoblom B, Gronberg BH, Benth JS, et al. Low muscle mass is associated with chemotherapyinduced haematological toxicity in advanced nonsmall cell lung cancer. Lung Cancer. 2015;90:85-91.
- 91. Griggs JJ, Mangu PB, Temin S, Lyman GH. Appropriate Chemotherapy Dosing for Obese Adult Patients With Cancer: American Society of Clinical Oncology Clinical Practice Guideline. J Oncol Pract. 2012;8:e59-e61.
- Sjoblom B, Benth JS, Gronberg BH, et al. Drug Dose Per Kilogram Lean Body Mass Predicts Hematologic Toxicity From Carboplatin-Doublet Chemotherapy in Advanced Non-Small-Cell Lung Cancer. Clin Lung Cancer. 2017;18:e129-e136.
- 93. Crosby V, D'Souza C, Bristow C, et al. Can body

composition be used to optimize the dose of platinum chemotherapy in lung cancer? A feasibility study. Support Care Cancer. 2017;25:1257-1261.

- Planchard D, Popat S, Kerr K, et al. Metastatic non-small cell lung cancer: ESMO Clinical Practice Guidelines for diagnosis, treatment and follow-up. Ann Oncol. 2019;30:863-870.
- Brahmer J, Reckamp KL, Baas P, et al. Nivolumab versus Docetaxel in Advanced Squamous-Cell Non-Small-Cell Lung Cancer. N Engl J Med. 2015;373:123-135.
- Borghaei H, Paz-Ares L, Horn L, et al. Nivolumab versus Docetaxel in Advanced Nonsquamous Non-Small-Cell Lung Cancer. N Engl J Med. 2015;373:1627-1639.
- Rittmeyer A, Barlesi F, Waterkamp D, et al. Atezolizumab versus docetaxel in patients with previously treated non-small-cell lung cancer (OAK): a phase 3, open-label, multicentre randomised controlled trial. Lancet. 2017;389:255-265.
- Herbst RS, Baas P, Kim DW, et al. Pembrolizumab versus docetaxel for previously treated, PD-L1positive, advanced non-small-cell lung cancer (KEYNOTE-010): a randomised controlled trial. Lancet. 2016;387:1540-1550.
- Mok TSK, Wu YL, Kudaba I, et al. Pembrolizumab versus chemotherapy for previously untreated, PD-L1-expressing, locally advanced or metastatic non-small-cell lung cancer (KEYNOTE-042): a randomised, open-label, controlled, phase 3 trial. Lancet. 2019;393:1819-1830.
- Reck M, Rodriguez-Abreu D, Robinson AG, et al. Pembrolizumab versus Chemotherapy for PD-L1-Positive Non-Small-Cell Lung Cancer. N Engl J Med. 2016;375:1823-1833.
- Antonia SJ, Villegas A, Daniel D, et al. Overall Survival with Durvalumab after Chemoradiotherapy in Stage III NSCLC. N Engl J Med. 2018;379:2342-2350.
- 102. Turner DC, Kondic AG, Anderson KM, et al. Pembrolizumab Exposure-Response Assessments Challenged by Association of Cancer Cachexia and Catabolic Clearance. Clin Cancer Res. 2018;24:5841-5849.
- 103. Bajaj G, Wang X, Agrawal S, Gupta M, Roy A, Feng Y. Model-Based Population Pharmacokinetic Analysis of Nivolumab in Patients With Solid Tumors. CPT Pharmacometrics Syst Pharmacol. 2017;6:58-66.
- 104. Chang CH, Qiu J, O'Sullivan D, et al. Metabolic

Competition in the Tumor Microenvironment Is a Driver of Cancer Progression. Cell. 2015;162:1229-1241.

 Jacobs SR, Herman CE, Maciver NJ, et al. Glucose uptake is limiting in T cell activation and requires CD28-mediated Akt-dependent and independent pathwavs. J Immunol. 2008:180:4476-4486. Summary, general discussion and future perspectives



Samenvatting

SAMENVATTING

Cachexie is een frequent voorkomend fenomeen in chronische obstructieve longziekten (COPD) en niet-kleincellig longkanker (NSCLC) en omvat het verlies van spier- en vetweefsel. Cachexie beïnvloed het klinisch beloop van de ziekte, heeft een negatieve invloed op de kwaliteit van leven en verhoogt het sterfterisico. Preventie en tijdige behandeling van cachexie vereisen adequate en vroege identificatie van risicopatiënten en vragen een gedetailleerd inzicht in de betrokken pathofysiologie. Dit proefschrift richt zich op het diagnosticeren van cachexie met behulp van beeldvorming dat wordt gebruikt in de huidige reguliere zorg en evalueert de invloed van ademhalingsmechanica en bruin vet weefselactiviteit op het energiemetabolisme in cachexie bij longziekten.

Momenteel evolueert de rol van medische beeldvorming van een diagnostisch hulpmiddel naar een uitgebreidere fenotypering van ziekten. **Hoofdstuk 2** bespreekt de mogelijkheid van nieuwe analysetechnieken die kunnen worden toegepast op traditionele beeldvormingsmodaliteiten om het begrip van de complexiteit van longziekten te verbeteren. Het extraheren van morfologische informatie en patronen van comorbiditeiten uit reeds beschikbare medische beelden die zijn gemaakt in routine klinische zorg, kan helpen om de ziektelast in kaart te brengen en om therapie te personaliseren.

Om skeletspierweefsel op computer tomografie scans te kwantificeren in de context van analyse van lichaamssamenstelling, wordt meestal het derde lumbale wervel niveau gebruikt. Echter, borstscans reiken doorgaans niet verder dan het eerste lumbale wervel niveau. **Hoofdstuk 3** laat zien dat het eerste lumbale wervel niveau kan worden gebruikt als alternatief voor het derde lumbale wervel niveau voor analyse van spiermassa op reguliere borstscans.

De huidige kennis over de pathofysiologie van cachexie bij COPD wordt in **hoofdstuk 4** besproken. Hoewel vaststaat dat verschillende componenten van het energieverbruik van het lichaam kunnen zijn verhoogd bij patiënten met COPD, is er weinig bekend over de invloed van ademhalingsmechanica en bruin vet op hypermetabolisme bij longziekten gerelateerde cachexie.

Patiënten met ernstig emfyseem hebben vaak hyperinflatie. In **hoofdstuk 5** wordt aangetoond dat succesvolle behandeling van deze patiënten met bronchoscopische longvolumereductie resulteert in toename en remodellering van skeletspieren. Bovendien is de toename van skeletspier en intramusculair vet geassocieerd met verbeterde inspanningscapaciteit na bronchoscopische longvolumereductie, onafhankelijk van hyperinflatiereductie.

Samenvatting

Bronchoscopische longvolumereductie werd ook gebruikt om de invloed van ademhalingsmechanica op de energiebehoefte bij emfyseem te bestuderen, aangezien longvolumereductie thoracale hyperinflatie vermindert, ademhalingsfrequentie vermindert en mechanische beperkingen op longvolume expansie vermindert, waardoor de ademmechanica verbetert. **Hoofdstuk 6** laat zien dat 90% van de patiënten met hyperinflatie die in aanmerking kwamen voor bronchoscopische longvolumereductie, een lage vetvrije massa hadden, ondanks een normaal BMI. Bovendien waren de patiënten hypermetabool (130% van voorspeld) en was het energieverbruik voor fysieke activiteit erg laag wegens de zeer lage fysieke activiteit. Een mediane hyperinflatie reductie van 25% (residuale volume daalde van 236% van voorspeld naar 193% van voorspeld) verbeterde hypermetabolisme niet bij patiënten met ernstig emfyseem.

Er werd gedacht aan activatie van bruin vetweefsel (BAT) of verbruining van wit vetweefsel, als mogelijke trigger voor hypermetabolisme. Terwijl wit vetweefsel het vermogen heeft om energie op te slaan, bezit BAT de eigenschap om, middels oxidatie van vetzuren en glucose, energie vrij te zetten in de vorm van warmte. **Hoofdstuk 7** presenteert de eerste prospectieve humane studie die de rol van BAT activatie in de ontwikkeling van cachexie in COPD onderzoekt. Vergeleken met leeftijd, geslacht en BMI gematchte niet-rokende controles, vertoonden COPD patiënten geen toegenomen BAT of veranderde genexpressie van BAT-markers in wit vetweefsel. Deze studie pleit tegen een rol van activering van BAT of verbruining van wit vetweefsel als een trigger voor hypermetabolisme in COPD. In deze studie was ook beoogd om het effect van BAT op kanker gerelateerde cachexie te bestuderen, maar het was onmogelijk om voldoende proefpersonen te includeren omdat de invasieve studiemethoden niet haalbaar bleken in deze kwetsbare populatie.

In kanker-gerelateerde cachexie hebben experimentele studies gesuggereerd dat kankertherapie die gericht is op de behandeling van de tumor ook andere weefsels kan aantasten en daardoor kan bijdragen aan verlies van die weefsels. Gelijktijdige chemoradiatie, de standaardbehandeling voor irresectabel lokaal uitgebreid NSCLC, veroorzaakt vaak bestralingsoesofagitis en bijbehorende dysfagie. Er wordt daarom aangenomen dat gewichtsverlies het gevolg is van bestralingsoesofagitis resulterend in verminderde voedselinname. Desalniettemin ondervond één op de zes patiënten met een stabiel gewicht voor aanvang van de gelijktijdige chemoradiatie, gewichtsverlies voorafgaand aan het begin van bestralingsesophagitis (**hoofdstuk 8**). Dit betekent dat er andere mechanismen zijn die gewichtsverlies veroorzaken. Bovendien werden lagere overlevingspercentages aangetoond voor patiënten met vroeg gewichtsverlies. Daarnaast kan chemotherapie zelf ook de lichaamssamenstelling beïnvloeden. **Hoofdstuk 9** laat inderdaad zien dat 69% van de patiënten met stadium IV NSCLC spiermassa verloren na twee cycli van eerstelijns chemotherapie. In vergelijking met patiënten met behouden spiermassa, vertoonden patiënten die spiermassa verloren, slechtere overlevingskansen. Een meer gedetailleerde kijk op de veranderingen in spierweefsels onthulde dat naast het verlies van spiermassa een afname van skeletspier radiatie attenuatie werd waargenomen, waarvan wordt aangenomen dat het vetinfiltratie in de spier weerspiegelt. Waarom sommige patiënten gevoeliger zijn voor spierafbraak dan anderen, vereist nader onderzoek.

In **hoofdstuk 10** werd het potentieel van radiomics beeldeigenschappen van skeletspieren onderzocht om toekomstig spierverlies te voorspellen. Radiomics is een methode om kwantitatieve kenmerken zoals vorm, grootte, intensiteit en textuur te extraheren uit medische beelden. Tot nu toe is radiomics vooral toegepast om tumorkenmerken te extraheren bij oncologische patiënten voor visualisatie van tumorheterogeniteit en het inschatten van de prognose. Radiomics beeldeigenschappen van skeletspieren waren niet voorspellend voor skeletspierverlies. We hebben echter longitudinale verschillen in radiomics beeldeigenschappen waargenomen tussen degenen die spiermassa hebben verloren en degenen die spiermassa hebben behouden. Toekomstige beeldvormingsanalyse gecombineerd met spierweefselanalyse kan de biologische processen ontrafelen die verband houden met de radiomics beeldeigenschappen.

Hoofdstuk 11 positioneert de bevindingen van het huidige proefschrift in een breder wetenschappelijk en klinisch perspectief, rekening houdend met de nieuwste ontwikkelingen op het gebied van COPD en longkanker management.



Valorisation

This chapter describes the valorisation of the research illustrated in this thesis, which is essentially the creation of value from knowledge [1,2]. This chapter therefore summarizes and reflects on how the findings of this thesis can be utilized outside the scientific field. First we will describe the main public relevance. Next, the main findings and their impact for health care and the future plans will be discussed.

According to the World Health Organization, chronic obstructive pulmonary disease (COPD) and lung cancer are both in the top 10 world's biggest killers. Globally, it is estimated that in 2015 3.17 and 1.69 million deaths were caused by COPD and lung cancer, respectively [3,4]. With ageing and rising smoking prevalence in developing nations, the prevalence of COPD and lung cancer is expected to further increase in the coming years [5,6]. Although COPD and lung cancer are primarily lung diseases, both often coexist with cachexia [7,8], which encompasses depletion of skeletal muscle and/or adipose tissue. Cachexia is associated with a range of adverse clinical outcomes including reduced exercise capacity [9], decreased health status [10] and diminished quality of life [11], it interferes with treatment efficacy [12] and impacts mortality rates [13].

Prevention and timely treatment of cachexia in patients with COPD and lung cancer requires adequate and early identification of patients at risk and detailed understanding of the pathophysiology involved. Overall, the research described in this doctoral thesis focussed on new insights in order to disentangle disrupted energy metabolism in cachexia. Cachexia is a complex process being a consequence of changes in the control of metabolism. This thesis focussed on putative triggers of increased energy metabolism in COPD and in lung cancer, amongst others with use of standard-of-care medical images.

In COPD, the prevailing believe is that increased energy metabolism is caused by high workload of breathing. An intervention to enhance breathing mechanics by reducing hyperinflation is an unique model to test this hypothesis. While this intervention to enhance breathing mechanics has resulted in improved pulmonary function and improved physical activity level, no systemic effects on the energy balance were observed.

Cachexia has long been regarded as a muscle wasting disorder, but more recently adipose tissue has started to gain attention as possible target for cachexia. In the overall disease trajectory in COPD there were no leads for a role of increased adipose tissue metabolism in cachexia. However, its role in a phase when cachexia develops progressively and patients exhibited metabolic derangements is unknown. In order to measure increased adipose tissue metabolism, extensive cold exposure followed by scanning is currently the only reliable protocol in a clinical setting. Nevertheless, this study protocol turned out to be too strenuous for the fragile cachectic population. Consequently, development of more appropriate study methodologies

to investigate the role of brown adipose tissue for this vulnerable population is subject for future research.

In the context of cachexia in lung cancer, extensive analysis of baseline standard-of-care medical images found no potential to predict future cachexia development. However longitudinally, this thesis showed a clear added value for the use of clinically acquires computed tomography scans for evaluation of body composition. Muscle wasting is clinically relevant and body composition data have the potential to improve risk stratification and personalize lifestyle interventions. It is therefore important to link longitudinal body composition analysis to evaluation of regular care images in order to gain more insight in cachexia development at an early stage. Currently, quantification of body composition occurs manually, which is a time consuming process. Automated methods will facilitate integration into regular clinical care. Thereby, imaging based tumour response evaluation during anti-tumour therapy can be combined with body composition analysis and can be captured in radiology reports. This should be combined with recording of body weight during each hospital visit of the patient in order to closely monitor and to assess risk stratification.

Optimal therapeutic intervention in cachexia depends on both proper insight into the precise mechanisms and timely identification of patients at risk. This research provided new scientific insights in the aetiology and showed a clear added value for the use of clinically acquires computed tomography scans for evaluation of body composition. Hopefully this will prove valuable for future research and contribute to optimization of current treatments and contribute to new treatment options for cachexia.

REFERENCES

- Vereniging Samenwerkende Nederlandse Universiteiten (VSNU). Raamwerk valorisatieindicatoren. 2013; http://www.vsnu.nl/ valorisatie-indicatoren.html.
- De Jonge B, Louwaars N. Valorizing science: whose values? Science & society series on convergence research. EMBO reports. 2009;10:535-539.
- World Health organization(WHO). Factsheet Chronic obstructive pulmonary disease (COPD).
 2017; http://www.who.int/news-room/factsheets/detail/chronic-obstructive-pulmonarydisease-(copd).
- World Health organization(WHO). Factsheet cancer. 2017; http://www.who.int/news-room/ fact-sheets/detail/cancer.
- Youlden DR, Cramb SM, Baade PD. The International Epidemiology of Lung Cancer: geographical distribution and secular trends. Journal of thoracic oncology : official publication of the International Association for the Study of Lung Cancer. 2008;3:819-831.
- Adeloye D, Chua S, Lee C, et al. Global and regional estimates of COPD prevalence: Systematic review and meta-analysis. Journal of global health. 2015;5:020415.
- Dewys WD, Begg C, Lavin PT, et al. Prognostic effect of weight loss prior to chemotherapy in cancer patients. Eastern Cooperative Oncology Group. The American journal of medicine. 1980;69:491-497.
- Schols AM, Soeters PB, Dingemans AM, Mostert R, Frantzen PJ, Wouters EF. Prevalence and characteristics of nutritional depletion in patients with stable COPD eligible for pulmonary rehabilitation. The American review of respiratory disease. 1993;147:1151-1156.
- Engelen MP, Schols AM, Baken WC, Wesseling GJ, Wouters EF. Nutritional depletion in relation to respiratory and peripheral skeletal muscle function in out-patients with COPD. Eur Respir J. 1994;7:1793-1797.
- Mostert R, Goris A, Weling-Scheepers C, Wouters EF, Schols AM. Tissue depletion and health related quality of life in patients with chronic obstructive pulmonary disease. Respir Med. 2000;94:859-867.
- Bye A, Sjoblom B, Wentzel-Larsen T, et al. Muscle mass and association to quality of life in non-small cell lung cancer patients. J Cachexia Sarcopenia Muscle. 2017;8:759-767.
- 12. Sjoblom B, Gronberg BH, Benth JS, et al. Low

muscle mass is associated with chemotherapyinduced haematological toxicity in advanced nonsmall cell lung cancer. Lung Cancer. 2015;90:85-91.

 Martin L, Birdsell L, Macdonald N, et al. Cancer cachexia in the age of obesity: skeletal muscle depletion is a powerful prognostic factor, independent of body mass index. J Clin Oncol. 2013;31:1539-1547.



List of publications

Karin J.C. Sanders, Roel Wierts, Wouter D. van Marken Lichtenbelt, Judith de Vos-Geelen, Guy Plasqui, Marco C.J.M. Kelders, Vera B. Schrauwen-Hinderling, Jan Bucerius, Anne-Marie C. Dingemans, Felix M. Mottaghy, Annemie M.W.J. Schols. Brown adipose tissue activation in relation to energy metabolism in emphysematous COPD patients. Submitted.

Karin J.C. Sanders, Karin Klooster, Lowie E.G.W. Vanfleteren, Guy Plasqui, Anne-Marie C. Dingemans, Dirk-Jan Slebos, Annemie M.W.J. Schols. Effect of bronchoscopic lung volume reduction in patients with advanced emphysema on energy balance regulation. Accepted.

Chris Burtin*, Jacques Bezuidenhout*, **Karin J.C. Sanders***, Anne-Marie C. Dingemans, Annemie M.W.J. Schols, Stephanie Peeters, Martijn A. Spruit, Dirk De Ruysscher. Handgrip weakness and low fat-free mass are prognostic factors for overall survival in patients with non-small cell lung cancer treated with curative-intent radiotherapy. J Cachexia Sarcopenia Muscle. 2020 Apr;11(2):424-431. *authors contributed equally.

Rosanne JH.C.G. Beijers, Harry R. Gosker, **Karin J.C. Sanders**, Chiel de Theije, Marco Kelders, Gerard Clarke, John F. Cryan, Bram van den Borst, Annemie M.W.J. Schols. Resveratrol and metabolic health in COPD: A proof-of-concept randomized controlled trial. Clin Nutr. 2020 Jan 13;S0261-5614(20)30007-8.

Evelyn E.C. de Jong, **Karin J.C. Sanders**, Timo M. Deist, Wouter van Elmpt, A. Jochems, Janita E. van Timmeren, Ralph T.H. Leijenaar, Juliette H.R.J. Degens, Annemie M.W.J. Schols, Anne-Marie C. Dingemans and Philippe Lambin. Can radiomics help to predict skeletal muscle response to chemotherapy in stage IV non-small cell lung cancer? Eur J Cancer. 2019 Sep 9;120:107-113.

Juliette H.R.J. Degens*, **Karin J.C. Sanders***, Evelyn E.C. de Jong, Harry J.M. Groen, Egbert F. Smit, Joachim G. Aerts, Annemie M.W.J. Schols, Anne-Marie C. Dingemans. The prognostic value of early onset, CT derived loss of muscle and adipose tissue during chemotherapy in metastatic non-small cell lung cancer. Lung Cancer. 2019 Jul;133:130-135. *authors contributed equally.

Karin J.C. Sanders, Juliette H.R.J. Degens, Anne-Marie C. Dingemans, Annemie M.W.J. Schols. Cross-sectional and longitudinal assessment of muscle from regular chest computed tomography scans: L1 and pectoralis muscle compared to L3 as reference in non-small cell lung cancer. Int J Chron Obstruct Pulmon Dis. 2019 Apr 3;14:781-789.

Ellen Boswijk, **Karin J.C. Sanders**, Evie P.M. Broeders, Marlies de Ligt, Guy H.E.J. Vijgen, Bas Havekes, Alma M.A. Mingels, Roel Wierts, Wouter D. van Marken Lichtenbelt, Patrick Schrauwen, Felix M. Mottaghy, Joachim E. Wildberger, Jan Bucerius. TSH suppression aggravates arterial inflammation- an 18F-FDG PET study in thyroid carcinoma patients. Eur J Nucl Med Mol Imaging. 2019 Jul;46(7):1428-1438.

Karin J.C. Sanders, Karin Klooster, Lowie E.G.W. Vanfleteren, Dirk-Jan Slebos, Annemie M.W.J. Schols. Computed tomography derived muscle remodelling after bronchoscopic lung volume reduction in advanced emphysema. Thorax. 2019 Feb;74(2):206-207.

Karin J.C. Sanders^{*}, Samuel Y. Ash^{*}, George R. Washko, Felix M. Mottaghy, Annemie M.W.J. Schols. Imaging approaches to understand disease complexity: Chronic Obstructive Pulmonary Disease as a clinical model. J Appl Physiol (1985). 2018 Feb 1;124(2):512-520. *authors contributed equally.

Karin J.C. Sanders, Lizza E. Hendriks, Esther G.C. Troost, Gerben P. Bootsma, Ruud M.A. Houben, Annemie M.W.J. Schols, Anne-Marie C. Dingemans. Early weight loss during chemoradiotherapy has detrimental impact on outcome in NSCLC. J Thorac Oncol. 2016 Jun;11(6):873-9.

Judith de Vos-Geelen*, **Karin J.C. Sanders***, Annemie M.W.J. Schols, Anne-Marie C. Dingemans. Pharmaceutical treatment options in cancer cachexia. Ned Tijdschr Oncol 2016;13(3):72-80. *authors contributed equally.

Karin J.C. Sanders*, Anita E.M. Kneppers*, Coby van de Bool, Ramon C.J. Langen, Annemie M.W.J. Schols. Cachexia in chronic obstructive pulmonary disease: new insights and therapeutic perspective. J Cachexia Sarcopenia Muscle. 2016 Mar;7(1):5-22. *authors contributed equally.

Karin J.C. Sanders, Mieke C. Hermans, Ger H. Koek. Persistently elevated hepatic transaminases: a patient with late-onset Pompe disease. Ned Tijdschr Geneeskd. 2016;160:A9904.


Acknowledgment

De afgelopen jaren waren voor mij meer dan het werken aan de manuscripten. Ik deed ervaring op in de wetenschap, leerde veel over mezelf en heb kennis mogen maken met inspirerende mensen. Zoals beschreven in de tien geboden van wetenschappelijk onderzoek, is het uitvoeren van promotieonderzoek niet iets wat je alleen doet ('Everyone needs help; It takes a village to do research') [Mireles-Cabodevila et al. Chest 2009]. Daarom wil ik de volgende mensen graag bedanken voor hun bijdrage aan de totstandkoming van dit proefschrift:

Allereerst ben ik alle patiënten en gezonde vrijwilligers zeer dankbaar voor hun participatie in ons onderzoek. Jullie medewerking was onmisbaar!

Mijn promotieteam, prof. dr. Schols, prof. dr. Dingemans, prof. dr. Mottaghy, jullie vormden het fundament van mijn promotie. Dank voor het mogelijk maken van dit onderzoek, jullie betrokkenheid, enthousiasme als er weer nieuwe studieresultaten binnen druppelden, snelle reacties op mijn vragen, jullie inzichten en kritische blik. Hartelijk dank voor deze leerzame ervaring en bovenal een leuke tijd!

De boordelingscommissie, bestaande uit prof. dr. K.H.C. Dejong, prof. dr. M. Fallon, dr. F.M.E. Franssen, prof. dr. M.A.E. de van der Schueren en prof. dr. J.E. Wildberger, wil ik bedanken voor het beoordelen van dit proefschrift.

De paranimfen, Juliette en Elke, gaaf dat ik te midden van jullie dit proefschrift mag verdedigen! Juliette, bedankt voor de vele gezellige momenten op en buiten het werk en voor je klushulp. Elke, dank voor de talloze koffiemomenten, de gesprekken en het oppassen op Julie.

Niks is zo belangrijk als fijne en behulpzame collega's. 'PUL collega's'; Chiel, Marco, Harry, Ramon, Wessel, Wouter, Pieter, Sarah, Judith, Anita, Mieke, Juanita, Niki, Caspar, Cheryl, Coby, Rianne, Martijn, Lisanne, Viviënne, Bregtje en Vasili: Dank voor jullie gezelligheid, het mij wegwijs maken als beginnend promovendus, jullie interesse, de klushulp, de hulp tijdens diverse meetdagen, jullie bijdragen aan mijn analyses en de kritische vragen tijdens de labmeetings.

Jules, mijn oud-kamergenootje. Heel erg bedankt voor de fijne en oprechte gesprekken welke een mooie afwisseling waren tussen het analyseren en schrijven van artikelen. Ik kijk ernaar uit om binnenkort als longartsen in opleiding weer samen op de werkvloer te mogen staan!

Stephanie, ontzettend veel dank voor jouw enorme inzet voor alle klinische studies die we samen hebben gedraaid. Jouw praktische, logistieke en administratieve bijdragen zijn zeer waardevol geweest. Dankjewel voor alles! Rosanne, bedankt voor het beantwoorden van mijn eindeloze stroom vragen omtrent METC, CTCM, meetmethodieken, proefpersonen, meetruimtes enzovoort, enzovoort...

Lizza, samen met jou heb ik mijn eerste stapjes gezet in het wetenschappelijk publiceren. Dankjewel dat ik verder mocht werken aan de door jou opgezette dataset van vroegtijdig gewichtsverlies. Daarnaast wil ik je bedanken voor het meehelpen met rekruteren van proefpersonen.

Evelyn, bedankt voor al je inspanningen van ons gezamenlijk radiomics cachexie project.

'Bruin vet collega's': Wouter, dank dat jouw deur altijd open stond. Judith, dank dat je mij hebt geïntroduceerd in de wereld van METC-aanvragen, CRFs en bruin vet metingen. Michiel en Marie-Fleur, bedankt dat ik altijd even mocht binnenlopen om vragen te stellen, voor jullie praktische hulp tijdens de meetdagen en jullie inzichten tijdens het analyseren van de data. Roel, dankzij jouw tomeloze inzet is het uiteindelijk gelukt om bruin vet te kunnen kwantificeren op PET-MRI beelden. Veel dank hiervoor.

'Klepjes collega's': Mijn dank gaat in het bijzonder uit naar Karin en Dirk-Jan, voor jullie gastvrijheid tijdens mijn bezoekjes aan Groningen, voor de inzet bij onze gezamenlijke studie naar energiemetabolisme, en wil ik mijn dank uitspreken voor de mogelijkheid om met de STELVIO data te werken. Lowie, bedankt voor het rekruteren van proefpersonen, jouw interesse, en met de hulp van het schrijven van de manuscripten.

Stephanie en Mark, dankjewel voor het beschikbaar stellen van jullie huis als ik weer eens ergens rustig wilde gaan werken aan mijn proefschrift.

Lieve familie, schoonfamilie en vrienden, het zal voor jullie vaak een raadsel zijn geweest waar ik mij de afgelopen jaren mee bezig hield. Zie hier het resultaat. Bedankt voor jullie support en belangstelling, maar bovenal voor de gezellige momenten.

Lieve papa en mama, van jongs af aan hebben jullie mij, Bart, Linda en Menno altijd gestimuleerd om uitdagingen niet uit de weg te gaan. Van meisje op de MAVO naar een afgerond promotieonderzoek en longarts in opleiding. Dankjewel voor jullie onvoorwaardelijke steun en vertrouwen. Lieve Koos, ik weet niet hoezeer ik jou moet bedanken voor alles. In korte tijd hebben we beiden onze promoties afgerond, zijn we gestart met de specialistenopleiding, hebben we een huis gekocht en verbouwd, hebben we een dochter gekregen en is er een zoon op komst. Ik ben er trots op hoe we alles samen hebben weten te combineren en ik kijk uit naar de komende jaren met ons prachtige gezin!

Julie, de titel 'doctor' is een mooie beloning voor het werk, maar de allermooiste titel 'moeder' heb jij mij gegeven!



Curriculum Vitae

Karin Sanders was born on February 2nd 1987, in Eindhoven, the Netherlands. After the completion of both an economics and sociology track and a health track in secondary school, she studied Molecular Life Sciences at the Maastricht University (Maastricht, the Netherlands). She received her Bachelor's degree in 2010, and subsequently enrolled in the 4-year Physician-Clinical Investigator Research Master at the Maastricht University. In 2014 she graduated and started her Ph.D. project under the supervision of prof. dr. Schols, prof. dr. Dingemans and prof. dr. Mottaghy at the department of Respiratory Medicine at Maastricht University Medical Centre+ (Maastricht, the Netherlands). In 2019 she started with her residency training of pulmonology at the Maastricht University Medical Centre+ (Maastricht, the Netherlands) under the supervision of prof. Hochstenbag.